

HELMERICH & PAYNE, INC.

FORM 10-Q (Quarterly Report)

Filed 08/14/02 for the Period Ending 06/30/02

Address 1437 S. BOULDER AVE. SUITE 1400

TULSA, OK, 74119

Telephone 918-742-5531

CIK 0000046765

Symbol HP

SIC Code 1381 - Drilling Oil and Gas Wells

Industry Oil & Gas Drilling

Sector Energy

Fiscal Year 09/30



HELMERICH & PAYNE INC

FORM 10-Q (Quarterly Report)

Filed 8/14/2002 For Period Ending 6/30/2002

Address UTICA AT 21ST ST

TULSA, Oklahoma 74114

Telephone 918-742-5531 CIK 0000046765

Industry Oil Well Services & Equipment

Sector Energy Fiscal Year 09/30



FORM 10-Q

SECURITIES AND EXCHANGE COMMISSION

Washington, D. C. 20549

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For quarterly period ended: June 30, 2002

[] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

HELMERICH & PAYNE, INC.

(Exact name of registrant as specified in its charter)

DELAWARE

(State or other jurisdiction of incorporation or organization)

73-0679879 (I.R.S. Employer I.D. Number)

UTICA AT TWENTY-FIRST STREET, TULSA, OKLAHOMA 74114

(Address of principal executive office) (Zip Code)

Registrant's telephone number, including area code: (918) 742-5531

Former name, former address and former fiscal year, if changed since last report:

NONE

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing

requirements for the past 90 days. YES X $\,$ NO $\,$ $\,$ $\,$ $\,$ $\,$ $\,$

CLASS
Common Stock, \$0.10 par value

OUTSTANDING AT JUNE 30, 2002 49.980.947

TOTAL NUMBER OF PAGES - 24

HELMERICH & PAYNE, INC.

INDEX

PART	I.	FINANCIAL INFORMATION	
	Item 1.	Financial Statements	
		Consolidated Condensed Balance Sheets as of June 30, 2002 and September 30, 2001	3
		Consolidated Condensed Statements of Income for the Three Months and Nine Months Ended June 30, 2002 and 2001	4
		Consolidated Condensed Statements of Cash Flows for the Nine Months Ended June 30, 2002 and 2001	5
		Consolidated Condensed Statement of Shareholders' Equity For the Nine Months Ended June 30, 2002	6
		Notes to Consolidated Condensed Financial Statements	7 - 14
	Item 2.	Management's Discussion and Analysis of Results of Operations and Financial Condition	15 - 21
	Item 3.	Quantitative and Qualitative Disclosures about Market Risk	22
PART	II. OTHER IN	FORMATION	22
	Item 1.	Legal Proceedings	22
	Item 6.	Exhibits and Reports on Form 8-K	22 - 23
	Signatures .		23
	Exhibit Index	x	24

PART I. FINANCIAL INFORMATION HELMERICH & PAYNE, INC.

Item 1. FINANCIAL STATEMENTS

CONSOLIDATED CONDENSED BALANCE SHEETS

(in thousands)

	(Unaudited) June 30, 2002	September 30, 2001		
ASSETS				
Current assets:				
Cash and cash equivalents	\$ 48,200			
Accounts receivable, net Inventories		147,235		
Prepaid expenses and other	26,964 19,263	28,934 32,281		
Total current assets	216,694	331,412		
Investments	167 356	200,286		
Property, plant and equipment, net	978,414	818,404		
Other assets	12,796	14,405		
Total assets	\$ 1,375,260	\$ 1,364,507		
LIABILITIES AND SHAREHOLDERS' EQUITY Current liabilities:				
Accounts payable	\$ 29,822	\$ 67,595		
Accrued liabilities	54,131	53,626		
Total current liabilities	83,953	121,221		
Noncurrent liabilities:	F0 000	F0 000		
Long-term notes payable Deferred income taxes	50,000 155,056	50,000 144,439		
Other	24,784	22,370		
Total noncurrent liabilities	229,840	216,809		
SHAREHOLDERS' EQUITY				
Common stock, par value \$.10 per				
share	5,353	5,353		
Preferred stock, no shares issued				
Additional paid-in capital Retained earnings	81,881	80,324		
Unearned compensation	986,559 (713)	943,105 (1,812)		
Accumulated other comprehensive income	36,453	49,309		
-				
Tana karan a akada akan barat	1,109,533	1,076,279		
Less treasury stock, at cost	48,066	49,802		
Total shareholders' equity	1,061,467	1,026,477		
Motel lightlifting and shoughtlifting and shoughtlifting	ė 1 27F 262	ė 1 264 FAR		
Total liabilities and shareholders' equity	\$ 1,375,260 =======	\$ 1,364,507 =======		

See accompanying notes to financial statements.

HELMERICH & PAYNE, INC. CONSOLIDATED CONDENSED STATEMENTS OF INCOME

(Unaudited)

(in thousands, except per share data)

		Three Mon June		Ended		Nine Mont		nded
		2002		2001		2002		2001
REVENUES: Sales and other operating revenues	Ś	159,912	Ś	212,573	Ś	486,668	Ś	621,138
Income from investments	·	25,555	·			28,522		
		185,467		217,222				
COST AND EXPENSES:								
Operating costs		91,464		111,120		290,714		312,242
Depreciation, depletion and amortization		25.728		21,341		74.614		62.103
Dry holes and abandonments								25,626
Taxes, other than income taxes		10,389		6,878 10,276		28,502		31,210
General and administrative		4,127		3,449		14,053		
Interest				(1,626)		32		(951)
		136,652				423,666		
INCOME BEFORE INCOME TAXES AND EQUITY IN INCOME OF AFFILIATES				65,784				
PROVISION FOR INCOME TAXES		21,259		25,679		39,354		74,832
EQUITY IN INCOME OF AFFILIATES,								
net of income taxes		662		332		2,524		1,409
NET INCOME	\$	28,218	Ś	40,437	Ś	54,694	Ś	116,026
				=======				
EARNINGS PER COMMON SHARE:								
Basic	\$	0.57	\$	0.80	\$	1.10	\$	2.31
Diluted	·	0.56			\$	1.09	\$	2.28
CASH DIVIDENDS	\$	0.08	\$	0.075	\$	0.23	\$	0.225
AVERAGE COMMON SHARES OUTSTANDING:								
Basic		49,855		50,467		49,793		50,159
Diluted				51,256				

The accompanying notes are an integral part of these statements.

HELMERICH & PAYNE, INC. CONSOLIDATED CONDENSED STATEMENT OF CASH FLOWS

(Unaudited)

(in thousands)

	Nine Mont June	
		2001
CASH FLOWS FROM OPERATING ACTIVITIES:		
Net Income	\$ 54,694	\$ 116,026
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation, depletion and amortization	74 614	62,103
Dry holes and abandonments	15,751	25,626
Equity in income of affiliates before income taxes		
Amortization of deferred compensation	1,099	1,112
Gain on sale of securities and non-monetary	,	,
investment income	(25,078)	(2,634)
Gain on sale of property, plant & equipment	(1,321)	(3,759)
Other, net	3	22
Change in assets and liabilities-		
Accounts receivable	25,408	(28,112)
Inventories	1,970	(5,274)
Prepaid expenses and other	14,627	11,200
Accounts payable	(33,960)	15,432
Accrued liabilities	(411)	(5,274) 11,200 15,432 22,451 (1,417)
Deferred income taxes Other noncurrent liabilities	18,495	(1,41/)
Other honcurrent Habilities	4,005	1,425
NET CASH PROVIDED BY OPERATING ACTIVITIES	145,200	211,303
		211,303
CACH BLOWG BROW INTERESTED ACRES/INTERES.		
CASH FLOWS FROM INVESTING ACTIVITIES: Capital expenditures, including dry hole costs	(257 404)	(100 (01)
Proceeds from sales of property, plant and equipment	(257,494)	(186,664)
Proceeds from sale of investments	4,024	24 438
1100ccab 110m bate of investments	4,624 41,489	
NET CASH USED IN INVESTING ACTIVITIES	(211,381)	(151,750)
CASH FLOWS FROM FINANCING ACTIVITIES:		
Dividends paid	(11,240)	(11,366)
Purchases of stock for Treasury		(1,921)
Proceeds from exercise of stock options	2,659	13,521
NET CASH PROVIDED BY (USED IN) FINANCING ACTIVITIES	/O E01\	224
NEI CASH PROVIDED DI (USED IN) FINANCING ACTIVILLES	(8,581)	234
NET INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS	(74 762)	59 787
CASH AND CASH EQUIVALENTS, BEGINNING OF PERIOD	122,962	59,787 108,087
CASH AND CASH EQUIVALENTS, END OF PERIOD		
OLDER THE CHOIC BY OF THE OF THE COD	\$ 48,200 ======	=======

HELMERICH & PAYNE, INC. CONSOLIDATED CONDENSED STATEMENT OF SHAREHOLDERS' EQUITY

(in thousands - except per share data)

		Common Stock Additional Paid-In U		IIn	earned	Re	etained	Trea	sury	Stock		
	Shares		mount	C	Capital		ensation	E	Earnings	Shares		Amount
Balance, September 30, 2001	53,529	\$	5,353		80,324	\$	(1,812)		943,105	3,676		(49,802)
Comprehensive Income:												
Net Income Other comprehensive income, Unrealized losses on available- for-sale securities, net of deferred taxes of \$7,894 Derivatives instruments gains, net of deferred taxes of \$16 Total other comprehensive income Comprehensive income									54,694			
Cash dividends (\$0.23 per share) Exercise of stock options Tax benefit of stock-based awards Amortization of deferred compensation					923 634		1,099		(11,240)	(128)		1,736
-												
Balance, June 30, 2002	53,529 =====	\$ ==	5,353 =====		81,881	\$ ==:	(713) =====		986,559	3,548		(48,066)
	Accumul Othe Comprehe Inco	r nsive me	Shar	otal eholo Equit	У							
Balance, September 30, 2001	\$ 49	,309	\$	1,026	5,477							
Comprehensive Income:												
Net Income Other comprehensive income, Unrealized losses on available-				54	1,694							
for-sale securities, net of deferred taxes of \$7,894 Derivatives instruments gains, net	(12	,881)		(12	2,881)							
of deferred taxes of \$16		25			25							
Total other comprehensive income Comprehensive income	(12	,856)			2,856) .,838							
Gh dividand- (60 22h)				/11	240)							
Cash dividends (\$0.23 per share) Exercise of stock options					.,240) 2,659							
Tax benefit of stock-based awards					634							
Amortization of deferred compensation					.,099							
Balance, June 30, 2002	\$ 36 =====	•	\$	1,061	,467							
	====		==									

HELMERICH & PAYNE, INC.

NOTES TO CONSOLIDATED CONDENSED FINANCIAL STATEMENTS

(Unaudited)

1. In the opinion of the Company, the accompanying unaudited condensed consolidated financial statements contain all adjustments, which consists only of those of a normal recurring nature, necessary to present fairly the results of the periods presented. The results of operations for the three and nine months ended June 30, 2002, and June 30, 2001, are not necessarily indicative of the results to be expected for the full year. These condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto in the Company's 2001 Annual Report on Form 10-K and the Company's 2002 First and Second Quarter Reports on Form 10-Q.

As announced on February 25, 2002, the Company and Key Production Company, Inc. (Key) have signed a definitive agreement that provides for Helmerich & Payne to contribute the assets and liabilities of the Oil and Gas Division to a newly formed subsidiary, Cimarex Energy Co. (Cimarex), and distribute on a pro-rata basis all of the shares of stock of Cimarex to the shareholders of Helmerich & Payne. Cimarex would then merge with Key. Cimarex Energy Co. will be a new publicly traded exploration and production company. The transaction will close after receipt of necessary Key shareholder and regulatory approvals. The Company has received a favorable private letter ruling from the Internal Revenue Service with regard to the transaction. Closing will likely occur prior to the end of the Company's fiscal year. Approval of the transaction by Helmerich and Payne shareholders will not be required.

- 2. The \$.075 cash dividend declared in March, 2002, was paid June 3, 2002. On June 5, 2002, a cash dividend of \$.08 per share was declared for shareholders of record on August 15, 2002, payable September 3, 2002.
- 3. Inventories consist of materials and supplies.
- 4. Income from investments includes \$15,468,000 (\$0.30 per share) and \$15,792,000 (\$0.31 per share) after-tax gains from sales of available-for-sale securities during the third quarter and first nine months of fiscal 2002. After-tax gains from security sales were \$1,423,000 and \$1,497,000 (\$0.03 per share) for the same periods in fiscal 2001.
- 5. The following is a summary of available-for-sale securities, which excludes those accounted for under the equity method of accounting. At June 30, 2002, the Company's investment in securities accounted for under the equity method is \$60,083,000.

	Cost	Unrealized Gains		alized osses	-	Fair Value
		(in	thousa	nds)		
Equity Securities 06	/30/02 \$ 46,927	\$ 60,950	\$	604	\$	107,273
Equity Securities 09	/30/01 \$ 63,778	\$ 84,257	\$	3,136	\$	144,899

(Unaudited)

6. Comprehensive Income -

The components of comprehensive income, net of related tax, are as follows (in thousands):

	Three Months Ended June 30,				Months Ended une 30,			
		2002		2001	2002		2001	
Net Income	\$	28,218	\$	40,437	\$ 54,694	\$	116,026	
Other comprehensive income: Net unrealized gain (loss) on								
securities Net unrealized gain (loss) on		(33,360)		939	(12,881)		(34,771)	
derivative instruments		(358)		6	 25		(22)	
Other comprehensive income (loss)		(33,718)		945	 (12,856)		(34,793)	
Comprehensive income (loss)	\$ ===	(5,500)	\$ ===	41,382	\$ 41,838	\$ ==:	81,233	

The components of accumulated other comprehensive income, net of related taxes, are as follows (in thousands):

	Jı	ne 30, 2002	Se	ept.30, 2001
Unrealized gains on securities, net Unrealized loss on derivative instruments	\$	37,414 (961)	\$	50,295 (986)
Accumulated other comprehensive income	 \$ ===	36,453	\$	49,309

7. At June 30, 2002, the Company had committed bank lines of credit totaling \$100 million; \$50 million expires in February 2003 and \$50 million expires in October 2003. Additionally, the Company had uncommitted credit facilities totaling \$10 million. Collectively, the Company had \$50 million in outstanding borrowings and outstanding letters of credit totaling \$10.6 million against these lines at June 30, 2002. Concurrent with a \$50 million borrowing under the facility that expires October 2003, the Company entered into an interest rate swap with a notional value of \$50 million and an expiration date of October 2003. The swap effectively converts this \$50 million facility from a floating rate of LIBOR plus 50 basis points to a fixed effective rate of 5.38 percent. Excluding the impact of the interest rate swap, the average interest rate for the borrowings at June 30, 2002, was approximately 2.34 percent on a 360-day basis.

In July 2002, the Company modified its short-term revolving credit arrangements by replacing the above mentioned \$50 million committed line that expires in February 2003 and a \$10 million uncommitted line with a \$125 million committed unsecured revolving credit facility, which expires in July, 2005. Each year the Company can request a one year extension on the facility. The facility requires quarterly interest payments with the principal payment due at the expiration date. The interest rate on the facility is LIBOR plus 87.5 basis points.

The Company is also in the process of negotiating a \$200 million intermediate term debt facility that would provide staged maturities ranging from 5 to 12 years. The Company expects to finalize this arrangement in mid-August 2002.

Under the credit agreements currently in effect, the Company must meet certain requirements regarding levels of debt, net worth, cash flows and earnings. The Company met all requirements at June 30, 2002.

(Unaudited)

8. Earnings per Share -

Basic earnings per share is based on the weighted-average number of common shares outstanding during the period. Diluted earnings per share include the dilutive effect of stock options and restricted stock.

A reconciliation of the weighted-average common shares outstanding on a basic and diluted basis is as follows (in thousands):

	Three Moni June		Nine Months Ended June 30,			
	2002	2001	2002	2001		
Basic weighted-average shares Effect of dilutive shares:	49,855	50,467	49,793	50,159		
Stock options	691	735	498	743		
Restricted stock	28	54	15	39		
	719	789	513	782		
Diluted weighted-average						
shares	50,574	51,256	50,306	50,941		
	========	========	========	========		

9. Income Taxes -

The Company's effective tax rate was 43% in the first nine months of fiscal 2002 compared to 39.5% in the same period of fiscal 2001. Costs and expenses related to certain foreign locations for which the Company does not receive a tax benefit resulted in the current year estimated effective tax rate of 43%. The two major reasons for the effective tax rate increase are that the Company had larger net operating loss carry forwards from Venezuela in fiscal 2001, and the Company does not receive a tax benefit from the devaluation losses in Argentina and Venezuela.

10. Interest Rate Risk Management -

The Company uses derivatives as part of an overall operating strategy to moderate certain financial market risks and its exposure to interest rate risk from long-term debt. To manage this risk, the Company has entered into an interest rate swap to exchange floating rate for fixed rate interest payments over the remaining life of the debt. As of June 30, 2002, the Company had an interest rate swap outstanding with a notional principal amount of \$50 million. (See Note 7)

The Company's accounting policy for these instruments is based on its designation of such instruments as hedging transactions. An instrument is designated as a hedge based in part on its effectiveness in risk reduction and one-to-one matching of derivative instruments to underlying transactions. The Company records all derivatives on the balance sheet at fair value.

(Unaudited)

For derivative instruments that are designated and qualify as a cash flow hedge (i.e., hedging the exposure of variability in expected future cash flows that is attributable to a particular risk), the effective portion of the gain or loss on the derivative instrument is reported as a component of other comprehensive income in stockholders' equity and reclassified into earnings in the same period or periods during which the hedged transaction affects earnings. The change in value of the derivative instrument in excess of the cumulative change in the present value of the future cash flows of the risk being hedged, if any, is recognized in the current earnings during the period of change.

The Company's interest rate swap has been designated as a cash flow hedge and is expected to be 100% effective in hedging the exposure of variability in the future interest payments attributable to the debt because the terms of the interest swap correlate with the terms of the debt.

11. Currency Devaluation -

The uncertainty regarding economic, banking and currency stability continues without improvement in Argentina. The development of a solution to the crisis is uncertain, increasing the potential for additional currency declines in the near term. The Argentine peso currently trades in the range of 3 to 3.6 pesos to one U.S. dollar. The Company has recorded \$1.2 million in pre-tax currency devaluation losses related to the peso during the first quarter of fiscal 2002. The Company could be exposed to additional currency losses of between \$1 million and \$2 million during the remainder of fiscal 2002. The Company currently has one rig under contract and working in Argentina.

Also, as a result of a severe decline in the value of the Venezuelan bolivar due to political instability and a change in the Venezuelan government's exchange policy, pre-tax currency devaluation losses of \$3.1 million and \$0.5 million were recorded in the first nine months and the third quarter of fiscal 2002, respectively. Subsequent to June 30, 2002, the bolivar has improved in value relative to the U.S. dollar. Should an additional devaluation of the bolivar occur, the Company could be exposed to additional currency losses of between \$0.5 million and \$1.4 million during the remainder of fiscal 2002. The Company currently has three rigs under contract and working in Venezuela.

12. Contingent Liabilities and Commitments -

Litigation Settlement -

The Company was a defendant in Verdin v. R&B Falcon Drilling USA, Inc., et al., a civil action in the United States District Court, Galveston, Texas. In May 2001, the Company reached an agreement in principle with Plaintiff's counsel to settle all claims pending court approval of the settlement. In the third quarter of fiscal 2001, the Company incurred a net charge of \$3.25 million to contract drilling expense based on the pending settlement. The Court approved the settlement on April 25, 2002. In June, 2002, the Company paid \$10 million to settle all claims in this litigation. The Company was reimbursed \$6.75 million in June, 2002 by the Company's insurer.

(Unaudited)

Other Matters -

The Company is a defendant to claims of drainage of gas from two properties that it operates. The royalty owner plaintiffs have filed suit on behalf of themselves and a class of similarly situated royalty owners in two 640 acre spacing units. The plaintiffs allege that the two units have suffered approximately 12 billion cubic feet of gross gas drainage. Although the plaintiffs have not specified in their pleadings the amount of damages alleged, the plaintiffs have orally stated that the royalty owner class has sustained actual damages of approximately \$6.2 million exclusive of interest and costs. The Company estimates that the share of such alleged damages attributable to its working interest ownership would total approximately \$1.0 million exclusive of interest and costs. Plaintiffs further allege that, as a former operator, the Company is liable for all damages attributable to the drainage. The Company believes that it is liable only for its working interest share of any actual damages attributable to the alleged drainage. In the event that the Company is held liable for the full amount of any actual damages, the Company will seek contribution, indemnification and/or other appropriate relief from all other working interest owners for their portion of the alleged drainage that is attributable to the interest of those other owners.

The Company, on a regular basis, makes commitments for the purchase of contract drilling equipment. At June 30, 2002, the Company had commitments outstanding of approximately \$170 million for the purchase of drilling equipment.

13. Segment Information -

The Company evaluates performance of its segments based upon operating profit or loss from operations before income taxes, which includes revenues from external and internal customers; operating costs; depreciation, depletion and amortization; dry holes and abandonments and taxes other than income taxes. Intersegment sales are accounted for in the same manner as sales to unaffiliated customers. Other includes investments in available-for-sale securities, equity owned investments, as well as corporate operations.

$\label{eq:helmerich def} \textbf{HELMERICH \& PAYNE, INC.} \\ \textbf{NOTES TO CONSOLIDATED CONDENSED FINANCIAL STATEMENTS - Continued}$

(Unaudited)

Summarized financial information of the Company's reportable segments for the nine months ended June 30, 2002, and 2001, is shown in the following table:

(in thousands)	External Sales	Inter- Segment	Total Sales	Operating Profit
JUNE 30, 2002				
Contract Drilling Domestic International	\$ 251,379 109,643	\$ 682 	\$ 252,061 109,643	\$ 55,709 11,840
	361,022	682	361,704	67,549
Oil & Gas Operations Exploration & Production Natural Gas Marketing	80,385 38,995		80,385 38,995	7,696 1,330
	119,380		119,380	9,026
Real Estate Other Eliminations	6,649 28,139 	1,128 (1,810)	7,777 28,139 (1,810)	4,073
Total	\$ 515,190 ======	\$ =======	\$ 515,190 =====	\$ 80,648 ======
(in thousands)	External Sales	Inter- Segment	Total Sales	Operating Profit
JUNE 30, 2001				
Contract Drilling Domestic International	\$ 228,769 114,346 343,115	\$ 2,331 2,331	\$ 231,100 114,346 345,446	\$ 67,995 19,311 87,306
Domestic	114,346		114,346 345,446 184,900 83,661	19,311 87,306 95,047 4,817
Domestic International Oil & Gas Operations Exploration & Production	114,346 343,115 184,900 83,661	2,331	114,346 345,446 184,900	19,311 87,306 95,047
Domestic International Oil & Gas Operations Exploration & Production	114,346 343,115 184,900 83,661	2,331	114,346 345,446 184,900 83,661 268,561	19,311 87,306 95,047 4,817

$\label{eq:helmerich def} \textbf{HELMERICH \& PAYNE, INC.} \\ \textbf{NOTES TO CONSOLIDATED CONDENSED FINANCIAL STATEMENTS - Continued}$

(Unaudited)

Summarized financial information of the Company's reportable segments for the quarters ended June 30, 2002, and 2001, is shown in the following table:

(in thousands)	External Inter- Sales Segment		Total Sales	Operating Profit
JUNE 30, 2002				
Contract Drilling Domestic International	\$ 80,586 31,380	\$ 144	\$ 80,730 31,380	\$ 14,360 3,547
	111,966	144	112,110	17,907
Oil & Gas Operations Exploration & Production Natural Gas Marketing	32,538 13,193		32,538 13,193	8,649 (109)
	45,731		45,731	8,540
Real Estate Other Eliminations	2,189 25,581 	368 (512)	2,557 25,581 (512)	1,340
Total	\$ 185,467 =======	\$ =======	\$ 185,467 =======	\$ 27,787 =======
(in thousands)	External Sales	Inter- Segment	Total Sales	Operating Profit
JUNE 30, 2001 Contract Drilling Domestic International	\$ 93,316 40,527	\$ 719 	\$ 94,035 40,527	\$ 30,305 7,958
	133,843	719	134,562	38,263
Oil & Gas Operations Exploration & Production Natural Gas Marketing	52,337 23,508	 	52,337 23,508	23,932 151
	75,845 		75,845 	24,083
Real Estate Other Eliminations	2,250 5,284 	383 (1,102)	2,633 5,284 (1,102)	1,008
Total	\$ 217,222 =======	\$ ========	\$ 217,222 =======	\$ 63,354 =======

((Unaudited)

The following table reconciles segment-operating profit per the table above to income before income taxes and equity in income of affiliate as reported on the Consolidated Condensed Statements of Income (in thousands).

		~	r Ended Nine 30, J				onths Ended ne 30,		
	:	2002		2001		2002		2001	
Segment operating profit	\$	27,787	\$	63,354	\$	80,648	\$	192,482	
Unallocated amounts:									
Income from investments		25,555		4,649		28,522		10,203	
General corporate expense		(4,127)		(3,449)		(14,053)		(11,662)	
Interest expense		684		1,626		(32)		951	
Corporate depreciation		(509)		(550)		(1,498)		(1,526)	
Other corporate expense		(575)		154		(2,063)		(999)	
Total unallocated amounts		21,028		2,430		10,876		(3,033)	
Income before income taxes and equity in income of									
affiliates	\$	48,815	\$	65,784	\$	91,524	\$	189,449	
	==:	======	==:	======	==	======	==	======	

The following table presents revenues from external customers by country based on the location of service provided (in thousands).

	Quarter Ended June 30,		Nine Months Ended June 30,	
	2002	2001	2002	2001
Revenues:				
United States	\$ 154,087	\$ 176,695	\$ 405,547	\$ 516,995
Venezuela	10,528	11,987	40,746	29,818
Ecuador	10,698	9,763	32,908	25,560
Colombia	1,915	6,089	7,946	20,155
Other Foreign	8,239	12,688	28,043	38,813
Total	\$ 185,467	\$ 217,222	\$ 515,190	\$ 631,341
	========	========	========	========

14. Impairment -

Included in depreciation, depletion and amortization for the three and nine month periods ended June 30, 2002 were impairment charges of \$2,464,000 and \$7,908,000, respectively for proved Exploration and Production properties. After tax, the impairment charges reduced net income by approximately \$1,528,000 and \$4,903,000 (\$0.03 and \$0.10 per share) for the three and nine month periods ended June 30, 2002, respectively. Included in depreciation, depletion and amortization for the three and nine month periods ended June 30, 2001 were impairment charges of \$642,000 and \$4,450,000 for proved Exploration and Production properties. After tax, the impairment charges reduced net income by approximately \$398,000 and \$2,759,000 (\$0.01 and \$0.05 per share) for the three and nine month periods ended June 30, 2001, respectively.

June 30, 2002

RISK FACTORS AND FORWARD-LOOKING STATEMENTS

The following discussion should be read in conjunction with the consolidated financial statements, notes and management's narrative analysis contained in the Company's 2001 Annual Report on Form 10-K and the Company's fiscal 2002 First and Second Quarter Report on Form 10-Q and the condensed consolidated financial statements and related notes included elsewhere herein. The Company's future operating results may be affected by various trends and factors, which are beyond the Company's control. These include, among other factors, fluctuations in natural gas and crude oil prices, expiration or termination of drilling contracts, currency exchange losses, changes in general economic and political conditions, rapid or unexpected changes in technologies and uncertain business conditions that affect the Company's businesses. Accordingly, past results and trends should not be used by investors to anticipate future results or trends.

With the exception of historical information, the matters discussed in Management's Discussion & Analysis of Results of Operations and Financial Condition includes forward-looking statements. These forward-looking statements are based on various assumptions. The Company cautions that, while it believes such assumptions to be reasonable and makes them in good faith, assumed facts almost always vary from actual results. The differences between assumed facts and actual results can be material. The Company is including this cautionary statement to take advantage of the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995 for any forward-looking statements made by, or on behalf of, the Company. The factors identified in this cautionary statement are important factors (but not necessarily all important factors) that could cause actual results to differ materially from those expressed in any forward-looking statement made by, or on behalf of, the Company.

RESULTS OF OPERATIONS

THIRD QUARTER 2002 VS THIRD QUARTER 2001

The Company reported net income of \$28,218,000 (\$0.56 per share) from revenues of \$185,467,000 for the third quarter ended June 30, 2002, compared to net income of \$40,437,000 (\$0.79 per share) from revenues of \$217,222,000 for the third quarter of the prior fiscal year. Net income in the third quarter of fiscal 2002 included \$15,468,000 (\$0.30 per share) of after-tax gains from the sale of available-for-sale securities compared to \$1,423,000 (\$0.03 per share) in the third quarter of fiscal 2001.

OIL & GAS DIVISION

EXPLORATION and PRODUCTION reported operating profit of \$8.6 million for the third quarter of fiscal 2002 compared to \$23.9 million for the same period of fiscal 2001. Oil & gas revenues decreased to \$32.5 million compared with \$52.3 million in 2001.

June 30, 2002 (Continued)

Natural gas revenues decreased to \$27.7 million from \$46.5 million, or 40 percent. The \$18.8 million decrease in gas revenues was the result of significantly lower gas prices (\$14.4 million decrease) and a 12% decrease in gas volumes (\$4.4 million decrease). Natural gas prices averaged \$2.94 per mcf and \$4.27 per mcf for the third quarter of fiscal 2002 and 2001, respectively, while natural gas volumes averaged 103.5 Mmcf/d and 117.8 Mmcf/d, respectively. Oil revenues decreased to \$4.5 million from \$5.4 million. The \$0.9 million decrease in oil revenues was the result of both lower prices (\$0.4 million decrease) and volumes (\$0.5 million decrease). Crude oil prices averaged \$24.34 per Bbl and \$26.12 per Bbl for the third quarter of fiscal 2002 and 2001, respectively, while crude oil volumes averaged 2,044 Bbls/d and 2,224 Bbls/d, respectively.

Exploration expenses were approximately \$5.6 million and \$9.5 million in the third quarter of fiscal 2002 and 2001, respectively. Dry hole costs were \$3.1 million for the quarter compared with \$2.6 million in 2001 and impairment of undeveloped leases was \$2.4 million compared with \$4.2 million in 2001. There were no geophysical expenses in the third quarter of fiscal 2002 compared to \$2.7 million in the same period of fiscal 2001. The decreases were the result of reduced exploration activity in fiscal 2002.

Operating expenses decreased to \$6.7 million in the third quarter of fiscal 2002 compared to \$9.6 million in the same period of fiscal 2001. Lower production taxes of \$2.5 million, as the result of lower gas prices than last year's third quarter, was the primary reason for the decrease.

The Company participated in an additional 29 wells during the third quarter, of which 24 were producing, waiting on pipeline connections or completing. Of the 29 wells drilled, 6 were wildcat wells, of which 3 were successful and 3 were dry holes. For the first nine months of fiscal 2002, the Company participated in 51 wells, of which 39 were completed or are completing, and 12 were dry holes. The Company expects to participate in 94 gross wells for the year.

NATURAL GAS MARKETING segment reported an operating loss of \$0.1 million in the third quarter of fiscal 2002 compared to an operating profit of \$0.4 million in the third quarter of fiscal 2001.

June 30, 2002 (Continued)

DOMESTIC DRILLING

Domestic contract drilling revenues for the third quarter of 2002 and 2001 were \$80.6 million and \$93.3 million, respectively. Operating profit for the third quarter of fiscal 2002 was \$14.4 million, compared with \$30.3 million in last year's third quarter. The decrease in operating profit is primarily the result of reduced performance in the Company's land operations. Land dayrates averaged approximately \$11,500 per day in the current quarter, compared to \$15,000 per day in the third quarter of fiscal 2001. Land rig utilization was 85% and 98% for the third quarter of 2002 and 2001, respectively. The average number of land rigs available in the U.S. has increased 30%, from 44.6 rigs in the third quarter of fiscal 2001, to 58.1 rigs during the current quarter. The increase is due to the additional FlexRigs constructed by the Company over the past year, as well as four rigs moved from international locations to the U.S.

Operating profit from offshore operations decreased approximately 19% in the current quarter compared to the third quarter of fiscal 2001, as rig utilization was 81% and 100% for the third quarter of 2002 and 2001, respectively. Offshore operating profit for the third quarter of 2002 was helped by the commencement of work for the Company's new offshore platform rigs 205 and 206. This improvement was more than offset with two rigs down for the entire quarter, with plans for modification, and another rig going to standby rates for the quarter.

As previously announced, the Company has begun construction of its third series of the H&P-designed FlexRig. The current series of rigs, named FlexRig3, is being assembled in the Company's construction facility in Houston. Two of the new FlexRig3s recently were deployed to the field to commence operations for customers. An additional 23 rigs are scheduled to be completed within the next 12-14 months. The next six FlexRig3s to be completed are already committed to specific customers for work. Rig utilization for all FlexRigs since 1998 have averaged over 95%. Upon the projected completion in late 2003 of the FlexRig3 project, it is estimated that the Company's land rig fleet in the U.S. will total 83 rigs, representing a 43% increase over the 58 rigs available during the current quarter.

INTERNATIONAL DRILLING

International Drilling's operating profit decreased to \$3.5 million from \$8.0 million. Revenues decreased to \$31.4 million from \$40.5 million. The operating profit decrease is due primarily to decreases in Venezuela, Colombia and Equatorial Guinea, partially offset by continued improvement from Ecuador as the result of increased margins. Colombia decreased from three rigs working in the third quarter of 2001 to one rig in the third quarter of fiscal 2002. The one remaining rig in Colombia is scheduled to end its contract in September 2002. The decrease in Venezuela is the result of lower margins compared to last year and in the third quarter of fiscal 2001 there was a one-time contract settlement of \$2.0 million

Rig utilization averaged 48% for the current quarter, compared to 60% in the third quarter of 2001. Rig activity continued to be low in all international countries except in Ecuador where the rig utilization averaged 86% for the quarter. Rig activity in Venezuela, where the Company operates 14 of its 33 international rigs, averaged 38% for the quarter.

June 30, 2002 (Continued)

OTHER

Other revenues increased approximately \$20.3 million over last year, including a \$22.5 million increase in gains from the sale of available-for-sale securities. In the current quarter, the Company sold all of its holdings in Bank One Corporation, Kerr McGee, Oneok and Sun Co. (see note 4). Dividend income decreased \$0.4 million for the quarter as compared to the third quarter of 2001. Interest income decreased to \$0.2 million compared to \$1.4 million in the third quarter of fiscal 2001 as interest rates and average cash invested both decreased in 2002.

Interest expense was a credit of \$0.7 million in the third quarter of fiscal 2002 compared to a credit of \$1.6 million for the same period of 2001. In 2002, third quarter interest expense included credits for capitalized interest and ad valorem tax settlements of \$1.2 million and \$0.2 million, respectively. For the same period of 2001, the capitalized interest was \$0.4 million and the credit for ad valorem tax settlements was \$1.9 million, respectively.

General and administrative expenses increased to \$4.1 million from \$3.4 million, due to increased professional services, pension expense and administrative labor and benefits.

The company's effective income tax rate increased to 43.5% for the third quarter of fiscal 2002 compared to 39.0% for the same period of 2001. The increase is due primarily to certain costs and expenses related to foreign locations for which the Company does not receive a tax benefit, including currency devaluation losses.

NINE MONTHS ENDED JUNE 30, 2002 VS NINE MONTHS ENDED JUNE 30, 2001

The Company reported net income of \$54,694,000 (\$1.09 per share) from revenues of \$515,190,000 for the nine months ended June 30, 2002, compared with net income of \$116,026,000 (\$2.28 per share) from revenues of \$631,341,000 for the first nine months of the prior fiscal year. Net income in the first nine months of fiscal 2002 and 2001 included \$15,792,000 (\$0.31 per share) from the sale of investment securities compared with \$1,497,000 (\$0.03 per share) for the same period of fiscal 2001.

OIL AND GAS DIVISION

EXPLORATION AND PRODUCTION reported an operating profit of \$7.7 million for the first nine months of fiscal 2002 compared to \$95.0 million for the same period of fiscal 2001. Oil & gas revenues decreased to \$80.4 million compared to \$184.9 million in 2001.

June 30, 2002 (Continued)

Natural gas revenues decreased to \$67.6 million from \$166.1 million, or 59 percent. The \$98.5 million decrease in gas revenues was the result of significantly lower gas prices (\$90.4 million decrease) and a 10% decrease in gas volumes (\$8.1 million decrease). Natural gas prices averaged \$2.34 per mcf and \$5.16 per mcf for the first nine months of fiscal 2002 and 2001 respectively. Natural gas volumes averaged 105.5 Mmcf/d and 117.5 Mmcf/d, respectively. Oil revenues decreased to \$12.0 million from \$18.0 million. The \$6.0 million decrease in oil revenues was the result of both lower prices (\$4.5 million decrease) and volumes (\$1.5 million decrease). Crude oil prices averaged \$21.33 per Bbl and \$28.65 per Bbl for the nine months of fiscal 2002 and 2001, respectively. Crude oil volumes averaged 2,052 Bbls/d and 2,303 Bbls/d, respectively.

Exploration expenses were approximately \$16.8 million and \$31.3 million in the first nine months of fiscal 2002 and 2001, respectively. A decrease in exploration activity in 2002, as a result of lower gas prices, is the primary reason for the \$14.5 million decrease in exploration expenses. Dry hole costs were \$6.8 million for the first nine months of fiscal 2002 compared to \$14.1 million in the same period of 2001. Impairment of undeveloped leases and geophysical expense were \$8.6 million and \$1.4 million, respectively, in the first nine months of fiscal 2002 compared to \$11.2 million and \$6.0 million, respectively, in the same period of fiscal 2001.

Operating expenses for the first nine months of fiscal 2002 were \$21.9 million compared to \$28.8 million in the same period of fiscal 2001. The \$6.9 million decrease is primarily the result of decreased production taxes in 2002 due to significantly lower commodity prices in the first nine months of fiscal 2002 compared to the same period in 2001.

Depreciation, depletion and amortization expense was \$30.7 million for the first nine months of fiscal 2002 compared to \$26.1 million in the same period of 2001. Impairment charges for producing properties of \$7.9 million and \$4.5 million were included in these respective amounts. After-tax, the impairment charge reduced net income in 2002 and 2001 by approximately \$4.9 million (\$0.10 per share) and \$2.8 million (\$0.05 per share), respectively. The remaining increase of \$1.2 million is due to higher depreciation rates as the result of lower reserves at September 30, 2001.

NATURAL GAS MARKETING segment reported an operating profit of \$1.3 million for the first nine months of fiscal 2002 compared to \$4.8 million in the same period of fiscal 2001. The significant decrease is the result of very favorable spot market prices in the first quarter of fiscal 2001, as gas prices were increasing dramatically. Those same conditions did not occur during the first nine months of fiscal 2002.

DOMESTIC DRILLING

Domestic Drilling's operating profit was \$55.7 million and \$68.0 million for the first nine months of fiscal 2002 and fiscal 2001, respectively. Revenues for the same periods were \$251.4 million and \$228.8, respectively. The decrease in operating profit is due

June 30, 2002 (Continued)

primarily to reduced performance in land operations, with average revenue per day of approximately \$12,600 in the first nine months of fiscal 2002, compared with \$15,420 in the same period of fiscal 2001. Land rig utilization was 83% and 95% for the first nine months of fiscal 2002 and 2001, respectively. Operating profit from offshore operations was down 3% for the first nine months of fiscal 2002, compared to the same period in fiscal 2001. Offshore rig utilization was 90% and 97% for the same periods, respectively.

Although land dayrates are still soft and difficult to predict in future months, there are signs that they are stabilizing, if commodity prices stay at present levels. Rig utilization for both land and offshore operations should remain at third quarter levels in the fourth quarter, as the two new platform rigs 205 and 206 will work a full quarter.

INTERNATIONAL DRILLING

International Drilling's operating profit decreased to \$11.8 million from \$19.3 million. Revenues decreased to \$109.6 from \$114.3 million. The decrease in operating profit is due primarily to a 15% reduction in revenue days and currency devaluation losses in Venezuela and Argentina. Operating profit in Colombia decreased \$4.9 million in the first nine months of fiscal 2002 compared to the same period of 2001 as there was one rig working in 2002 and an average of 3.4 rigs in 2001. Equatorial Guinea operating profit decreased \$1.2 million due to an increase in operating expenses. Operating profit in Ecuador increased \$4.6 million for the first nine months of 2002 compared to the same period in 2001 as revenue per day increased by 25%.

In Argentina, operating profit decreased \$2.3 million in 2002 compared with 2001 as the average rigs working decreased from 2.8 rigs in the first nine months of 2001 to 1.3 rigs in 2002. Argentina also recorded a currency devaluation loss of \$1.2 million in the first nine months of fiscal 2002 due to devaluation of the Argentina peso. Although some stability has been achieved, there is still significant uncertainty regarding economic, banking and currency stability.

Operations in Venezuela incurred currency valuation losses in the first nine months of fiscal 2002 of \$3.1 million compared to \$0.6 million in the same period of fiscal 2001. The severe decline in the value of the Venezuelan Bolivar relative to the U.S. dollar is the result of a change in the government's exchange policy and general instability in the country. Should additional devaluation of the bolivar occur, the Company could be exposed to additional currency losses of between \$0.5 million and \$1.4 million during the remainder of fiscal 2002.

OTHER

Other revenues increased approximately \$17 million in the first nine months of 2002 compared to the same period of 2001. Gains from the sale of available-for-sale securities in the first nine months of 2002 were \$25.4 million compared to \$2.6 million in the same period of 2001. Dividend income decreased from \$3.1 million in 2001 to \$1.8 million in 2002 as the result of reduced equity holdings and lower money market investments in fiscal 2002. Interest income was \$1.3 million in the first nine months of fiscal 2002 compared to \$4.5 million in the same period of 2001. The decrease is the result of lower interest rates and decreased cash balances in fiscal 2002.

June 30, 2002 (Continued)

Interest expense for the first nine months of fiscal 2001 was a credit of \$1.0 million compared with an expense of \$32,000 in fiscal 2002. In the first nine months of fiscal 2002, interest expense included credits for capitalized interest and ad valorem tax settlements of \$1.8 million and \$0.3 million, respectively. For the same period of fiscal 2001, the credits were \$0.8 million and \$2.3 million, respectively.

General and administrative expense increased from \$11.7 million in the first nine months of fiscal 2001 to \$14.1 million in the same period of fiscal 2002. The increase is related to labor and employee benefits, advertising costs, professional services related to establishing the Company's Exploration & Production Division as a separate public entity, and pension expense.

The Company's effective income tax rate increased to 43.0% for the nine months compared to 39.9% for the first nine months of fiscal 2001. The increase is due primarily to certain costs and expenses related to foreign locations for which the Company does not receive a tax benefit, including currency devaluation losses.

LIQUIDITY AND CAPITAL RESOURCES

Net cash provided by operating activities was \$145.2 million for the first nine months of fiscal 2002, compared with \$211.3 million for the same period in 2001. The decrease in cash flows was primarily the result of significantly lower cash flow from the Exploration and Production segment due to decreased commodity prices. Capital expenditures were \$257.5 million and \$186.7 million for the first nine months of 2002 and 2001, respectively. The increase is primarily the result of new rig construction in the Company's U.S. land operations.

The Company anticipates capital expenditures to be approximately \$350 million for fiscal 2002. Internally generated cash flows are projected to be approximately \$200 million for fiscal 2002 and cash balances were \$48 million at June 30, 2002. The Company's indebtedness totaled \$50 million as of June 30, 2002, as described in Note 7 to the Consolidated Financial Statements. To help finance the future capital expenditure requirements, the Company is currently negotiating a \$200 million intermediate term debt facility. This facility will provide for staged maturities from 5 to 12 years. The Company expects to finalize this facility in mid-August 2002. To provide short-term flexibility, the Company increased its revolving bank lines of credit to a total of \$175 million. (see Note 7)

In the third quarter of fiscal 2002, the Company sold its remaining shares in Bank One Corporation (175,000 shares), Kerr McGee (150,000 shares), Oneok (450,000 shares) and Sun Co. (312,546 shares) for total proceeds of approximately \$36.8 million. The Company's remaining stock portfolio has a market value of approximately \$180 million.

There were no other significant changes in the Company's financial position since September 30, 2001.

PART II. OTHER INFORMATION HELMERICH & PAYNE, INC.

June 30, 2002

Item 3. QUANTITATIVE AND QUALITATIVE DISCLOSURE ABOUT MARKET RISK

For a description of the Company's market risks, see "Item 7 (a). Quantitative and Qualitative Disclosures About Market Risk" in the Company's Annual Report on Form 10-K for the fiscal year ended September 30, 2001, and Note 10 to the Consolidated condensed Financial Statements contained in Part I hereof.

PART II. OTHER INFORMATION

Item 1. LEGAL PROCEEDINGS

The discussion of legal proceedings under the heading "Litigation Settlement" as disclosed in Note 12 to the Consolidated Condensed Financial Statements contained in Part I hereof is hereby incorporated by reference.

Item 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) Exhibits

The following documents are included as exhibits to this Form 10-Q. Those exhibits below incorporated by reference herein are indicated as such. If not so indicated, such exhibits are filed herewith.

Exhibit	
Number	Description
10.1	Form of Director Nonqualified Stock Option Agreement Under 2000 Helmerich & Payne, Inc. Stock Incentive Plan
10.2	Form of Change of Control Agreement for Helmerich & Payne Exploration and Production Employees
10.3	Form of Change of Control Agreement for Helmerich & Payne Drilling Employees
10.4	Second Amendment to Credit Agreement, dated as of July 16, 2002, by and among Helmerich & Payne International Drilling Co., Helmerich & Payne, Inc., and Bank One, Oklahoma, N.A.
10.5	Credit Agreement, dated as of July 16, 2002, among Helmerich & Payne International Drilling Co., Helmerich & Payne, Inc., The Several Lenders From Time to Time Party Hereto, and Bank of Oklahoma, National Association

PART II. OTHER INFORMATION HELMERICH & PAYNE, INC.

June 30, 2002 (continued)

(b) Reports on Form 8-K

For the three months ended June 30, 2002, registrant furnished, on April 24, 2002, one form 8-K reporting under Item 9, Regulation FD Disclosure, attaching a press release announcing results of operations and certain supplemental information, including financial statements.

SIGNATURES

HELMERICH & PAYNE, INC.

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Certification of CEO and CFO Pursuant to 18 U.S.C. Section 1350, As Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report of Helmerich & Payne, Inc. (the "Company") on Form 10-Q for the period ending June 30, 2002 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Hans Helmerich, as Chief Executive Officer of the Company, and Douglas E. Fears, as Chief Financial Officer of the Company, each hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, to the best of his knowledge, that:

- (1) The Report fully complies with the requirements of section 13(a) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and result of operations of the Company.

/s/ DOUGLAS E. FEARS
-----Douglas E. Fears
Chief Financial Officer
August 14, 2002

EXHIBIT INDEX

The following documents are included as exhibits to this Form 10-Q. Those exhibits below incorporated by reference herein are indicated as such. If not so indicated, such exhibits are filed herewith.

Exhibit Number	Description
10.1	Form of Director Nonqualified Stock Option Agreement Under 2000 Helmerich & Payne, Inc. Stock Incentive Plan
10.2	Form of Change of Control Agreement for Helmerich & Payne Exploration and Production Employees
10.3	Form of Change of Control Agreement for Helmerich & Payne Drilling Employees
10.4	Second Amendment to Credit Agreement, dated as of July 16, 2002, by and among Helmerich & Payne International Drilling Co., Helmerich & Payne, Inc., and Bank One, Oklahoma, N.A.
10.5	Credit Agreement, dated as of July 16, 2002, among Helmerich & Payne International Drilling Co., Helmerich & Payne, Inc., The Several Lenders From Time to Time Party Hereto, and Bank of Oklahoma, National Association

Exhibit 10.1

DIRECTOR NONQUALIFIED STOCK OPTION AGREEMENT UNDER 2000 HELMERICH & PAYNE, INC. STOCK INCENTIVE PLAN

THIS DIRECTOR NONQUALIFIED STOCK OPTION AGREEMENT (the	"Option Agreement"), made as of the day of	,
2002 (the "Date of Grant") at Tulsa, Oklahoma by and between	(the "Participant") and Helmerich & Payne, Inc. (the	;
"Company"):		

WITNESSETH:

WHEREAS, Participant is a director of the Company, and it is important to the Company that Participant be encouraged to remain in the service of the Company; and

WHEREAS, in recognition of such facts, the Company desires to provide to Participant an opportunity to purchase shares of the Common Stock of the Company, as hereinafter provided, pursuant to the "Helmerich & Payne, Inc. 2000 Stock Incentive Plan" (the "Plan"), a copy of which has been provided to Participant; and

WHEREAS, any capitalized terms used but not defined herein have the same meanings given them in the Plan.

NOW, THEREFORE, in consideration of the mutual covenants hereinafter set forth and for good and valuable consideration, Participant and the Company hereby agree as follows:

SECTION 1. Grant of Stock Option. The Company hereby grants to Participant a nonqualified stock option (the "Stock Option") to purchase all or any part of ______ shares of its Common Stock, par value \$.10 (the "Stock"), under and subject to the terms and conditions of this Option Agreement and the Plan which is incorporated herein by reference and made a part hereof for all purposes. The purchase price for each share to be purchased hereunder shall be ______ (the "Option Price") which equals the Fair Market Value of the Common Stock covered by this Stock Option on the Date of Grant.

SECTION 2. Vesting. Subject to the applicable provisions of the Plan and this Option Agreement, Participant's Stock Option shall be fully vested and immediately exercisable on the Date of Grant.

SECTION 3. Term of Stock Option. Subject to earlier termination as provided in this Option Agreement or in the Plan, the Stock Option shall expire at the close of business ten years from the Date of Grant and may not be exercised after such expiration date.

SECTION 4. Transferability of Stock Option.

- (a) General. Except as provided in Section 4(b) hereof, the Stock Option shall not be transferable otherwise than by will or the laws of descent and distribution, and the Stock Option may be exercised, during the lifetime of Participant, only by Participant. More particularly (but without limiting the generality of the foregoing), the Stock Option may not be assigned, transferred (except as provided above and in Section 4(b) hereof), pledged or hypothecated in any way, shall not be assignable by operation of law and shall not be subject to execution, attachment, or similar process. Any attempted assignment, transfer, pledge, hypothecation or other disposition of the Stock Option contrary to the provisions hereof shall be null and void and without effect.
- (b) Limited Transferability of Stock Options. The Stock Options may be transferred by Participant to (i) the ex-spouse of Participant pursuant to the terms of a domestic relations order, (ii) the spouse, children or grandchildren of Participant ("Immediate Family Members"), (iii) a trust or trusts for the exclusive benefit of such Immediate Family Members, or (iv) a partnership in which such Immediate Family Members are the only partners; provided that there may be no consideration for any such transfer and subsequent transferrs of transferred Transferred Stock Options shall continue to be subject to the same terms and conditions as were applicable immediately prior to transfer, provided that for purposes of this Section 4(b) the term "Participant" shall

be deemed to refer to the transferee. No transfer pursuant to this Section 4(b) shall be effective to bind the Company unless the Company shall have been furnished with written notice of such transfer together with such other documents regarding the transfer as the Committee shall request.

SECTION 5. Timing of Exercise Upon Termination of Service. Upon Participant's termination of service as a director of the Company, Participant, or the representative of a deceased Participant, shall be entitled to exercise the Stock Option during the remaining term of the Stock Option.

SECTION 6. Method of Exercising Stock Option.

- (a) Procedures for Exercise. The manner of exercising the Stock Option herein granted shall be by written notice to the Secretary of the Company at the time the Stock Option, or part thereof, is to be exercised, and in any event prior to the expiration of the Stock Option. Such notice shall state the election to exercise the Stock Option, the number of shares of Stock to be purchased upon exercise, the form of payment to be used, and shall be signed by the person so exercising the Stock Option.
- (b) Form of Payment. Payment in full for shares of Stock purchased under this Option Agreement shall accompany Participant's notice of exercise, together with payment for any applicable withholding taxes. Payment shall be made (i) in cash or by check, draft or money order payable to the order of the Company; (ii) by delivering Stock having a Fair Market Value on the date of payment equal to the amount of the Option Price but only to the extent such exercise of an Option would not result in an accounting charge with respect to the use of such shares to pay the Option Price; or (iii) a combination thereof. In addition to the foregoing procedure which may be available for the exercise of the Stock Option, Participant may deliver to the Company a notice of exercise which includes an irrevocable instruction to the Company to deliver the Stock certificate representing the shares of Stock being purchased, issued in the name of Participant, to a broker approved by the Company and authorized to trade in the Common Stock of the Company. Upon receipt of such notice, the Company shall acknowledge receipt of the executed notice of exercise and forward this notice to the broker. Upon receipt of the copy of the notice which has been acknowledged by the Company, and without waiting for issuance of the actual Stock certificate with respect to the exercise of the Stock Option, the broker may sell the Stock or any portion thereof. The broker shall deliver directly to the Company that portion of the sales proceeds sufficient to cover the Option Price and withholding taxes, if any. For all purposes of effecting the exercise of the Stock Option, the date on which Participant gives the notice of exercise to the Company, together with payment for the shares of Stock being purchased and any applicable withholding taxes, shall be the "date of exercise to the Company, together with payment for the shares of Stock being purchased and any applicable withholding taxes, shall be the "date of exercise and full payment as provided herein.
- (c) Further Information. In the event the Stock Option is exercised, pursuant to the foregoing provisions of this Section 6, by any person due to the death of Participant, such notice shall also be accompanied by appropriate proof of the right of such person to exercise the Stock Option. The notice so required shall be given by personal delivery to the Secretary of the Company or by registered or certified mail, addressed to the Secretary of the Company at Utica at Twenty-First, Tulsa, Oklahoma 74114, and it shall be deemed to have been given when it is so personally delivered or when it is deposited in the United States mail in an envelope addressed to the Company, as aforesaid, properly stamped for delivery as a registered or certified letter.

SECTION 7. Securities Law Restrictions. The Stock Option shall be exercised and Stock issued only upon compliance with the Securities Act of 1933, as amended (the "Act"), and any other applicable securities law, or pursuant to an exemption therefrom. If deemed necessary by the Company to comply with the Act or any applicable laws or regulations relating to the sale of securities, Participant, at the time of exercise and as a condition imposed by the Company, shall represent, warrant and agree that the shares of Stock subject to the Stock Option are being purchased for investment and not with any present intention to resell the same and without a view to distribution, and Participant shall, upon the request of the Company, execute and deliver to the Company an agreement to such effect. Participant acknowledges that any Stock certificate representing Stock purchased under such circumstances will be issued with a restricted securities legend.

SECTION 8. Notices. All notices or other communications relating to the Plan and this Option Agreement as it relates to Participant shall be in writing and shall be delivered personally or mailed (U.S. Mail) by the Company to Participant at the then current address as maintained by the Company or such other address as Participant may advise the Company in writing.

SECTION 9. Conflicts. In the event of any conflicts between this Option Agreement and the Plan, the Plan shall control. In the event any provision hereof conflicts with applicable law, that provision shall be severed and the remaining provisions shall remain enforceable.

SECTION 10. No Part of Other Plans. The benefits provided under this Agreement or the Plan shall not be deemed to be a part of or considered in the calculation of any other benefit provided by the Company, a Subsidiary or an Affiliated Entity to Participant.

SECTION 11. Participant and Award Subject to Plan. As specific consideration to the Company for the Award, Participant agrees to be bound by the terms of the Plan and this Agreement.

IN WITNESS WHEREOF, the parties have executed this Option Agreement as of the day and year first above written.

HELMERICH & PAYNE, INC., a Delaware corporation

By:

"C	OMPANY''	

"PARTICIPANT"

Exhibit 10.2

CHANGE OF CONTROL AGREEMENT

THIS CHANGE OF CONTROL AGREEMENT (the "	'Agreement") entered into between HELMERICH & PA	YNE, INC., a Delaware	
corporation ("Helmerich & Payne"), and	, an individual (the "Executive"), dated as of the	day of	.,
2002			

The Board of Directors of Helmerich & Payne (the "Board") has determined that it is in the best interests of the Company and its shareholders to assure that the Company will have the continued dedication of the Executive, notwithstanding the possibility, threat, or occurrence of a "Change of Control" (as defined in Section 2 of this Agreement) of the Company. The Board believes it is important to diminish the inevitable distraction of the Executive by virtue of the personal uncertainties and risks created by a pending or threatened Change of Control, and to encourage the Executive's full attention and dedication to the affairs of the Company during the term of this Agreement and upon the occurrence of such event. The Board also believes the Company is best served by providing the Executive with compensation arrangements upon a Change of Control which provide the Executive with individual financial security and which are competitive with those of other corporations. In order to accomplish these objectives, the Board has caused Helmerich & Payne to enter into this Agreement. For the purposes of Section 2 of this Agreement, "Company" means Helmerich & Payne; and, for all other purposes in this Agreement, "Company" means Helmerich & Payne and any of its subsidiaries as defined in the Helmerich & Payne, Inc. E&P Severance Plan (the "Severance Plan"); provided, however, in the event of the sale or distribution of more than 50% of the operating assets of the E&P Division (as defined in Section 2(e) of this Agreement), and the successor to such assets (the "E&P Successor") assumes the obligations of the Company under this Agreement, the term "Company" shall thereafter mean the E&P Successor where applicable.

NOW, THEREFORE, IT IS HEREBY AGREED AS FOLLOWS:

- 1. Certain Definitions.
- (a) The "Effective Date" shall be the first date during the "Change of Control Period" (as defined in Section 1(b) of this Agreement) on which a Change of Control (as defined below) occurs, and, except as provided in the following sentence, no amount shall be paid or benefits provided under this Agreement if the Executive's employment is terminated for any reason prior to a Change of Control.
- (b) The "Change of Control Period" is the period commencing on the date hereof and ending on the earlier to occur of (i) the second anniversary of the Effective Date or (ii) the first day of the month next following the Executive's attainment of age 65 ("Normal Retirement Date").
- 2. Change of Control. For the purpose of this Agreement, a "Change of Control" shall mean:

- (a) The acquisition by any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) (a "Person") of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of 15% or more of either (i) the then outstanding shares of common stock of the Company (the "Outstanding Company Common Stock") or (ii) the combined voting power of the then outstanding voting securities of the Company entitled to vote generally in the election of directors (the "Outstanding Company Voting Securities"); provided, however, that the following acquisitions shall not constitute a Change of Control: (i) any acquisition directly from the Company, (ii) any acquisition by the Company, or (iii) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Company or any corporation controlled by the Company; or
- (b) Individuals who, as of the date hereof, constitute the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board; provided, however, that any individual becoming a director subsequent to the date hereof whose election, appointment or nomination for election by the Company's shareholders was approved by a vote of at least a majority of the directors then comprising the Incumbent Board shall be considered as though such individual were a member of the Incumbent Board, but excluding, for purposes of this definition, any such individual whose initial assumption of office occurs as a result of an actual or threatened election contest with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board; or
- (c) Approval by the shareholders of the Company of a reorganization, share exchange, merger or consolidation or acquisition of assets of another corporation (a "Business Combination"), in each case, unless, following such Business Combination, (i) all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Company Common Stock and Outstanding Company Voting Securities immediately prior to such Business Combination will beneficially own, directly or indirectly, more than 70% of, respectively, the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, as the case may be, of the corporation resulting from such Business Combination (including, without limitation, a corporation which as a result of such transaction will own the Company through one or more subsidiaries) in substantially the same proportions as their ownership immediately prior to such Business Combination of the Outstanding Company Common Stock and Outstanding Company Voting Securities, as the case may be, (ii) no Person (excluding any employee benefit plan (or related trust) of the Company or such corporation resulting from such Business Combination) will beneficially own, directly or indirectly, 15% or more of, respectively, the then outstanding shares of common stock of the corporation resulting from such Business Combination or the combined voting power of the then outstanding voting securities of such corporation, except to the extent that such ownership existed prior to the Business Combination, and (iii) at least a majority of the members of the board of directors of the corporation resulting from such Business Combination were members of the Incumbent Board at the time of the execution of the initial agreement, or of the action of the Board, providing for such Business Combination or were elected, appointed or nominated by the Board; or

- (d) Approval by the shareholders of the Company of (i) a complete liquidation or dissolution of the Company or, (ii) the sale or other disposition of all or substantially all of the assets of the Company, other than to a corporation, with respect to which following such sale or other disposition (A) more than 70% of, respectively, the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding voting securities of such corporation entitled to vote generally in the election of directors is then beneficially owned, directly or indirectly, by all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Company Common Stock and Outstanding Company Voting Securities immediately prior to such sale or other disposition in substantially the same proportion as their ownership immediately prior to such sale or other disposition, of the Outstanding Company Common Stock and Outstanding Company Voting Securities, as the case may be; (B) less than 15% of, respectively, the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding voting securities of such corporation entitled to vote generally in the election of directors will be beneficially owned, directly or indirectly, by any Person (excluding any employee benefit plan (or related trust) of the Company or such corporation), except to the extent that such Person owned 15% or more of the Outstanding Company Common Stock or Outstanding Company Voting Securities prior to the sale or disposition; and (C) at least a majority of the members of the board of directors of such corporation were members of the Incumbent Board at the time of the execution of the initial agreement, or of the action of the Board, providing for such sale or other disposition of assets of the Company or were elected, appointed or nominated by the Board.
- (e) The foregoing Sections (a) -- (d) notwithstanding, a Change of Control shall also be deemed to have occurred upon the occurrence of a business transaction (or a series of transactions) involving the direct or indirect transfer or disposition (whether by sale, merger, reorganization, spin-off, stock dividend, stock split or otherwise) to the E&P Successor of more than 50% of the operating assets of the Company's oil and gas exploration and production division (the "E&P Division"), if after such transaction, the E&P Successor holds more than 50% of the operating assets of the E&P Division, and the Company or a "subsidiary" (as defined in the Severance Plan and as determined immediately prior to such transaction) owns less than 70% of the outstanding shares of the voting securities of the E&P Successor (if a corporation), or less than a 70% interest in the profits or assets of the E&P Successor (if other than a corporation).
- 3. Agreement Not Employment Contract. This Agreement shall be considered solely as a "severance agreement" obligating the Company to pay to the Executive certain amounts of compensation in the event and only in the event of his termination of employment after the Effective Date for the reasons and at the time specified herein. Apart from the obligation of the Company to provide the amounts of additional compensation as provided in this Agreement, the Company shall at all times retain the right to terminate the employment of the Executive since the obligation of the Company to the Executive shall only be considered as an employment relationship which exists between the Company and the Executive which may be terminated at will by either party subject to the obligation of the Company to make payment and perform its obligations as provided in this Agreement.

4. Termination.

- (a) Death or Disability. This Agreement shall terminate automatically upon the Executive's death. If the Company determines in good faith that the Disability of the Executive has occurred (pursuant to the definition of "Disability" set forth below), it may give to the Executive written notice of its intention to terminate the Executive's employment. In such event, the Executive's employment with the Company shall terminate effective on the 30th day after the date of such notice (the "Disability Effective Date"), provided that, within such time period, the Executive shall not have returned to full-time performance of the Executive's duties. For purposes of this Agreement, "Disability" means disability (either physical or mental) which, at least twenty-six (26) weeks after its commencement, is determined by a physician selected by the Company or its insurers and acceptable to the Executive or the Executive's legal representative to be total and permanent (such agreement as to acceptability not to be withheld unreasonably).
- (b) Cause. The Company may terminate the Executive's employment for "Cause." For purposes of this Agreement, termination of the Executive's employment by the Company for Cause shall mean termination for one of the following reasons: (i) the conviction of the Executive of a felony by a federal or state court of competent jurisdiction; (ii) an act or acts of dishonesty taken by the Executive and intended to result in substantial personal enrichment of the Executive at the expense of the Company; or (iii) the Executive's "willful" failure to follow a direct, reasonable and lawful written order from his supervisor, within the reasonable scope of the Executive's duties, which failure is not cured within thirty (30) days. Further, for purposes of this Section (b):
- (1) No act or failure to act, on the Executive's part shall be deemed "willful" unless done, or omitted to be done, by the Executive not in good faith and without reasonable belief that the Executive's action or omission was in the best interest of the Company.
- (2) The Executive shall not be deemed to have been terminated for Cause unless and until there shall have been delivered to the Executive a copy of a resolution duly adopted by the affirmative vote of not less than three-fourths (3/4ths) of the entire membership of the Board at a meeting of the Board called and held for such purpose (after reasonable notice to the Executive and an opportunity for the Executive, together with the Executive's counsel, to be heard before the Board), finding that in the good faith opinion of the Board the Executive was guilty of conduct set forth in clauses (i), (ii) or (iii) above and specifying the particulars thereof in detail.
- (c) Substantial Downturn. The Company may terminate the Executive's employment after the occurrence of a "Substantial Downturn" in the oil and gas industry. For purposes of this Agreement, Substantial Downturn shall mean a severe downturn in the oil and gas industry which shall be measured by the following objective criteria:
- (i) The West Texas Intermediate Price for Crude Oil remains at or below \$10/barrel for sixty (60) consecutive business days, or

- (ii) The price for each MMBtu of natural gas as quoted for the Henry Hub listing in "Gas Daily" remains at or below \$1.25 for sixty (60) consecutive business days.
- (d) Good Reason. The Executive's employment may be terminated by the Executive for Good Reason. For purposes of this Agreement, "Good Reason" means:
- (i) Any reduction in the Executive's (x) annual base salary ("Base Salary") in effect immediately prior to the Effective Date or (y) Annual Bonus. For purposes of this Agreement, "Annual Bonus" shall mean the amount equal to the average of the annual bonus paid to the Executive by the Company during the two years immediately preceding the year in which the Effective Date occurs.
- (ii) The Company's requiring the Executive to be based at any office or location that is more than 25 miles from the office or location at which the Executive is based on the Effective Date except for periodic travel reasonably required in the performance of the Executive's responsibilities.
- (iii) Any purported termination by the Company of the Executive's employment otherwise than as expressly permitted by this Agreement.
- (iv) Any failure by the Company to comply with and satisfy Section 10(c) of this Agreement.
- (e) Notice of Termination. Any termination by the Company for Cause or after the occurrence of a Substantial Downturn, or by the Executive for Good Reason, shall be communicated by Notice of Termination to the other party hereto given in accordance with Section 13(b) of this Agreement. For purposes of this Agreement, a "Notice of Termination" means a written notice which (i) indicates the specific termination provisions in this Agreement relied upon, (ii) sets forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of the Executive's employment under the provision so indicated, and (iii) if the Date of Termination (as defined below) is other than the date of receipt of such notice, specifies the termination date (which date shall be not more than 15 days after the giving of such notice). The failure by the Executive to set forth in the Notice of Termination any fact or circumstance which contributes to a showing of Good Reason shall not waive any right of the Executive hereunder or preclude the Executive from asserting such fact or circumstance in enforcing his rights hereunder.
- (f) Date of Termination. "Date of Termination" means the date of receipt of the Notice of Termination by either the Company or the Executive as the case may be or any later date specified therein; provided, however, if the Executive's employment is terminated by reason of death or Disability, the Date of Termination shall be the date of death of the Executive or the effective date of Disability, as the case may be.
- 5. Obligations of the Company upon Termination Following Change of Control.

- (a) Good Reason; Termination Other Than for Cause, Disability, Substantial Downturn or Death. If, within 24 months after the Effective Date, the Company terminates the Executive's employment other than for Cause, Disability, the occurrence of a Substantial Downturn, or death, or if the Executive terminates his employment for Good Reason:
- (i) The Company shall pay to the Executive in a lump sum in cash within 30 days after the Date of Termination the aggregate of the following amounts and the Company shall provide the following benefits:
- A. To the extent not theretofore paid, the Executive's Base Salary through the Date of Termination;
- B. The product of (i) the Annual Bonus and (ii) a fraction, the numerator of which is the number of days in the current fiscal year through the Date of Termination and the denominator of which is 365;
- C. The product obtained by multiplying two (2) times the sum of (i) the Base Salary and (ii) the Annual Bonus; and
- D. Any accrued vacation pay not yet paid by the Company.
- (ii) For the 24-month period following the Termination Date, or such longer period as any plan, program, practice or policy may provide, the Company shall continue benefits to the Executive and/or the Executive's family at least equal to those which would have been provided to them in accordance with the plans, programs, practices and policies, including health insurance and life insurance of the Company during the 90-day period immediately preceding the Date of Termination. In the event the Company is unable to provide such benefits due to restrictions imposed by the Internal Revenue Code (the "Code"), ERISA, or such other applicable law, the Company will provide Executive with an alternative equivalent benefit.
- (iii) The Company shall pay up to \$5,000 in outplacement counseling services utilized by the Executive.
- (iv) In the event of a Change of Control pursuant to Section 2(e), any options to purchase stock or rights to receive restricted stock granted under a plan adopted by the E&P Successor and held by the Executive shall be immediately and automatically vested, fully earned and exercisable upon the Date of Termination unless previously exercised or forfeited.
- (b) Limitations on Company Obligations. Notwithstanding the foregoing, the Executive's termination of employment with Helmerich & Payne resulting from the creation and/or spin-off of the E&P Successor (including the subsequent merger of the E&P Successor with another entity) shall not constitute a termination of employment for purposes of this Agreement.

- 6. Non-Exclusivity of Rights. Nothing in this Agreement shall prevent or limit the Executive's continuing or future participation in any benefit, bonus, incentive or other plans, programs, policies or practices, provided by the Company and for which the Executive may qualify, nor shall anything herein limit or otherwise affect such rights as the Executive may have under any stock option or other agreements with the Company. Amounts which are vested benefits or which the Executive is otherwise entitled to receive under any plan, policy, practice or program of the Company at or subsequent to the Date of Termination shall be payable in accordance with such plan, policy, practice or program.
- 7. Full Settlement. The Company's obligation to make the payments provided for in this Agreement and otherwise to perform its obligations hereunder shall not be affected by any set-off, counterclaim, recoupment, defense or other claim, right or action which the Company may have against the Executive or others. In no event shall the Executive be obligated to seek other employment or take any other action by way of mitigation of the amounts payable to the Executive under any of the provisions of this Agreement. The Company agrees to pay, to the full extent permitted by law, up to \$50,000 in legal fees and all expenses which the Executive may reasonably incur as a result of any contest (regardless of the outcome thereof) by the Company or others of the validity or enforceability of, or liability under, any provision of this Agreement or any guarantee of performance thereof (including as a result of any contest by the Executive about the amount of any payment pursuant to Section 8 of this Agreement), plus in each case interest at the applicable Federal rate provided for in Section 7872(f)(2) of the Code.
- 8. Maximum Payments. It is the objective of this Agreement to maximize the Executive's Net After-Tax Benefit (as defined herein) if payments or benefits provided under this Agreement are subject to excise tax under Section 4999 of the Code. Therefore, in the event it is determined that any payment or benefit by the Company or otherwise to or for the benefit of the Executive, whether paid or payable or distributed or distributable pursuant to the terms of this Agreement or otherwise, including, by example and not by way of limitation, acceleration by the Company or otherwise of the date of vesting or payment or rate of payment under any plan, program, arrangement or agreement of the Company, would be subject to the excise tax imposed by Section 4999 of the Code or any interest or penalties with respect to such excise tax (such excise tax, together with any such interest and penalties, are hereinafter collectively referred to as the "Excise Tax"), then the Company shall first make a calculation under which such payments or benefits provided to the Executive under this Agreement are reduced to the extent necessary so that no portion thereof shall be subject to the excise tax imposed by Section 4999 of the Code (the "4999 Limit"). The Company shall then compare (x) the Executive's Net After-Tax Benefit assuming application of the 4999 Limit with (y) the Executive's Net After-Tax Benefit without the application of the 4999 Limit and the Executive shall be entitled to the greater of (x) or (y). "Net After-Tax Benefit" shall mean the sum of (i) all payments and benefits which the Executive receives or is then entitled to receive from the Company, less (ii) the amount of federal income taxes payable with respect to the payments and benefits described in (i) above calculated at the maximum marginal income tax rate for each year in which such payments and benefits shall be paid to the Executive (based upon the rate for such year as set forth in the Code at the time of the first payment of the foregoing), less (iii) the amount of excise taxes imposed with respect to the payments and benefits described in (i) above by Section 4999 of the Code. The determination of whether a payment or benefit constitutes an excess parachute payment shall be made by tax

counsel selected by the Company and reasonably acceptable to the Executive. The costs of obtaining this determination shall be borne by the Company.

- 9. Confidential Information. The Executive shall hold in a fiduciary capacity for the benefit of the Company all secret or confidential information, knowledge or data relating to the Company, and respective businesses, which shall have been obtained by the Executive during the Executive's employment by the Company and which shall not be or become public knowledge (other than by acts by the Executive or his representatives in violation of this Agreement). After termination of the Executive's employment with the Company, the Executive shall not, without the prior written consent of the Company, communicate or divulge any such information, knowledge or data to anyone other than the Company and those designated by it.
- 10. Successors.
- (a) This Agreement is personal to the Executive and without the prior written consent of the Company shall not be assignable by the Executive otherwise than by will or the laws of descent and distribution. This Agreement shall inure to the benefit of and be enforceable by the Executive's legal representatives.
- (b) This Agreement shall inure to the benefit of and be binding upon the Company and its successors and assigns.
- (c) The Company will require any successor (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business and/or assets of the Company to assume expressly and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform it if no such succession had taken place. As used in this Agreement, "Company" shall mean the Company as hereinbefore defined and any successor to all or a portion of its business and/or assets which assumes and agrees to perform this Agreement by operation of law, or otherwise.
- 11. Release. The Executive agrees, if his employment is terminated under circumstances entitling Executive to payments under Section 5(a)(i) of this Agreement that, in consideration for the payments described in Section 5(a)(i), he will execute a General Release in substantially the form of Exhibit A attached hereto, through which the Executive releases the Company from any and all claims as may relate to or arise out of his employment relationship (excluding claims the Executive may have under any "employee pension plan" as described in Section 3(3) of ERISA or any claims under this Agreement). The form of the release may be modified as needed to reflect changes in the applicable law or regulations that are needed to provide a legally enforceable and binding release to the Company at the time of execution.
- 12. Indemnification and Insurance. The Executive shall be indemnified and held harmless by the Company during the term of this Agreement and following any termination of this Agreement for any reason whatsoever in the same manner as would any other key management employee of the Company with respect to acts or omissions occurring prior to (a) the termination of this Agreement or (b) the termination of employment of the Executive. In addition, during the term of this Agreement and for a period of five years following the termination of this Agreement for any reason whatsoever, the Executive shall be covered by a

Company held Directors and Officers liability insurance policy covering acts or omissions occurring prior to (a) the termination of this Agreement or (b) the termination of employment of the Executive. Provided, in no event will the obligation of the Company to indemnify the Executive or provide Directors and Officers insurance to the Executive under this Section 12 be less than the obligation and insurance coverage which the Company provided to the Executive immediately prior to the occurrence of a Change of Control.

13. Miscellaneous.

- (a) This Agreement shall be governed by and construed in accordance with the laws of the State of Oklahoma, without reference to principles of conflict of laws. The captions of this Agreement are not part of the provisions hereof and shall have no force or effect. This Agreement may not be amended or modified otherwise than by a written agreement executed by the parties hereto or their respective successors and legal representatives.
- (b) All notices and other communications hereunder shall be in writing and shall be given by hand delivery to the other party or by registered or certified mail, return receipt requested, postage prepaid, addressed as follows:

If to the Executive:

At his last known address evidenced on the Company's payroll records

If to the Company: Helmerich & Payne, Inc. Utica at Twenty-First Tulsa, Oklahoma 74114 Attention: General Counsel

or to such other address as either party shall have furnished to the other in writing in accordance herewith. Notice and communications shall be effective when actually received by the addressee.

- (c) The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement.
- (d) The Company may withhold from any amounts payable under this Agreement such federal, state or local taxes as shall be required to be withheld pursuant to any applicable law or regulation.
- (e) The Executive's failure to insist upon strict compliance with any provision hereof shall not be deemed to be a waiver of such provision or any other provision thereof.
- (f) This Agreement contains the entire understanding of the Company and the Executive with respect to the subject matter hereof.

- (g) THE EXECUTIVE AND THE COMPANY ACKNOWLEDGE THAT THE EMPLOYMENT OF THE EXECUTIVE BY THE COMPANY IS "AT WILL," AND MAY BE TERMINATED BY EITHER THE EXECUTIVE OR THE COMPANY AT ANY TIME, SUBJECT TO THE COMPANY'S OBLIGATION TO PROVIDE ADDITIONAL COMPENSATION AS PROVIDED IN THIS AGREEMENT. UPON A TERMINATION OF THE EXECUTIVE'S EMPLOYMENT PRIOR TO THE EFFECTIVE DATE, THERE SHALL BE NO FURTHER RIGHTS UNDER THIS AGREEMENT.
- 14. No Trust. No action under this Agreement by the Company or its Board of Directors shall be construed as creating a trust, escrow or other secured or segregated fund, in favor of the Executive or his beneficiary. The status of the Executive and his beneficiary with respect to any liabilities assumed by the Company hereunder shall be solely those of unsecured creditors of the Company. Any asset acquired or held by the Company in connection with liabilities assumed by it hereunder, shall not be deemed to be held under any trust, escrow or other secured or segregated fund for the benefit of the Executive or his beneficiary or to be security for the performance of the obligations of the Company, but shall be, and remain a general, unpledged, unrestricted asset of the Company at all times subject to the claims of general creditors of the Company.
- 15. No Assignability. Neither the Executive nor his beneficiary, nor any other person shall acquire any right to or interest in any payments payable under this Agreement, otherwise than by actual payment in accordance with the provisions of this Agreement, or have any power to transfer, assign, anticipate, pledge, mortgage or otherwise encumber, alienate or transfer any rights hereunder in advance of any of the payments to be made pursuant to this Agreement or any portion thereof which is expressly declared to be nonassignable and nontransferable. No right or benefit hereunder shall in any manner be liable for or subject to the debts, contracts, liabilities, or torts of the person entitled to such benefit.

IN WITNESS WHEREOF, the Executive has hereunto set his hand and, pursuant to the authorization from its Board of Directors, Helmerich & Payne has caused these presents to be executed in its name on its behalf, all as of the day and year first above written.

"EXECUTIVE"

HELMERICH & PAYNE, INC., a Delaware corporation

<u>By:</u>
"HELMERICH & PAYNE"

EXHIBIT A

NOTICE. VARIOUS LAWS, INCLUDING TITLE VII OF THE CIVIL RIGHTS ACT OF 1964, THE CIVIL RIGHTS ACT OF 1866, THE PREGNANCY DISCRIMINATION ACT OF 1978, THE EQUAL PAY ACT, THE CIVIL RIGHTS ACT OF 1991, THE AGE DISCRIMINATION IN EMPLOYMENT ACT, THE REHABILITATION ACT OF 1973, THE AMERICANS WITH DISABILITIES ACT, THE EMPLOYEE RETIREMENT INCOME SECURITY ACT AND THE VETERANS REEMPLOYMENT RIGHTS ACT (ALL AS AMENDED FROM TIME TO TIME), PROHIBIT EMPLOYMENT DISCRIMINATION BASED ON SEX, RACE, COLOR, NATIONAL ORIGIN, RELIGION, AGE, DISABILITY, ELIGIBILITY FOR COVERED EMPLOYEE BENEFITS AND VETERAN STATUS. YOU MAY ALSO HAVE RIGHTS UNDER LAWS SUCH AS THE OLDER WORKER BENEFIT PROTECTION ACT OF 1990, THE WORKER ADJUSTMENT AND RETRAINING ACT OF 1988, THE FAIR LABOR STANDARDS ACT, THE FAMILY AND MEDICAL LEAVE ACT, THE OCCUPATIONAL HEALTH AND SAFETY ACT AND OTHER FEDERAL, STATE AND/OR MUNICIPAL STATUTES, ORDERS OR REGULATIONS PERTAINING TO LABOR, EMPLOYMENT AND/OR EMPLOYEE BENEFITS. THESE LAWS ARE ENFORCED THROUGH THE UNITED STATES DEPARTMENT OF LABOR AND ITS AGENCIES, INCLUDING THE EQUAL EMPLOYMENT OPPORTUNITY COMMISSION (EEOC), AND VARIOUS STATE AND MUNICIPAL LABOR DEPARTMENTS, FAIR EMPLOYMENT BOARDS, HUMAN RIGHTS COMMISSIONS AND SIMILAR AGENCIES.

THIS GENERAL RELEASE IS BEING PROVIDED TO YOU IN CONNECTION WITH THE SPECIAL, INDIVIDUALIZED
SEVERANCE PACKAGE OUTLINED IN SECTION 5(A) OF YOUR CHANGE OF CONTROL AGREEMENT DATED,
2002 (THE "AGREEMENT"). THE FEDERAL OLDER WORKER BENEFIT PROTECTION ACT REQUIRES THAT YOU HAVE AT
LEAST TWENTY-ONE (21) DAYS, IF YOU WANT IT, TO CONSIDER WHETHER YOU WISH TO SIGN A RELEASE SUCH AS THIS
ONE IN CONNECTION WITH A SPECIAL, INDIVIDUALIZED SEVERANCE PACKAGE. YOU HAVE UNTIL THE CLOSE OF
BUSINESS TWENTY-ONE (21) DAYS FROM THE DATE YOU RECEIVE THIS GENERAL RELEASE TO MAKE YOUR DECISION.
YOU MAY NOT SIGN THIS GENERAL RELEASE UNTIL, AT THE EARLIEST, YOUR OFFICIAL DATE OF SEPARATION FROM
EMPLOYMENT,

BEFORE EXECUTING THIS GENERAL RELEASE YOU SHOULD REVIEW THESE DOCUMENTS CAREFULLY AND CONSULT WITH YOUR ATTORNEY.

YOU MAY REVOKE THIS GENERAL RELEASE WITHIN SEVEN (7) DAYS AFTER YOU SIGN IT AND IT SHALL NOT BECOME EFFECTIVE OR ENFORCEABLE UNTIL THAT REVOCATION PERIOD HAS EXPIRED. IF YOU DO NOT ACCEPT THE SEVERANCE PACKAGE AND SIGN AND RETURN THIS GENERAL RELEASE, OR IF YOU EXERCISE YOUR RIGHT TO REVOKE THE GENERAL RELEASE AFTER SIGNING IT, YOU WILL NOT BE ELIGIBLE FOR THE SPECIAL, INDIVIDUALIZED SEVERANCE PACKAGE. ANY REVOCATION MUST BE IN WRITING AND MUST BE RECEIVED BY HELMERICH & PAYNE, INC., ATTENTION: DIRECTOR, HUMAN RESOURCES, HELMERICH & PAYNE, INC., UTICA AT TWENTY-FIRST, TULSA, OK 74114, WITHIN THE SEVEN-DAY PERIOD FOLLOWING YOUR EXECUTION OF THIS GENERAL RELEASE.

GENERAL RELEASE

In consideration of the special, individualized severance package offered to me by Helmerich & Payne, Inc. and the separation benefits I will receive as reflected in Section 5(a) of my Change of Control Agreement dated _______, 2002 (the "Agreement"), I hereby release and discharge Helmerich & Payne, Inc. and its predecessors, successors, affiliates, parent, subsidiaries and partners and each of those entities' employees, officers, directors and agents (hereafter collectively referred to as the "Company") from all claims, liabilities, demands, and causes of action, known or unknown, fixed or contingent, which I may have or claim to have against the Company either as a result of my past employment with the Company and/or the severance of that relationship and/or otherwise, and hereby waive any and all rights I may have with respect to and promise not to file a lawsuit to assert any such claims.

This General Release includes, but is not limited to, claims arising under Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1866, the Pregnancy Discrimination Act of 1978, the Equal Pay Act, the Civil Rights Act of 1991, the Age Discrimination in Employment Act, the Rehabilitation Act of 1973, the Americans With Disabilities Act, the Employee Retirement Income Security Act or 1974 and the Veterans Reemployment Rights Act (all as amended from time to time). This General Release also includes, but is not limited to, any rights I may have under the Older Workers Benefit Protection Act of 1990, the Worker Adjustment and Retraining Act of 1988, the Fair Labor Standards Act, the Family and Medical Leave Act, the Occupational Health and Safety Act and any other federal, state and/or municipal statutes, orders or regulations pertaining to labor, employment and/or employee benefits. This General Release also applies to any claims or rights I may have growing out of any legal or equitable restrictions on the Company's rights not to continue an employment relationship with its employees, including any express or implied employment contracts, and to any claims I may have against the Company for fraudulent inducement or misrepresentation, defamation, wrongful termination or other retaliation claims in connection with workers' compensation or alleged "whistleblower" status or on any other basis whatsoever.

It is specifically agreed, however, that this General Release does not have any effect on any rights or claims I may have against the Company which arise after the date I execute this General Release or on any vested rights I may have under any of the Company's qualified or non-qualified benefit plans or arrangements as of or after my last day of employment with the Company, or on any of the Company's obligations under the Agreement or as otherwise required under the Consolidated Omnibus Budget and Reconciliation Act of 1985 (COBRA).

I have carefully reviewed and fully understand all the provisions of the Agreement and General Release, including the foregoing Notice. I have not relied on any representation or statement, oral or written, by the Company or any of its representatives, which is not set forth in those documents.

The Agreement and this General Release, including the foregoing Notice, set forth the entire agreement between me and the Company with respect to this subject. I understand that my receipt and retention of the separation benefits covered by the Agreement are contingent not only on my execution of this General Release, but also on my continued compliance with my other obligations under the Agreement. I acknowledge that the Company gave me twenty-one (21)

days to consider whether I wish to accept or reject the separation benefits I am eligible to receive under the Agreement in exchange for this
General Release. I also acknowledge that the Company advised me to seek independent legal advice as to these matters, if I chose to do so. I
hereby represent and state that I have taken such actions and obtained such information and independent legal or other advice, if any, that I
believed were necessary for me to fully understand the effects and consequences of the Agreement and General Release prior to signing those
documents.

Exhibit 10.3

CHANGE OF CONTROL AGREEMENT

THIS CHANGE OF CONTROL AGREEMENT (the	"Agreement") entered into between HELMERICH & PAYNE, INC., a Delaware
corporation ("Helmerich & Payne"), and	, an individual (the "Executive"), dated as of the day of
, 2002.	

The Board of Directors of Helmerich & Payne (the "Board") has determined that it is in the best interests of the Company and its shareholders to assure that the Company will have the continued dedication of the Executive, notwithstanding the possibility, threat, or occurrence of a "Change of Control" (as defined in Section 2 of this Agreement) of the Company. The Board believes it is important to diminish the inevitable distraction of the Executive by virtue of the personal uncertainties and risks created by a pending or threatened Change of Control, and to encourage the Executive's full attention and dedication to the affairs of the Company during the term of this Agreement and upon the occurrence of such event. The Board also believes the Company is best served by providing the Executive with compensation arrangements upon a Change of Control which provide the Executive with individual financial security and which are competitive with those of other corporations. In order to accomplish these objectives, the Board has caused Helmerich & Payne to enter into this Agreement. For the purposes of Section 2 of this Agreement, "Company" means Helmerich & Payne, and/or any successor to all or a portion of its business and/or assets which assumes and agrees to perform this Agreement.

NOW, THEREFORE, IT IS HEREBY AGREED AS FOLLOWS:

1. Certain Definitions.

- (a) The "Effective Date" shall be the first date during the "Change of Control Period" (as defined in Section 1(b) of this Agreement) on which a Change of Control (as defined below) occurs, and, except as provided in the following sentence, no amount shall be paid or benefits provided under this Agreement if the Executive's employment is terminated for any reason prior to a Change of Control. Anything in this Agreement to the contrary notwithstanding, if the Executive's employment with the Company is terminated prior to the date on which a Change of Control occurs, and it is reasonably demonstrated that such termination (i) was at the request of a third party who has taken steps reasonably calculated to effect a Change of Control or (ii) otherwise arose in connection with or anticipation of a Change of Control, then for all purposes of this Agreement the "Effective Date" shall mean the date immediately prior to the date of such termination.
- (b) The "Change of Control Period" is the period commencing on the date hereof and ending on the earlier to occur of (i) the third anniversary of such date or (ii) the first day of the month next following the Executive's attainment of age 65 ("Normal Retirement Date"); provided, however, that commencing on the date two (2) years after the date hereof, and on each annual anniversary of such date (such date and each annual anniversary thereof is hereinafter referred to as the "Renewal Date"), the Change of Control Period shall be automatically extended so as to terminate on the earlier of (i) two years from such Renewal Date or (ii) the first day of the month coinciding with or next following the Executive's Normal Retirement Date, unless, at least 60 days prior to the Renewal Date, the Company shall give

notice that the Change of Control Period shall not be so extended in which event this Agreement shall continue for the remainder of its then current term and terminate as provided herein.

- 2. Change of Control. For the purpose of this Agreement, a "Change of Control" shall mean:
- (a) The acquisition by any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) (a "Person") of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of 15% or more of either (i) the then outstanding shares of common stock of the Company (the "Outstanding Company Common Stock") or (ii) the combined voting power of the then outstanding voting securities of the Company entitled to vote generally in the election of directors (the "Outstanding Company Voting Securities"); provided, however, that the following acquisitions shall not constitute a Change of Control: (i) any acquisition directly from the Company, (ii) any acquisition by the Company, or (iii) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Company or any corporation controlled by the Company; or
- (b) Individuals who, as of the date hereof, constitute the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board; provided, however, that any individual becoming a director subsequent to the date hereof whose election, appointment or nomination for election by the Company's shareholders was approved by a vote of at least a majority of the directors then comprising the Incumbent Board shall be considered as though such individual were a member of the Incumbent Board, but excluding, for purposes of this definition, any such individual whose initial assumption of office occurs as a result of an actual or threatened election contest with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board; or
- (c) Approval by the shareholders of the Company of a reorganization, share exchange, merger or consolidation or acquisition of assets of another corporation (a "Business Combination"), in each case, unless, following such Business Combination, (i) all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Company Common Stock and Outstanding Company Voting Securities immediately prior to such Business Combination will beneficially own, directly or indirectly, more than 70% of, respectively, the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, as the case may be, of the corporation resulting from such Business Combination (including, without limitation, a corporation which as a result of such transaction will own the Company through one or more subsidiaries) in substantially the same proportions as their ownership immediately prior to such Business Combination of the Outstanding Company Common Stock and Outstanding Company Voting Securities, as the case may be, (ii) no Person (excluding any employee benefit plan (or related trust) of the Company or such corporation resulting from such Business Combination) will beneficially own, directly or indirectly, 15% or more of, respectively, the then outstanding shares of common stock of the corporation resulting from such Business Combination or the combined voting power of the then outstanding voting securities of such corporation, except to the extent that such ownership existed prior to the Business Combination, and (iii) at least a majority of the members of the board of directors of

the corporation resulting from such Business Combination were members of the Incumbent Board at the time of the execution of the initial agreement, or of the action of the Board, providing for such Business Combination or were elected, appointed or nominated by the Board; or

- (d) Approval by the shareholders of the Company of (i) a complete liquidation or dissolution of the Company or, (ii) the sale or other disposition of all or substantially all of the assets of the Company, other than to a corporation, with respect to which following such sale or other disposition (A) more than 70% of, respectively, the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding voting securities of such corporation entitled to vote generally in the election of directors is then beneficially owned, directly or indirectly, by all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Company Common Stock and Outstanding Company Voting Securities immediately prior to such sale or other disposition in substantially the same proportion as their ownership immediately prior to such sale or other disposition, of the Outstanding Company Common Stock and Outstanding Company Voting Securities, as the case may be; (B) less than 15% of, respectively, the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding voting securities of such corporation entitled to vote generally in the election of directors will be beneficially owned, directly or indirectly, by any Person (excluding any employee benefit plan (or related trust) of the Company or such corporation), except to the extent that such Person owned 15% or more of the Outstanding Company Common Stock or Outstanding Company Voting Securities prior to the sale or disposition; and (C) at least a majority of the members of the board of directors of such corporation were members of the Incumbent Board at the time of the execution of the initial agreement, or of the action of the Board, providing for such sale or other disposition of assets of the Company or were elected, appointed or nominated by the Board.
- 3. Agreement Not Employment Contract. This Agreement shall be considered solely as a "severance agreement" obligating the Company to pay to the Executive certain amounts of compensation in the event and only in the event of his termination of employment after the Effective Date for the reasons and at the time specified herein. Apart from the obligation of the Company to provide the amounts of additional compensation as provided in this Agreement, the Company shall at all times retain the right to terminate the employment of the Executive since the obligation of the Company to the Executive shall only be considered as an employment relationship which exists between the Company and the Executive which may be terminated at will by either party subject to the obligation of the Company to make payment and perform its obligations as provided in this Agreement.

4. Termination.

(a) Death or Disability. This Agreement shall terminate automatically upon the Executive's death. If the Company determines in good faith that the Disability of the Executive has occurred (pursuant to the definition of "Disability" set forth below), it may give to the Executive written notice of its intention to terminate the Executive's employment. In such event, the Executive's employment with the Company shall terminate effective on the 30th day after the date of such notice (the "Disability Effective Date"), provided that, within such time period, the Executive shall not have returned to full-time performance of the Executive's duties.

For purposes of this Agreement, "Disability" means disability (either physical or mental) which, at least twenty-six (26) weeks after its commencement, is determined by a physician selected by the Company or its insurers and acceptable to the Executive or the Executive's legal representative to be total and permanent (such agreement as to acceptability not to be withheld unreasonably).

- (b) Cause. The Company may terminate the Executive's employment for "Cause." For purposes of this Agreement, termination of the Executive's employment by the Company for Cause shall mean termination for one of the following reasons: (i) the conviction of the Executive of a felony by a federal or state court of competent jurisdiction; (ii) an act or acts of dishonesty taken by the Executive and intended to result in substantial personal enrichment of the Executive at the expense of the Company; or (iii) the Executive's "willful" failure to follow a direct, reasonable and lawful written order from his supervisor, within the reasonable scope of the Executive's duties, which failure is not cured within thirty (30) days. Further, for purposes of this Section (b):
- (1) No act or failure to act, on the Executive's part shall be deemed "willful" unless done, or omitted to be done, by the Executive not in good faith and without reasonable belief that the Executive's action or omission was in the best interest of the Company.
- (2) The Executive shall not be deemed to have been terminated for Cause unless and until there shall have been delivered to the Executive a copy of a resolution duly adopted by the affirmative vote of not less than three-fourths (3/4ths) of the entire membership of the Board at a meeting of the Board called and held for such purpose (after reasonable notice to the Executive and an opportunity for the Executive, together with the Executive's counsel, to be heard before the Board), finding that in the good faith opinion of the Board the Executive was guilty of conduct set forth in clauses (i), (ii) or (iii) above and specifying the particulars thereof in detail.
- (c) Substantial Downturn. The Company may terminate the Executive's employment after the occurrence of a "Substantial Downturn" in the oil and gas industry. For purposes of this Agreement, Substantial Downturn shall mean a severe downturn in the oil and gas industry which shall be measured by the following objective criteria:
- (i) The West Texas Intermediate Price for Crude Oil remains at or below \$10/barrel for sixty (60) consecutive business days, or
- (ii) The price for each MMBtu of natural gas as quoted for the Henry Hub listing in "Gas Daily" remains at or below \$1.25 for sixty (60) consecutive business days.
- (d) Good Reason. The Executive's employment may be terminated by the Executive for Good Reason. For purposes of this Agreement, "Good Reason" means:
- (i) The assignment to the Executive of any duties inconsistent in any respect with the Executive's position (including status, offices, titles and reporting requirements), authority, duties or responsibilities held immediately prior to the Effective Date, or any other action by the Company which results in a diminution in such position, compensation, benefits, authority, duties or responsibilities.

- (ii) Any reduction in the Executive's (x) annual base salary ("Base Salary") in effect immediately prior to the Effective Date or (y) Annual Bonus. For purposes of this Agreement, ("Annual Bonus") shall mean the amount equal to the average of the annual bonus paid to the Executive by the Company during the two fiscal years immediately preceding the year in which the Effective Date occurs.
- (iii) The Company's requiring the Executive to be based at any office or location that is more than 25 miles from the office or location at which the Executive is based on the Effective Date, except for periodic travel reasonably required in the performance of the Executive's responsibilities.
- (iv) Any reduction by more than 10% in the overall level of the Executive's benefits under the Company's group life insurance, medical, health, accident, disability, incentive, savings, and retirement plans including all tax qualified and nonqualified plans or programs.
- (v) Any purported termination by the Company of the Executive's employment otherwise than as expressly permitted by this Agreement.
- (vi) Any failure by the Company to comply with and satisfy Section 10(c) of this Agreement.

For purposes of this Section 4(d), any good faith determination of "Good Reason" made by the Executive shall be conclusive.

- (e) Notice of Termination. Any termination by the Company for Cause or after the occurrence of a Substantial Downturn, or by the Executive for Good Reason, shall be communicated by Notice of Termination to the other party hereto given in accordance with Section 13(b) of this Agreement. For purposes of this Agreement, a "Notice of Termination" means a written notice which (i) indicates the specific termination provisions in this Agreement relied upon, (ii) sets forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of the Executive's employment under the provision so indicated, and (iii) if the Date of Termination (as defined below) is other than the date of receipt of such notice, specifies the termination date (which date shall be not more than 15 days after the giving of such notice). The failure by the Executive to set forth in the Notice of Termination any fact or circumstance which contributes to a showing of Good Reason shall not waive any right of the Executive hereunder or preclude the Executive from asserting such fact or circumstance in enforcing his rights hereunder.
- (f) Date of Termination. "Date of Termination" means the date of receipt of the Notice of Termination by either the Company or the Executive as the case may be or any later date specified therein; provided, however, if the Executive's employment is terminated by reason of death or Disability, the Date of Termination shall be the date of death of the Executive or the effective date of Disability, as the case may be.

- 5. Obligations of the Company upon Termination Following Change of Control.
- (a) Good Reason; Termination Other Than for Cause, Disability, Substantial Downturn or Death. If, within 24 months after the Effective Date, the Company shall terminate the Executive's employment other than for Cause, Disability, the occurrence of a Substantial Downturn, or death, or if the Executive shall terminate his employment for Good Reason:
- (i) The Company shall pay to the Executive in a lump sum in cash within 30 days after the Date of Termination the aggregate of the following amounts and the Company shall provide the following benefits:
- A. To the extent not theretofore paid, the Executive's Base Salary through the Date of Termination.
- B. The product of (i) the Annual Bonus or if higher, the Annual Bonus paid to the Executive for the last full fiscal year (if any) prior to the Date of Termination (the "Recent Bonus") and (ii) a fraction, the numerator of which is the number of days in the current fiscal year through the Date of Termination and the denominator of which is 365.
- C. The product obtained by multiplying two (2) times the sum of (i) the Base Salary and (ii) Annual Bonus, or if higher, the Recent Bonus.
- D. In the case of compensation previously deferred by the Executive, all amounts previously deferred (together with any accrued earnings thereon) and not yet paid by the Company, and any accrued vacation pay not yet paid by the Company.
- (ii) For the 24-month period following the Termination Date, or such longer period as any plan, program, practice or policy may provide, the Company shall continue benefits to the Executive and/or the Executive's family at least equal to those which would have been provided to them in accordance with the plans, programs, practices and policies, including health insurance and life insurance, in accordance with the most favorable plans, practices, programs or policies of the Company during the 90-day period immediately preceding the Effective Date. In the event the Company is unable to provide such benefits due to restrictions imposed by the Code, ERISA, or such other applicable law, the Company will provide the Executive with an alternate equivalent benefit.
- (iii) The Company shall pay up to \$5,000 in outplacement counseling services utilized by the Executive.
- (iv) Any options to purchase stock or rights to receive restricted stock granted under a plan adopted by Helmerich & Payne held by the Executive shall be immediately and automatically vested, fully earned and exercisable upon the Date of Termination unless previously exercised or forfeited.
- (v) Any unvested benefits under the Company's supplemental savings and retirement plans (nonqualified) shall become 100% vested and nonforfeitable and all

such benefits under such supplemental plans shall be paid in a lump sum within thirty (30) days of the Executive's Date of Termination.

- 6. Non-Exclusivity of Rights. Nothing in this Agreement shall prevent or limit the Executive's continuing or future participation in any benefit, bonus, incentive or other plans, programs, policies or practices, provided by the Company and for which the Executive may qualify, nor shall anything herein limit or otherwise affect such rights as the Executive may have under any stock option or other agreements with the Company. Amounts which are vested benefits or which the Executive is otherwise entitled to receive under any plan, policy, practice or program of the Company at or subsequent to the Date of Termination shall be payable in accordance with such plan, policy, practice or program.
- 7. Full Settlement. The Company's obligation to make the payments provided for in this Agreement and otherwise to perform its obligations hereunder shall not be affected by any set-off, counterclaim, recoupment, defense or other claim, right or action which the Company may have against the Executive or others. In no event shall the Executive be obligated to seek other employment or take any other action by way of mitigation of the amounts payable to the Executive under any of the provisions of this Agreement. The Company agrees to pay, to the full extent permitted by law, up to \$50,000 in legal fees and all expenses which the Executive may reasonably incur as a result of any contest (regardless of the outcome thereof) by the Company or others of the validity or enforceability of, or liability under, any provision of this Agreement or any guarantee of performance thereof (including as a result of any contest by the Executive about the amount of any payment pursuant to Section 8 of this Agreement), plus in each case interest at the applicable Federal rate provided for in Section 7872(f)(2) of the Code.
- 8. Maximum Payments. It is the objective of this Agreement to maximize the Executive's Net After-Tax Benefit (as defined herein) if payments or benefits provided under this Agreement are subject to excise tax under Section 4999 of the Code. Therefore, in the event it is determined that any payment or benefit by the Company to or for the benefit of the Executive, whether paid or payable or distributed or distributable pursuant to the terms of this Agreement or otherwise, including, by example and not by way of limitation, acceleration by the Company or otherwise of the date of vesting or payment or rate of payment under any plan, program or arrangement of the Company, would be subject to the excise tax imposed by Section 4999 of the Code or any interest or penalties with respect to such excise tax, together with any such interest and penalties, are hereinafter collectively referred to as the "Excise Tax"), the Company shall first make a calculation under which such payments or benefits provided to the Executive under this Agreement are reduced to the extent necessary so that no portion thereof shall be subject to the excise tax imposed by Section 4999 of the Code (the "4999 Limit"). The Company shall then compare (x) the Executive's Net After-Tax Benefit assuming application of the 4999 Limit with (y) the Executive's Net After-Tax Benefit without the application of the 4999 Limit and the Executive shall be entitled to the greater of (x) or (y). "Net After-Tax Benefit" shall mean the sum of (i) all payments and benefits which the Executive receives or is then entitled to receive from the Company, less (ii) the amount of federal income taxes payable with respect to the payments and benefits described in (i) above calculated at the maximum marginal income tax rate for each year in which such payments and benefits shall be paid to the Executive (based upon the rate for such year as set forth in the Code at the time of the first payment of the foregoing), less (iii) the amount of excise taxes imposed with respect to the payments and benefits described in (i) above by Section 4999 of the Code. The determination of

whether a payment or benefit constitutes an excess parachute payment shall be made by tax counsel selected by the Company and reasonably acceptable to the Executive. The costs of obtaining this determination shall be borne by the Company.

- 9. Confidential Information. The Executive shall hold in a fiduciary capacity for the benefit of the Company all secret or confidential information, knowledge or data relating to the Company, and respective businesses, which shall have been obtained by the Executive during the Executive's employment by the Company and which shall not be or become public knowledge (other than by acts by the Executive or his representatives in violation of this Agreement). After termination of the Executive's employment with the Company, the Executive shall not, without the prior written consent of the Company, communicate or divulge any such information, knowledge or data to anyone other than the Company and those designated by it.
- 10. Successors.
- (a) This Agreement is personal to the Executive and without the prior written consent of the Company shall not be assignable by the Executive otherwise than by will or the laws of descent and distribution. This Agreement shall inure to the benefit of and be enforceable by the Executive's legal representatives.
- (b) This Agreement shall inure to the benefit of and be binding upon the Company and its successors and assigns.
- (c) The Company will require any successor (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business and/or assets of the Company to assume expressly and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform it if no such succession had taken place. As used in this Agreement, "Company" shall mean the Company as hereinbefore defined and any successor to all or a portion of its business and/or assets which assumes and agrees to perform this Agreement by operation of law, or otherwise.
- 11. Release. The Executive agrees, if his employment is terminated under circumstances entitling Executive to payments under Section 5(a)(i) of this Agreement that, in consideration for the payments described in Section 5(a)(i), he will execute a General Release in substantially the form of Exhibit A attached hereto, through which the Executive releases the Company from any and all claims as may relate to or arise out of his employment relationship (excluding claims the Executive may have under any "employee pension plan" as described in Section 3(3) of ERISA or any claims under this Agreement). The form of the release may be modified as needed to reflect changes in the applicable law or regulations that are needed to provide a legally enforceable and binding release to the Company at the time of execution.
- 12. Indemnification and Insurance. The Executive shall be indemnified and held harmless by the Company during the term of this Agreement and following any termination of this Agreement for any reason whatsoever in the same manner as would any other key management employee of the Company with respect to acts or omissions occurring prior to (a) the termination of this Agreement or (b) the termination of employment of the Executive. In addition, during the term of this Agreement and for a period of five years following the termination of this Agreement for any reason whatsoever, the Executive shall be covered by a Company held Directors and Officers liability insurance policy covering acts or omissions

occurring prior to (a) the termination of this Agreement or (b) the termination of employment of the Executive. Provided, in no event will the obligation of the Company to indemnify the Executive or provide Directors and Officers insurance to the Executive under this Section 12 be less than the obligation and insurance coverage which the Company provided to the Executive immediately prior to the occurrence of a Change of Control.

13. Miscellaneous.

- (a) This Agreement shall be governed by and construed in accordance with the laws of the State of Oklahoma, without reference to principles of conflict of laws. The captions of this Agreement are not part of the provisions hereof and shall have no force or effect. This Agreement may not be amended or modified otherwise than by a written agreement executed by the parties hereto or their respective successors and legal representatives.
- (b) All notices and other communications hereunder shall be in writing and shall be given by hand delivery to the other party or by registered or certified mail, return receipt requested, postage prepaid, addressed as follows:

If to the Executive:

At his last known address evidenced on the Company's payroll records

If to the Company: Helmerich & Payne, Inc. Utica at Twenty-First Tulsa, Oklahoma 74114 Attention: General Counsel

or to such other address as either party shall have furnished to the other in writing in accordance herewith. Notice and communications shall be effective when actually received by the addressee.

- (c) The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement.
- (d) The Company may withhold from any amounts payable under this Agreement such federal, state or local taxes as shall be required to be withheld pursuant to any applicable law or regulation.
- (e) The Executive's failure to insist upon strict compliance with any provision hereof shall not be deemed to be a waiver of such provision or any other provision thereof.
- (f) This Agreement contains the entire understanding of the Company and the Executive with respect to the subject matter hereof.
- (g) THE EXECUTIVE AND THE COMPANY ACKNOWLEDGE THAT THE EMPLOYMENT OF THE EXECUTIVE BY THE COMPANY IS "AT WILL," AND MAY BE TERMINATED BY EITHER THE EXECUTIVE OR THE COMPANY AT ANY TIME, SUBJECT TO THE COMPANY'S OBLIGATION TO PROVIDE ADDITIONAL COMPENSATION AS PROVIDED IN THIS AGREEMENT. UPON A TERMINATION OF THE

EXECUTIVE'S EMPLOYMENT PRIOR TO THE EFFECTIVE DATE, THERE SHALL BE NO FURTHER RIGHTS UNDER THIS AGREEMENT.

- 14. No Trust. No action under this Agreement by the Company or its Board of Directors shall be construed as creating a trust, escrow or other secured or segregated fund, in favor of the Executive or his beneficiary. The status of the Executive and his beneficiary with respect to any liabilities assumed by the Company hereunder shall be solely those of unsecured creditors of the Company. Any asset acquired or held by the Company in connection with liabilities assumed by it hereunder, shall not be deemed to be held under any trust, escrow or other secured or segregated fund for the benefit of the Executive or his beneficiary or to be security for the performance of the obligations of the Company, but shall be, and remain a general, unpledged, unrestricted asset of the Company at all times subject to the claims of general creditors of the Company.
- 15. No Assignability. Neither the Executive nor his beneficiary, nor any other person shall acquire any right to or interest in any payments payable under this Agreement, otherwise than by actual payment in accordance with the provisions of this Agreement, or have any power to transfer, assign, anticipate, pledge, mortgage or otherwise encumber, alienate or transfer any rights hereunder in advance of any of the payments to be made pursuant to this Agreement or any portion thereof which is expressly declared to be nonassignable and nontransferable. No right or benefit hereunder shall in any manner be liable for or subject to the debts, contracts, liabilities, or torts of the person entitled to such benefit.

IN WITNESS WHEREOF, the Executive has hereunto set his hand and, pursuant to the authorization from its Board of Directors, Helmerich & Payne has caused these presents to be executed in its name on its behalf, all as of the day and year first above written.

"EXECUTIVE"

HELMERICH & PAYNE, INC., a Delaware corporation

By "HELMERICH & PAYNE"

EXHIBIT A

NOTICE. VARIOUS LAWS, INCLUDING TITLE VII OF THE CIVIL RIGHTS ACT OF 1964, THE CIVIL RIGHTS ACT OF 1866, THE PREGNANCY DISCRIMINATION ACT OF 1978, THE EQUAL PAY ACT, THE CIVIL RIGHTS ACT OF 1991, THE AGE DISCRIMINATION IN EMPLOYMENT ACT, THE REHABILITATION ACT OF 1973, THE AMERICANS WITH DISABILITIES ACT, THE EMPLOYEE RETIREMENT INCOME SECURITY ACT AND THE VETERANS REEMPLOYMENT RIGHTS ACT (ALL AS AMENDED FROM TIME TO TIME), PROHIBIT EMPLOYMENT DISCRIMINATION BASED ON SEX, RACE, COLOR, NATIONAL ORIGIN, RELIGION, AGE, DISABILITY, ELIGIBILITY FOR COVERED EMPLOYEE BENEFITS AND VETERAN STATUS. YOU MAY ALSO HAVE RIGHTS UNDER LAWS SUCH AS THE OLDER WORKER BENEFIT PROTECTION ACT OF 1990, THE WORKER ADJUSTMENT AND RETRAINING ACT OF 1988, THE FAIR LABOR STANDARDS ACT, THE FAMILY AND MEDICAL LEAVE ACT, THE OCCUPATIONAL HEALTH AND SAFETY ACT AND OTHER FEDERAL, STATE AND/OR MUNICIPAL STATUTES, ORDERS OR REGULATIONS PERTAINING TO LABOR, EMPLOYMENT AND/OR EMPLOYEE BENEFITS. THESE LAWS ARE ENFORCED THROUGH THE UNITED STATES DEPARTMENT OF LABOR AND ITS AGENCIES, INCLUDING THE EQUAL EMPLOYMENT OPPORTUNITY COMMISSION (EEOC), AND VARIOUS STATE AND MUNICIPAL LABOR DEPARTMENTS, FAIR EMPLOYMENT BOARDS, HUMAN RIGHTS COMMISSIONS AND SIMILAR AGENCIES.

THIS GENERAL RELEASE IS BEING PROVIDED TO YOU IN CONNECTION WITH THE SPECIAL, INDIVIDUALIZED
SEVERANCE PACKAGE OUTLINED IN SECTION 5(A)(I) OF YOUR CHANGE OF CONTROL AGREEMENT DATED,
2002 (THE "AGREEMENT"). THE FEDERAL OLDER WORKER BENEFIT PROTECTION ACT REQUIRES THAT YOU HAVE AT
LEAST TWENTY-ONE (21) DAYS, IF YOU WANT IT, TO CONSIDER WHETHER YOU WISH TO SIGN A RELEASE SUCH AS THIS
ONE IN CONNECTION WITH A SPECIAL, INDIVIDUALIZED SEVERANCE PACKAGE. YOU HAVE UNTIL THE CLOSE OF
BUSINESS TWENTY-ONE (21) DAYS FROM THE DATE YOU RECEIVE THIS GENERAL RELEASE TO MAKE YOUR DECISION.
YOU MAY NOT SIGN THIS GENERAL RELEASE UNTIL, AT THE EARLIEST, YOUR OFFICIAL DATE OF SEPARATION FROM
EMPLOYMENT,

BEFORE EXECUTING THIS GENERAL RELEASE YOU SHOULD REVIEW THESE DOCUMENTS CAREFULLY AND CONSULT WITH YOUR ATTORNEY.

YOU MAY REVOKE THIS GENERAL RELEASE WITHIN SEVEN (7) DAYS AFTER YOU SIGN IT AND IT SHALL NOT BECOME EFFECTIVE OR ENFORCEABLE UNTIL THAT REVOCATION PERIOD HAS EXPIRED. IF YOU DO NOT ACCEPT THE SEVERANCE PACKAGE AND SIGN AND RETURN THIS GENERAL RELEASE, OR IF YOU EXERCISE YOUR RIGHT TO REVOKE THE GENERAL RELEASE AFTER SIGNING IT, YOU WILL NOT BE ELIGIBLE FOR THE SPECIAL, INDIVIDUALIZED SEVERANCE PACKAGE. ANY REVOCATION MUST BE IN WRITING AND MUST BE RECEIVED BY HELMERICH & PAYNE, INC., ATTENTION: DIRECTOR, HUMAN RESOURCES, HELMERICH & PAYNE, INC., UTICA AT TWENTY-FIRST, TULSA, OK 74114, WITHIN THE SEVEN-DAY PERIOD FOLLOWING YOUR EXECUTION OF THIS GENERAL RELEASE.

GENERAL RELEASE

In consideration of the special, individualized severance package offered to me by Helmerich & Payne, Inc. and the separation benefits I will receive as reflected in Section 5(a)(i) of my Change of Control Agreement dated ________, 2002 (the "Agreement"), I hereby release and discharge Helmerich & Payne, Inc. and its predecessors, successors, affiliates, parent, subsidiaries and partners and each of those entities' employees, officers, directors and agents (hereafter collectively referred to as the "Company") from all claims, liabilities, demands, and causes of action, known or unknown, fixed or contingent, which I may have or claim to have against the Company either as a result of my past employment with the Company and/or the severance of that relationship and/or otherwise, and hereby waive any and all rights I may have with respect to and promise not to file a lawsuit to assert any such claims.

This General Release includes, but is not limited to, claims arising under Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1866, the Pregnancy Discrimination Act of 1978, the Equal Pay Act, the Civil Rights Act of 1991, the Age Discrimination in Employment Act, the Rehabilitation Act of 1973, the Americans With Disabilities Act, the Employee Retirement Income Security Act or 1974 and the Veterans Reemployment Rights Act (all as amended from time to time). This General Release also includes, but is not limited to, any rights I may have under the Older Workers Benefit Protection Act of 1990, the Worker Adjustment and Retraining Act of 1988, the Fair Labor Standards Act, the Family and Medical Leave Act, the Occupational Health and Safety Act and any other federal, state and/or municipal statutes, orders or regulations pertaining to labor, employment and/or employee benefits. This General Release also applies to any claims or rights I may have growing out of any legal or equitable restrictions on the Company's rights not to continue an employment relationship with its employees, including any express or implied employment contracts, and to any claims I may have against the Company for fraudulent inducement or misrepresentation, defamation, wrongful termination or other retaliation claims in connection with workers' compensation or alleged "whistleblower" status or on any other basis whatsoever.

It is specifically agreed, however, that this General Release does not have any effect on any rights or claims I may have against the Company which arise after the date I execute this General Release or on any vested rights I may have under any of the Company's qualified benefit plans or arrangements as of or after my last day of employment with the Company or on any of the Company's obligations under the Agreement.

I have carefully reviewed and fully understand all the provisions of the Agreement and General Release, including the foregoing Notice. I have not relied on any representation or statement, oral or written, by the Company or any of its representatives, which is not set forth in those documents.

The Agreement and this General Release, including the foregoing Notice, set forth the entire agreement between me and the Company with respect to this subject. I understand that my receipt and retention of the separation benefits covered by the Agreement are contingent not only on my execution of this General Release, but also on my continued compliance with my other obligations under the Agreement. I acknowledge that the Company gave me twenty-one (21) days to consider whether I wish to accept or reject the separation benefits I am eligible to receive under the Agreement in exchange for this General Release. I also acknowledge that the

Company advised me to seek independent legal advice as to these matters, if I chose to do so. I hereby represent and state that I have taken such actions and obtained such information and independent legal or other advice, if any, that I believed were necessary for me to fully understand the effects and consequences of the Agreement and General Release prior to signing those documents.
Dated this day of,

Exhibit 10.4

SECOND AMENDMENT

This Second Amendment (this "Second Amendment") is entered into as of July 16, 2002 by and among Helmerich & Payne International Drilling Co., a Delaware corporation (the "Borrower"), Helmerich & Payne, Inc., a Delaware corporation (the "Parent"), and Bank One, Oklahoma, N.A., a national banking association (the "Bank").

WHEREAS, the Parent and the Bank entered into that certain Credit Agreement dated as of October 27, 1998, as amended pursuant to that certain First Amendment to Credit Agreement dated as of May 20, 1999 (as so amended, the "Existing Agreement"), pursuant to which the Bank made loans to the Parent on the terms and conditions set forth therein;

WHEREAS, simultaneously with the effectiveness of this Second Amendment in accordance with the provisions of Section 6 hereof, (i) the Parent shall be deemed to have paid the \$50,000,000 principal amount of the Advances outstanding under the Existing Agreement, (ii) the Borrower shall be deemed to have borrowed \$50,000,000 from the Bank, and (iii) the Parent has assigned to the Borrower, and the Borrower has assumed, the indebtedness, obligations and liabilities (including, without limitation, all interest accrued but unpaid on the \$50,000,000 of Advances outstanding under the Existing Agreement) of the Parent under the Existing Agreement and the other Loan Documents (as defined in the Existing Agreement) pursuant to an Assignment and Assumption Agreement dated as of the date hereof among the Parent, the Borrower and the Bank;

WHEREAS, the Bank agreed to such assignment provided that (i) the Parent guaranties all of the indebtedness, obligations and liabilities of the Borrower under the Existing Agreement, as amended by this Second Amendment, and the other Loan Documents as provided for in Section 11 of the Existing Agreement as amended by this Second Amendment, and (ii) Helmerich & Payne Rasco, Inc., Helmerich & Payne (Columbia) Drilling Co. and Helmerich & Payne del Ecuador, Inc. guaranty all of the indebtedness, obligations and liabilities of the Borrower under the Existing Agreement, as amended by this Second Amendment, and the other Loan Documents by entering into a Guaranty Agreement in substantially the form attached hereto as Exhibit "C";

WHEREAS, the Parent and the Borrower have requested that the Bank consent to the Parent's planned spin-off of its Oil and Gas Division notwithstanding the provisions of the Existing Agreement that prohibit such spin-off; and

WHEREAS, as a result, the parties hereto desire to amend the Existing Agreement in certain respects as more fully described herein;

NOW, THEREFORE, in consideration of the undertakings set forth herein and other good and valuable consideration, the receipt of which is hereby acknowledged, the Borrower and the Bank hereby agree as follows:

SECTION 1. DEFINED TERMS. Capitalized terms used and not otherwise defined in this Second Amendment shall have the meanings attributed to them in Section 1 of the Existing Agreement as amended by this Second Amendment.

SECTION 2. AMENDMENT OF EXISTING AGREEMENT. Upon the satisfaction of the conditions precedent set forth in Section 6 of this Second Amendment, the Existing Agreement shall be amended as follows:

(i) The first paragraph of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new first paragraph:

"THIS CREDIT AGREEMENT dated as of October 27, 1998, as amended pursuant to a First Amendment to Credit Agreement dated as of May 20, 1999, as assigned to and assumed by Helmerich & Payne International Drilling Co. pursuant to an Assignment and Assumption Agreement dated as of July 16, 2002, and as amended pursuant to a Second Amendment dated as of July 16, 2002, is among HELMERICH & PAYNE INTERNATIONAL DRILLING CO., a Delaware corporation, HELMERICH & PAYNE, INC., a Delaware corporation, and Bank One, Oklahoma, N.A., a national banking association."

- (ii) Subsection 1.2 of the Existing Agreement is hereby amended by adding thereto in the appropriate alphabetical order the following new definitions:
- "Administrative Agent. `Administrative Agent' shall mean Bank of Oklahoma, National Association, in its capacity as Administrative Agent for the Lenders under the Bank of Oklahoma Agreement."
- "Agreement. `Agreement' shall mean this credit agreement among the Parent, the Borrower and the Bank, as amended pursuant to a First Amendment to Credit Agreement dated as of May 20, 1999, entered into by the Parent and the Bank, as assigned to and assumed by the Borrower pursuant to an Assignment and Assumption Agreement dated as of July 16, 2002, among the Parent, the Borrower and the Bank, as amended pursuant to a Second Amendment dated as of July 16, 2002 among the Borrower, the Parent and the Bank, and as it may be further amended, supplemented or otherwise modified and in effect from time to time."

"Alternate Base Rate. `Alternate Base Rate' shall mean, for any day, a rate of interest per annum equal to the higher of (i) the Prime Rate for such day, and (ii) the sum of the Federal Funds Effective Rate for such day plus 1/2% per annum."

Alternate Base Rate Advance. `Alternate Base Rate Advance' shall mean an Advance which, except as otherwise provided in Subsection 2.11 hereof, bears interest at the Alternate Base Rate."

"Applicable Percentage. `Applicable Percentage' shall mean (i) for any day from and including the date hereof through and including August 29, 2002, 0.50% per annum (50 bps), and (ii) for any day thereafter, 0.70% per annum (70 bps)."

"Bank. 'Bank' shall mean Bank One, Oklahoma, N.A., a national banking association, and its successors and assigns."

"Bank of Oklahoma Agreement. `Bank of Oklahoma Agreement' shall mean that certain Credit Agreement dated July 16, 2002 by and among the Borrower, the Parent, the Lenders from time to time party thereto, and Bank of Oklahoma, National Association, as Administrative Agent for such Lenders in the form of the copy thereof delivered to the Bank by the Parent, whether or not a Credit Agreement identical to or based on such copy is ever executed by the Borrower or the Parent or ever becomes effective in accordance with its terms, and without regard to any amendment, supplement or other modification to such Draft or any amendment, supplement or other modification to any Credit Agreement entered into by the Borrower or the Parent."

"Borrower, 'Borrower' shall mean Helmerich & Payne International Drilling Co., a Delaware corporation, and its successors and assigns."

"Borrowing Date. 'Borrowing Date' shall mean a date on which an Advance is made hereunder."

"Conversion/Continuation Notice. 'Conversion/Continuation Notice' shall have the meaning given such term in Subsection 2.9 hereof."

"Eurodollar Advance. `Eurodollar Advance' shall mean an Advance which, except as otherwise provided in Subsection 2.11 hereof, bears interest at the applicable Eurodollar Rate."

"Eurodollar Base Rate. `Eurodollar Base Rate' shall mean, with respect to a Eurodollar Advance for the relevant Interest Period, the applicable British Bankers' Association LIBOR rate for deposits in U.S. dollars as reported by any generally recognized financial information service as of 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, and having a maturity equal to such Interest Period, provided that, if no such British Bankers' Association LIBOR rate is available to the Bank, the applicable Eurodollar Base Rate for the relevant Interest Period shall instead be the rate determined by the Bank to be the rate at which the Bank or one of its Affiliate banks offers to place deposits in U.S. dollars with first-class banks in the London interbank market at approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, in the approximate amount of the Bank's relevant Eurodollar Loan and having a maturity equal to such Interest Period."

"Eurodollar Rate. `Eurodollar Rate' shall mean, with respect to a Eurodollar Advance for the relevant Interest Period, the sum of (i) the quotient of (a) the Eurodollar Base Rate applicable to such Interest Period, divided by (b) one minus the Reserve Requirement (expressed as a decimal) applicable to such Interest Period, plus (ii) the Applicable Percentage."

"Excluded Taxes. `Excluded Taxes' shall mean, in the case of the Bank or any applicable Lending Installation, taxes imposed on its overall net income, and franchise taxes imposed on it, by (i) the jurisdiction under the laws of which the Bank is incorporated or organized or (ii) the jurisdiction in which the Bank's principal executive office or the Bank's applicable Lending Installation is located."

"Federal Funds Effective Rate. `Federal Funds Effective Rate' shall mean, for any day, an interest rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published for such day (or, if such day is not a Business Day, for the immediately preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations at approximately 10:00 a.m. (Chicago time) on such day on such transactions received by the Bank from three Federal funds brokers of recognized standing selected by the Bank in its sole discretion."

"Lending Installation. `Lending Installation' shall mean the main office of the Bank in Chicago, Illinois or such other office, branch, subsidiary or affiliate of the Bank as may be selected by the Bank pursuant to Subsection 2.16 hereof."

"Other Taxes. 'Other Taxes' shall have the meaning given such term in Subsection 2.22(ii) hereof."

"Parent. 'Parent' shall mean Helmerich & Payne, Inc., a Delaware corporation, and its successors and assigns."

"Regulation D. `Regulation D' shall mean Regulation D of the Board of Governors of the Federal Reserve System as from time to time in effect and any successor thereto or other regulation or official interpretation of said Board of Governors relating to reserve requirements applicable to member banks of the Federal Reserve System."

"Reserve Requirement. `Reserve Requirement' shall mean, with respect to an Interest Period, the maximum aggregate reserve requirement (including all basic, supplemental, marginal and other reserves) which is imposed under Regulation D on Eurocurrency liabilities."

"Second Amendment. 'Second Amendment' shall mean that certain Second Amendment dated as of July 16, 2002 among the Parent, the Borrower and the Bank pursuant to which this Agreement was amended as provided therein."

"Second Amendment Effective Date. 'Second Amendment Effective Date' shall have the meaning given such term in Section 6 of the Second Amendment."

"Taxes. `Taxes' shall mean any and all present or future taxes, duties, levies, imposts, deductions, charges or withholdings, and any and all liabilities with respect to the foregoing, but excluding Excluded Taxes and Other Taxes."

"Type. `Type' shall mean, with respect to any Advance, its nature as a Alternate Base Rate Advance or a Eurodollar Advance."

- (iii) The definition of "Advance" set forth in Section 1.2 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new definition of "Advance":
- "Advance. 'Advance' shall mean a borrowing hereunder, (i) made by the Bank on the same Borrowing Date, or (ii) converted or continued by the Bank on the same date of conversion or continuation, consisting, in either case, of the same Type and, in the case of Eurodollar Loans, for the same Interest Period."
- (iv) The definitions of "Borrower Entity" and "Business Day" set forth in Section 1.2 of the Existing Agreement are hereby amended by deleting then in their entirety and substituting in lieu thereof the following new definitions of "Borrower Entity" and "Business Day":

"Borrower Entity. 'Borrower Entity' shall mean all of the Parent, the Borrower, the Guarantors and each Subsidiary of the Parent, the Borrower and the Guarantors, or any of them, as indicated by the context in which such term is used.

"Business Day. 'Business Day' shall mean (i) with respect to any borrowing, payment or rate selection of Eurodollar Advances, a day (other than a Saturday or Sunday) on which banks generally are open in Chicago, Illinois, New York, New York and Tulsa, Oklahoma for the conduct of substantially all of their commercial lending activities, interbank wire transfers can be made on the Fedwire system and dealings in United States dollars are carried on in the London interbank market and (ii) for all other purposes, a day (other than a Saturday or Sunday) on which banks generally are open in Chicago, Illinois and Tulsa, Oklahoma for the conduct of substantially all of their commercial lending activities and interbank wire transfers can be made on the Fedwire system."

(v) The definition of "Disbursement Request" set forth in Section 1.2 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new definition of "Disbursement Request":

"Disbursement Request. 'Disbursement Request' shall have the meaning given such term in Subsection 2.8 hereof."

(vi) The definitions of "Guaranty" and "Guarantors" set forth in Section 1.2 of the Existing Agreement are hereby amended by deleting them in their entirety and substituting in lieu thereof the following new definitions of "Guaranty" and "Guarantors":

"Guaranty' shall mean (i) until the Second Amendment Effective Date, that certain Guaranty Agreement dated as of October 27, 1998, in substantially the form of Exhibit "C" to the Credit Agreement, executed by the Guarantors named therein in favor of the Bank, pursuant to Section 3 hereof, as amended, supplemented or otherwise modified from time to time, and (ii) on and after the Second Amendment Effective Date, that certain Guaranty Agreement dated as of July 16, 2002, in substantially the form of Exhibit "C" to the Second Amendment, executed by each Guarantor (other than the Parent) in favor of the Bank, pursuant to Section 3 hereof, as amended, supplemented or otherwise modified from time to time.

"Guarantors.' Guarantors' shall mean collectively, the Parent, Helmerich & Payne Rasco, Inc., an Oklahoma corporation, Helmerich & Payne (Columbia) Drilling Co., an Oklahoma corporation, Helmerich & Payne del Ecuador, Inc., an Oklahoma corporation, and each Person that becomes a Material Subsidiary after the Second Amendment Effective Date."

(vii) The definition of "Interest Period" set forth in Section 1.2 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new definition of "Interest Period":

"Interest Period. 'Interest Period' shall mean, with respect to a Eurodollar Advance, a period of one month commencing on a Business Day selected by the Borrower pursuant to this Agreement. Such Interest Period shall end on the day which corresponds numerically to such date one month thereafter, provided, however, that if there is no such numerically corresponding day in such next month, such Interest Period shall end on the last Business Day of such next month. If an Interest Period would otherwise end on a day which is not a Business Day, such Interest Period shall end on the next succeeding Business Day, provided, however, that if said next succeeding Business Day falls in a new calendar month, such Interest Period shall end on the immediately preceding Business Day."

(viii) The definition of "Material Adverse Effect" set forth in Section 1.2 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new definition of "Material Adverse Effect":

"Material Adverse Effect.' Material Adverse Effect' shall mean a material adverse effect on or material impairment of (i) the validity or enforceability of any Loan Document or the rights, benefits or remedies of the Bank under any Loan Document, (ii) the condition (financial or otherwise), operations, business, assets, liabilities or prospects of the Consolidated Group taken as a whole, or (iii) the ability of the Borrower and the Guarantors to perform or fulfill their obligations under the Loan Documents."

(ix) The definition of "Prime Rate" set forth in Section 1.2 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new definition of "Prime Rate":

"Prime Rate. 'Prime Rate' shall mean a rate per annum equal to the prime rate of interest announced from time to time by the Bank or its parent (which is not necessarily the lowest rate charged to any customer), changing when and as said prime rate changes."

(x) The definition of "Revolving Note" set forth in Section 1.2 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new definition of "Revolving Note":

"Revolving Note.' Revolving Note' shall mean a promissory note in the original principal amount of Fifty Million and No/100 Dollars (\$50,000,000.00) executed by the Borrower in order to evidence all Advances made under the Revolving Credit Facility pursuant to Section 2 hereof, substantially in the form of Exhibit "A" attached hereto, as the same by be amended, supplemented or otherwise modified, renewed or extended from time to time."

- (xi) Section 1 of the Existing Agreement is hereby amended by adding thereto after Subsection 1.5 of the Existing Agreement the following new Subsection 1.6:
- "1.6 Bank of Oklahoma Agreement Definitions. Terms used herein that are defined in Article I of the Bank of Oklahoma Agreement and that are not otherwise defined in this Agreement shall have the respective meanings set forth in Article I of the Bank of Oklahoma Agreement. For purposes hereof, all of the provisions of Article I of the Bank of Oklahoma Agreement, together with related definitions and ancillary provisions, are hereby incorporated herein by reference, mutatis mutandis, and shall be deemed to continue in effect for the Bank's benefit as in effect on the Second Amendment Effective Date, whether or not the Bank of Oklahoma Agreement remains in effect or is amended, waived or otherwise modified by the parties thereto."

- (xii) Section 2 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new Section 2:
- "2. LENDING AGREEMENT. Subject to the terms and conditions of this Agreement and the Loan Documents, and in reliance upon the representations and warranties contained herein and therein:
- "2.1 Revolving Credit Facility. The Bank agrees to establish a revolving credit facility, to be designated as the "Revolving Credit Facility", in an aggregate principal amount equal to the Revolving Commitment. From and including the date of this Agreement and prior to the Maturity Date, the Bank agrees, on the terms and conditions set forth in this Agreement, to make Advances to the Borrower from time to time in amounts not to exceed in the aggregate at any one time outstanding the amount of the Revolving Commitment. Subject to the terms of this Agreement, the Borrower may borrow, repay and reborrow at any time prior to the Maturity Date. The Revolving Commitment shall expire on the Maturity Date.
- "2.2 Required Payments; Termination. Any outstanding Advances, together with all unpaid interest accrued thereon, and all other unpaid amounts due under this Agreement and the other Loan Documents shall be paid in full by the Borrower on the Maturity Date.
- "2.3 Types of Advances. The Advances may be Alternate Base Rate Advances or Eurodollar Advances, or a combination thereof, selected by the Borrower in accordance with Subsections 2.8 and 2.9 hereof.
- "2.4 Fees. At or prior to the Closing, the Borrower paid a nonrefundable commitment fee in the amount of \$25,000. The Borrower also agrees to pay to the Bank a non-use fee of 0.125% per annum on the daily unused portion of the Revolving Commitment from the date hereof to and including the Maturity Date, payable quarterly in arrears on the last day of each March, June, September and December, beginning on December 31, 1998, and on the Maturity Date.
- "2.5 Reductions in Revolving Commitment. The Borrower may permanently reduce the Revolving Commitment in whole, or in part in integral multiples of \$1,000,000, upon at least three Business Days' prior written notice to the Bank, which notice shall specify the amount of any such reduction, provided, however, that the amount of the Revolving Commitment may not be reduced below the aggregate principal amount of the outstanding Advances. All accrued and unpaid non-use fees shall be payable on the effective date of any termination of the Revolving Commitment.
- "2.6 Minimum Amount of Each Advance. Each Eurodollar Advance shall be in the minimum amount of \$1,000,000 (and in multiples of \$1,000,000 if

in excess thereof), and each Alternate Base Rate Advance shall be in the minimum amount of \$100,000 (and in multiples of \$100,000 if in excess thereof), provided, however, that any Alternate Base Rate Advance may be in the amount of the unused Revolving Commitment.

- "2.7 Optional Principal Payments. The Borrower may from time to time pay, without penalty or premium, all outstanding Alternate Base Rate Advances, or, in a minimum aggregate amount of \$100,000 or any integral multiple of \$100,000 in excess thereof, any portion of the outstanding Alternate Base Rate Advances upon one Business Days' prior notice to the Bank. The Borrower may from time to time pay, subject to the payment of any funding indemnification amounts required by Subsection 2.21 hereof but without penalty or premium, all outstanding Eurodollar Advances, or, in a minimum aggregate amount of \$1,000,000 or any integral multiple of \$1,000,000 in excess thereof, any portion of the outstanding Eurodollar Advances upon three Business Days' prior notice to the Bank.
- "2.8 Method of Selecting Types for New Advances. The Borrower shall select the Type of Advance from time to time. The Borrower shall give the Bank irrevocable notice (a "Disbursement Request") not later than 11:00 a.m. (Chicago time) on the Borrowing Date of each Alternate Base Rate Advance and at least three Business Days before the Borrowing Date for each Eurodollar Advance, specifying:
- (i) the Borrowing Date, which shall be a Business Day, of such Advance,
- (ii) the aggregate amount of such Advance, and
- (iii) the Type of Advance selected.

Not later than noon (Chicago time) on each Borrowing Date, the Bank shall make available its Advance or Advances in funds immediately available in Chicago to the Borrower by credit to the Borrower's primary operating account maintained with the Bank or any of its Affiliates or by wire transfer, as instructed by the Borrower in a written notice delivered to the Bank with the applicable Disbursement Request.

"2.9 Conversion and Continuation of Outstanding Advances. Alternate Base Rate Advances shall continue as Alternate Base Rate Advances unless and until such Alternate Base Rate Advances are converted into Eurodollar Advances pursuant to this Subsection 2.9 or are repaid in accordance with Subsection 2.7 hereof. Each Eurodollar Advance shall continue as a Eurodollar Advance until the end of the then applicable Interest Period therefor, at which time such Eurodollar Advance shall be automatically converted into a new Eurodollar Advance with a one month Interest Period unless (x) such Eurodollar Advance is

or was repaid in accordance with Subsection 2.7 hereof, or (y) the Borrower shall have given the Bank a Conversion/Continuation Notice (as defined below) requesting that, at the end of such Interest Period, such Eurodollar Advance be converted to an Alternate Base Rate Advance. Subject to the terms of Subsection 2.6 hereof, the Borrower may elect from time to time to convert all or any part of an Alternate Base Rate Advance into a Eurodollar Advance. The Borrower shall give the Bank irrevocable notice (a "Conversion/Continuation Notice") of each conversion of an Alternate Base Rate Advance into a Eurodollar Advance not later than 11:00 a.m. (Chicago time) at least three Business Days prior to the date of the requested conversion or continuation, specifying:

- (i) the requested date, which shall be a Business Day, of such conversion or continuation,
- (ii) the aggregate amount and Type of the Advance which is to be converted or continued, and
- (iii) the amount of such Advance which is to be converted into or continued as a Eurodollar Advance.
- "2.10 Changes in Interest Rate, etc. Each Alternate Base Rate Advance shall bear interest on the outstanding principal amount thereof, for each day from and including the date such Advance is made or is converted from a Eurodollar Advance into an Alternate Base Rate Advance pursuant to Subsection 2.9 hereof, to but excluding the date it is paid or is converted into a Eurodollar Advance pursuant to Subsection 2.9 hereof, at a rate per annum equal to the Alternate Base Rate for such day. Changes in the rate of interest on that portion of any Advance maintained as an Alternate Base Rate Advance will take effect simultaneously with each change in the Alternate Base Rate. Each Eurodollar Advance shall bear interest on the outstanding principal amount thereof from and including the first day of the Interest Period applicable thereto to (but not including) the last day of such Interest Period at the interest rate determined by the Bank as applicable to such Eurodollar Advance in accordance with the terms hereof. No Interest Period may end after the Maturity Date.
- "2.11 Rates Applicable After Default. Notwithstanding anything to the contrary contained in Subsection 2.8, 2.9 or 2.10 hereof, during the continuance of an Event of Default the Bank may, at its option, by notice to the Borrower (which notice may be revoked at the option of the Bank), declare that no Advance may be made as, converted into or continued as a Eurodollar Advance. During the continuance of an Event of Default the Bank may, at its option, by notice to the Borrower (which notice may be revoked at the option of the Bank), declare that (i) each Eurodollar Advance shall bear interest for the remainder of the applicable Interest Period at the rate otherwise applicable to such Interest Period plus 2% per annum and (ii) each Alternate Base Rate Advance shall bear interest at a rate per annum equal to the Alternate Base Rate in effect from time to time

plus 2% per annum, provided that, during the continuance of an Event of Default under Subsection 8.9 hereof, the interest rates set forth in clauses (i) and (ii) above shall be applicable to all Advances without any election or action on the part of the Bank.

- "2.12. Method of Payment. All payments due to the Bank under this Agreement shall be made, without setoff, deduction, or counterclaim, in immediately available funds to the Bank at the Bank's address specified pursuant to Subsection 10.3 hereof, or at any other Lending Installation of the Bank specified in writing by the Bank to the Borrower, by noon (local time) on the date when due. The Bank is hereby authorized (with notice to the Borrower) to charge any ordinary deposit account of the Borrower maintained with the Bank for each payment of principal, interest and fees as it becomes due hereunder.
- "2.13 Revolving Note. The Advances from time to time outstanding under the Revolving Credit Facility shall be evidenced by the Revolving Note, which shall be made, executed and delivered by the Borrower at the Closing. Notwithstanding the principal amount stated on the face of the Revolving Note, the actual principal amount due from the Borrower on account of the Revolving Note shall be the sum of all Advances made by the Bank pursuant to the Revolving Credit Facility, less all principal payments actually received by the Bank in collected funds. All Advances and payments under the Revolving Credit Facility shall be recorded by the Bank in its books and records, and the unpaid principal balance so recorded shall be presumptive evidence of the principal amount owing under the Revolving Credit Facility.
- "2.14 Telephonic Notices. The Borrower hereby authorizes the Bank to extend, convert or continue Advances, effect selections of Types of Advances and to transfer funds based on telephonic notices made by any person or persons the Bank in good faith believes to be acting on behalf of the Borrower, it being understood that the foregoing authorization is specifically intended to allow Disbursement Requests and Conversion/Continuation Notices to be given telephonically. The Borrower agrees to deliver promptly to the Bank a written confirmation, if such confirmation is requested by the Bank, of each telephonic notice signed by an Executive Officer. If the written confirmation differs in any material respect from the action taken by the Bank, the records of the Bank shall govern absent manifest error.
- "2.15 Interest Payment Dates; Interest and Fee Basis. Interest accrued on each Alternate Base Rate Advance shall be payable on each March, June, September and December, commencing with the first such date to occur after the date hereof and at maturity. Interest accrued on each Eurodollar Advance shall be payable on the last day of its applicable Interest Period, on any date on which the Eurodollar Advance is prepaid, whether by acceleration or otherwise, and at maturity. Interest and non-use fees shall be calculated for actual days elapsed on the basis of a 360-day year. Interest shall be payable for the day an Advance is

made but not for the day of any payment on the amount paid if payment is received prior to noon (local time) at the place of payment. If any payment of principal of or interest on an Advance shall become due on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day and, in the case of a principal payment, such extension of time shall be included in computing interest in connection with such payment.

- "2.16. Lending Installations. The Bank may book its Advances at any Lending Installation selected by the Bank and may change its Lending Installation from time to time. All terms of this Agreement shall apply to any such Lending Installation and the Advances and the Revolving Note shall be deemed held by the Bank for the benefit of any such Lending Installation. The Bank may, by written notice to the Borrower in accordance with Subsection 10.3 hereof, designate replacement or additional Lending Installations through which Advances will be made by it and for whose account payments due hereunder are to be made.
- "2.17 Maximum Lawful Interest Rate. It is not the intention of the Bank or any of the Borrower Entities to violate the laws of any applicable jurisdiction relating to usury or other restrictions on the maximum lawful interest rate. The Loan Documents and all other agreements between the Borrower Entities and the Bank, whether now existing or hereafter arising and whether written or oral, are hereby limited so that in no event shall the interest paid or agreed to be paid to the Bank for the use, forbearance or detention of money loaned, or for the payment or performance of any covenant or obligation contained herein or in any other Loan Document, exceed the maximum amount permissible under applicable law. If from any circumstances whatsoever fulfillment of any provision hereof or of any other Loan Document, at the time the performance of such provision shall be due, shall involve transcending the limit of validity prescribed by law, then, ipso facto, the obligation to be fulfilled shall be reduced to the limit of such validity. If from any such circumstances the Bank shall ever receive anything of value deemed interest under applicable law which would exceed interest at the highest lawful rate, such excessive interest shall beg applied to the reduction of the principal amount owning hereunder, and not to the payment of interest, or if such excessive interest exceeds any unpaid balance of principal, such excess shall be refunded to the appropriate Borrower Entity. All sums paid or agreed to be paid to the Bank for the use, forbearance or detention of monies shall, to the extent permitted by applicable law, be amortized, prorated, allocated and spread throughout the full term of such indebtedness until payment in full so that the rate of interest on account of such indebtedness is uniform throughout the term thereof. This Subsection 2.17 shall control every other provision of the Loan Documents and all other agreements between the Bank and the Borrower Entities contemplated thereby.
- "2.18 Yield Protection. If, on or after the date of this Agreement, the adoption of any law or any governmental or quasi-governmental rule, regulation,

policy, guideline or directive (whether or not having the force of law), or any change in the interpretation or administration thereof by any governmental or quasi-governmental authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by the Bank or applicable Lending Installation with any request or directive (whether or not having the force of law) of any such authority, central bank or comparable agency:

- (i) subjects the Bank or any applicable Lending Installation to any Taxes, or changes the basis of taxation of payments (other than with respect to Excluded Taxes) to the Bank in respect of its Eurodollar Advances, or
- (ii) imposes or increases or deems applicable any reserve, assessment, insurance charge, special deposit or similar requirement against assets of, deposits with or for the account of, or credit extended by, the Bank or any applicable Lending Installation (other than reserves and assessments taken into account in determining the interest rate applicable to Eurodollar Advances), or
- (iii) imposes any other condition the result of which is to increase the cost to the Bank or any applicable Lending Installation of making, funding or maintaining its Eurodollar Advances or reduces any amount receivable by the Bank or any applicable Lending Installation in connection with its Eurodollar Advances, or requires the Bank or any applicable Lending Installation to make any payment calculated by reference to the amount of Eurodollar Advances held or interest received by it, by an amount deemed material by the Bank,

and the result of any of the foregoing is to increase the cost to the Bank or applicable Lending Installation of making or maintaining its Eurodollar Advances or the Revolving Commitment or to reduce the return received by the Bank or applicable Lending Installation in connection with such Eurodollar Advances or the Revolving Commitment, then, within 15 days of demand by the Bank, the Borrower shall pay the Bank such additional amount or amounts as will compensate the Bank for such increased cost or reduction in amount received.

"2.19 Changes in Capital Adequacy Regulations. If the Bank determines the amount of capital required or expected to be maintained by the Bank, any Lending Installation or any corporation controlling the Bank is increased as a result of a Change, then, within 15 days of demand by the Bank, the Borrower shall pay the Bank the amount necessary to compensate for any shortfall in the rate of return on the portion of such increased capital which the Bank determines is attributable to this Agreement, its Advances or the Revolving Commitment (after taking into account the Bank's policies as to capital adequacy). "Change"

means (i) any change after the date of this Agreement in the Risk-Based Capital Guidelines, or (ii) any adoption of or change in any other law, governmental or quasi-governmental rule, regulation, policy, guideline, interpretation, or directive (whether or not having the force of law) after the date of this Agreement which affects the amount of capital required or expected to be maintained by the Bank or any Lending Installation or any corporation controlling the Bank. "Risk-Based Capital Guidelines" means (i) the risk-based capital guidelines in effect in the United States on the date of this Agreement, including transition rules, and (ii) the corresponding capital regulations promulgated by regulatory authorities outside the United States implementing the July 1988 report of the Basle Committee on Banking Regulation and Supervisory Practices Entitled "International Convergence of Capital Measurements and Capital Standards," including transition rules, and any amendments to such regulations adopted prior to the date of this Agreement.

- "2.20 Availability of Types of Advances. If the Bank determines that maintenance of its Eurodollar Advances at a suitable Lending Installation would violate any applicable law, rule, regulation, or directive, whether or not having the force of law, or if the Bank determines that (i) deposits of a type and maturity appropriate to match fund Eurodollar Advances are not available or (ii) the interest rate applicable to Eurodollar Advances does not accurately reflect the cost of making or maintaining Eurodollar Advances, then the Bank shall suspend the availability of Eurodollar Advances and require any affected Eurodollar Advances to be repaid or converted to Alternate Base Rate Advances, subject to the payment of any funding indemnification amounts required by Subsection 2.21 hereof.
- "2.21 Funding Indemnification. If any payment of a Eurodollar Advance occurs on a date which is not the last day of the applicable Interest Period, whether because of acceleration, prepayment or otherwise, or a Eurodollar Advance is not made on the date specified by the Borrower for any reason other than default by the Bank, the Borrower will indemnify the Bank for any loss or cost incurred by it resulting therefrom, including, without limitation, any loss or cost in liquidating or employing deposits acquired to fund or maintain such Eurodollar Advance.
- "2.22 Taxes. (i) All payments by the Borrower to or for the account of the Bank hereunder or under the Revolving Note shall be made free and clear of and without deduction for any and all Taxes. If the Borrower shall be required by law to deduct any Taxes from or in respect of any sum payable hereunder to the Bank, (a) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Subsection 2.22) the Bank receives an amount equal to the sum it would have received had no such deductions been made, (b) the Borrower shall make such deductions, (c) the Borrower shall pay the full amount deducted to the relevant authority in accordance with applicable law and (d) the Borrower shall

furnish to the Bank the original copy of a receipt evidencing payment thereof within 30 days after such payment is made.

- "(ii) In addition, the Borrower hereby agrees to pay any present or future stamp or documentary taxes and any other excise or property taxes, charges or similar levies which arise from any payment made hereunder or under the Revolving Note or from the execution or delivery of, or otherwise with respect to, this Agreement or the Revolving Note ("Other Taxes").
- (iii) The Borrower hereby agrees to indemnify the Bank for the full amount of Taxes or Other Taxes (including, without limitation, any Taxes or Other Taxes imposed on amounts payable under this Subsection 2.22) paid by the Bank as a result of the Revolving Commitment, any Advances made by it hereunder, or otherwise in connection with its participation in this Agreement and any liability (including penalties, interest and expenses) arising therefrom or with respect thereto. Payments due under this indemnification shall be made within 30 days of the date the Bank makes demand therefor pursuant to Subsection 2.23 hereof.
- (iv) If the Bank determines that it has actually received or realized any refund of tax, any reduction of, or credit against, its tax liabilities or otherwise recovered any amount in connection with any deduction or withholding, or payment of any additional amount, by the Borrower pursuant to Subsection 2.18 hereof or this Subsection 2.22, the Bank shall reimburse the Borrower an amount that the Bank shall, in its sole discretion, determine is equal to the net benefit, after tax, which was actually obtained by the Bank as a consequence of such refund, reduction, credit or recovery; provided, that nothing in this Subsection 2.22(iv) shall require the Bank to make available its tax returns (or any other information relating to its taxes which it deems to be confidential).
- "2.23 Bank Statements; Survival of Indemnity. To the extent reasonably possible, the Bank shall designate an alternate Lending Installation with respect to its Eurodollar Advances to reduce any liability of the Borrower to the Bank under Subsections 2.18, 2.19 and 2.22 hereof or to avoid the unavailability of Eurodollar Advances under Subsection 2.20 hereof, so long as such designation is not, in the judgment of the Bank, disadvantageous to the Bank. The Bank shall deliver a written statement of the Bank to the Borrower as to the amount due, if any, under Subsection 2.18, 2.19, 2.21 or 2.22 hereof. Such written statement shall set forth in reasonable detail the calculations upon which the Bank determined such amount and shall be final, conclusive and binding on the Borrower in the absence of manifest error. Determination of amounts payable under such Subsections in connection with a Eurodollar Advance shall be calculated as though the Bank funded its Eurodollar Advance through the purchase of a deposit of the type and maturity corresponding to the deposit used as a reference in determining the Eurodollar Rate applicable to such Advance, whether in fact that is the case or not. Unless otherwise provided herein, the amount specified in the written

statement of the Bank shall be payable on demand after receipt by the Borrower of such written statement. The obligations of the Borrower under Subsections 2.18, 2.19, 2.21 and 2.22 hereof shall survive payment of the Indebtedness and termination of this Agreement."

(xiii) Subsection 3 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new Subsection 3:

"3 COLLATERAL. To secure the Indebtedness (including, without limitation, all Advances outstanding under the Revolving Credit Facility, any amount subject to an Interest Rate Swap and any obligation by the Borrower owed to an Affiliate of the Bank), the Borrower shall cause (i) the Parent to execute the Second Amendment and thereby maintain in favor of the Bank pursuant to Section 11 hereof an absolute, unconditional and continuing guaranty of the payment and performance of the Revolving Credit Facility and all other Indebtedness, without limitation as to amount, (ii) the Guarantors (other than the Parent) in existence on the Second Amendment Effective Date to deliver a fully executed Guaranty pursuant to which the Guarantors (other than the Parent), jointly and severally, shall maintain in favor of the Bank an absolute, unconditional and continuing guaranty of the payment and performance of the Revolving Credit Facility and all other Indebtedness, without limitation as to amount, and (iii) each future Material Subsidiary to guarantee the prompt payment and performance when due of the Guaranteed Obligations (as defined in the Guaranty) in accordance with the terms and provisions of the Guaranty. As soon as practicable and in any event within 30 days after any Person becomes a direct or indirect Material Subsidiary, the Borrower shall provide the Bank with written notice thereof and shall cause such Person to execute a Guaranty Supplement in substantially the same form as Schedule I to the Guaranty. Notwithstanding the foregoing, the Bank agrees to release any Material Subsidiary from its obligations under the Guaranty if the Capital Stock of such Subsidiary is sold or transferred pursuant to an Asset Disposition permitted under Section 8.5 of the Bank of Oklahoma Agreement as incorporated into this Agreement pursuant to Section 7 of this Agreement or if such Subsidiary becomes a party to a merger or consolidation in connection with a permitted Asset Disposition (such release to be delivered upon the consummatio

(xiv) Subsection 5.5 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new Subsection 5.5:

"5.5 Financial Condition. The audited consolidated and consolidating balance sheets of the Parent and its consolidated Subsidiaries dated as of September 30, 1999, September 30, 2000, and September 30, 2001, respectively, together, in each case, with the related audited statements of income and cash flows, and the unaudited company-prepared consolidated balance sheet for the Parent and its consolidated Subsidiaries, dated as of March 31, 2002, together

with related consolidated statements of income and cash flows, copies of which have previously been provided to the Bank, were prepared in accordance with GAAP consistently applied throughout the periods covered thereby, are complete and correct in all material respects and present fairly the financial condition (including disclosure of all material liabilities, contingent or otherwise) and results of operations of the Persons and for the periods specified, subject in the case of interim company-prepared statements to normal year-end adjustments and the absence of footnotes. Except as set forth on Schedule 6.5 to the Bank of Oklahoma Agreement and except for the transactions referred to in Section 12 hereof, since September 30, 2001, there has been no circumstance, development or event which has had or could reasonably be expected to have a Material Adverse Effect."

- (xv) Section 6 of the Existing Agreement is hereby amended by deleting Subsections 6.1, 6.2, 6.3, 6.4, 6.5 and 6.6 in their entirety and substituting in lieu thereof the following new Subsections 6.1, 6.2, 6.3, 6.4, 6.5 and 6.6:
- "6.1 Annual Financial Statements and Information. As soon as delivered to the Administrative Agent, but in any event within 95 days after the close of each fiscal year of the Consolidated Group, the Borrower will furnish to the Bank the financial statements and all other information required to be delivered to the Administrative Agent under Section 7.1(a) of the Bank of Oklahoma Agreement.
- "6.2 Quarterly Financial Statements and Information. As soon as delivered to the Administrative Agent, but in any event within 50 days after the close of each fiscal quarter of the Consolidated Group, the Borrower will furnish to the Bank the financial statements and all other information required to be delivered to the Administrative Agent under Section 7.1(b) of the Bank of Oklahoma Agreement.
- "6.3 Other Reports and Notifications.
- "6.3.1 Other Financial Information. With reasonable promptness upon any such request, the Borrower will furnish the Bank with such other information concerning the businesses, operations and financial condition of the Borrower Entities as may be reasonably requested from time to time by the Bank.
- "6.3.2 Other Information. As soon as delivered to the Administrative Agent, but in any event no later than the respective times required under Sections 7.1(c) and 7.1(d) of the Bank of Oklahoma Agreement, all certificates, reports, notices and other information required to be delivered to the Administrative Agent under Sections 7.1(c) and 7.1(d) of the Bank of Oklahoma Agreement.

- "6.3.3 Additional Information. As soon as delivered to the Administrative Agent, but in any event no later than the respective times required under Sections 7.1(e), 7.1(f), 7.1(g), 7.1(h) and 7.1(i) of the Bank of Oklahoma Agreement, all certificates, reports, notices and other information required to be delivered to the Administrative Agent under Sections 7.1(e), 7.1(f), 7.1(g), 7.1(h) and 7.1(i) of the Bank of Oklahoma Agreement.
- "6.3.4 Other Notifications. The Borrower will promptly notify the Bank, but in any event within ten (10) days, after it knows that any of the following has occurred: (i) a Default or an Event of Default; (ii) any change in the assets, liabilities, financial condition, business, operations, affairs or circumstances of any of the Borrower Entities which has or is reasonably likely to have a Material Adverse Effect; (iii) any material change in the accounting practices and procedures of the Borrower Entities, including a change in fiscal year; (iv) any change in the principal place of business of any of the Borrower Entities; or (v) any merger, consolidation or corporate reorganization.
- "6.4 Preservation of Existence and Franchises. Except as a result of or in connection with a merger or consolidation of a Subsidiary permitted under Section 8.3 of the Bank of Oklahoma Agreement or an Asset Disposition involving the Voting Stock of a Subsidiary permitted under Section 8.5 of the Bank of Oklahoma Agreement, each Credit Party will do all things necessary to preserve and keep in full force and effect its existence, rights, franchises and authority.
- "6.5 Use of Proceeds. Advances under the Revolving Credit Facility shall be used by the Borrower for general corporate purposes and for the purposes of (i) financing working capital requirements,
- (ii) financing capital expenditures for the Borrower's worldwide contract drilling and real estate businesses, (iii) retiring debt, and
- (iv) funding company stock repurchase programs.

"6.6 INTENTIONALLY DELETED."

(xvi) Subsection 6.8 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new Subsection 6.8:

"6.8 INTENTIONALLY DELETED."

- (xvii) Section 6 of the Existing Agreement is hereby amended by deleting Subsections 6.11, 6.12 and 6.13 in their entirety and substituting in lieu thereof the following new Subsections 6.11, 6.12 and 6.13:
- "6.11 Additional Affirmative Covenants. The Borrower will perform, comply with and observe for the Bank's benefit, and will cause each Borrower

Entity to perform, comply with and observe for the Bank's benefit, the agreements set forth in Sections 7.3, 7.4, 7.5, 7.6, 7.7, 7.8, 7.11 and 7.12 of the Bank of Oklahoma Agreement. For purposes hereof, all of the provisions of such Sections of the Bank of Oklahoma Agreement, together with related definitions and ancillary provisions, are hereby incorporated herein by reference, mutatis mutandis, and shall be deemed to continue in effect for the Bank's benefit as in effect on the Second Amendment Effective Date, whether or not the Bank of Oklahoma Agreement remains in effect or is amended, waived or otherwise modified by the parties thereto.

"6.12 INTENTIONALLY DELETED.

"6.13 INTENTIONALLY DELETED."

(xviii) Section 7 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new Section 7:

"7. NEGATIVE COVENANTS. Until the Indebtedness has been paid in full and all of the Bank's obligations hereunder have been terminated, the Borrower will, unless the Bank shall otherwise consent in writing, perform, comply with and observe for the Bank's benefit, and will cause each Borrower Entity to perform, comply with and observe for the Bank's benefit, the agreements set forth in Article VIII of the Bank of Oklahoma Agreement as modified pursuant to the next sentence of this Section 7. For purposes hereof, all of the provisions of Article VIII of the Bank of Oklahoma Agreement, together with related definitions and ancillary provisions, are hereby incorporated herein by reference, mutatis mutandis, and shall be deemed to continue in effect for the Bank's benefit as in effect on the Second Amendment Effective Date, whether or not the Bank of Oklahoma Agreement remains in effect or is amended, waived or otherwise modified by the parties thereto; provided, however, for purposes hereof, Section 8.5(iv) of the Bank of Oklahoma Agreement as incorporated herein shall be deemed amended to read as follows:

(iv) the sale, lease, transfer or disposition of any Property by any member of the Consolidated Group to another member of the Consolidated Group, provided that, if the Property is transferred to a member of the Consolidated Group which is not a Guarantor and such transfer would cause it to become a Material Subsidiary, the Borrower shall cause such member to provide a Guaranty Supplement in accordance with the requirements of Section 3 hereof."

(xix) Section 8 of the Existing Agreement is hereby amended by deleting Subsections 8.1 in its entirety and substituting in lieu thereof the following new Subsections 8.1:

- "8.1 Nonpayment of Revolving Note. If the Borrower shall (i) fail to pay any principal of the Revolving Note as and when such payment shall become due and payable (whether at stated maturity, upon a mandatory prepayment, or otherwise), or (ii) fail to pay any interest on the Revolving Note as and when such payment shall become due and payable (whether at stated maturity, upon a mandatory prepayment, or otherwise) and such failure shall continue unremedied for a period of five days; or"
- (xx) Section 8 of the Existing Agreement is hereby amended by deleting Subsections 8.4, 8.5, 8.6, 8.7, 8.8, 8.9, 8.10, 8.11, 8.12, 8.13 and 8.14 in their entirety and substituting in lieu thereof the following new Subsections 8.4, 8.5, 8.6, 8.7, 8.8, 8.9, 8.10, 8.11, 8.12, 8.13 and 8.14:
- "8.4 Breach of Certain Covenants. If there shall occur a (i) default in the due performance or observance of any term, covenant or agreement contained in Subsection 6.4, 6.5 or 6.7 of this Agreement;
- (ii) default in the due performance or observance of any term, covenant or agreement contained in Section 7.4, 7.11 or 7.12 of the Bank of Oklahoma Agreement as incorporated into this Agreement pursuant to Subsection 6.11 of this Agreement; or (iii) default in the due performance or observance of any term, covenant or agreement contained in Section 8.3, 8.4, 8.5, 8.6, 8.7 or 8.8 of the Bank of Oklahoma Agreement as incorporated into this Agreement pursuant to Section 7 of this Agreement; or
- "8.5 Breach of Other Specified Covenants. If there shall occur a
- (i) default in the due performance or observance of any term, covenant or agreement contained in Subsection 6.1, 6.2, 6.3.1 or 6.3.2 of this Agreement and such default shall continue unremedied for a period of at least five days after the earlier of an Executive Officer of a Credit Party becoming aware of such default or notice thereof by the Bank; or (ii) default in the due performance or observance of any term, covenant or agreement contained in Section 8.1 or 8.2 of the Bank of Oklahoma Agreement as incorporated into this Agreement pursuant to Section 7 of this Agreement and such default shall continue unremedied for a period of at least five days after the earlier of an Executive Officer of a Credit Party becoming aware of such default or notice thereof by the Bank; or
- "8.6 Other Breaches. If there shall occur a default in the due performance or observance of any term, covenant or agreement (other than those referred to in Subsection 8.1, 8.2, 8.3, 8.4 or 8.5 of this Agreement) contained in this Agreement or any other Loan Documents and such default shall continue unremedied for a period of at least 30 days after the earlier of an Executive Officer of a Credit Party becoming aware of such default or notice thereof by the Bank; or
- "8.7 Unenforceability of Loan Documents. If any Loan Document or any provision thereof shall for any reason cease to be a valid, binding and enforceable

obligation of the Borrower or any Guarantor, or if the Borrower or any Guarantor shall so state in writing; or

"8.8 Guaranty. Except as the result of or in connection with a merger or disposition of a Guarantor permitted under Section 8.3 or Section 8.5 of the Bank of Oklahoma Agreement as incorporated into this Agreement pursuant to Section 7 hereof, the Guaranty, Section 11 hereof or any provision of either thereof shall cease to be in full force and effect, or any Guarantor or any Person acting by or on behalf of any Guarantor shall deny or disaffirm such Guarantor's obligations under Section 11 hereof or the Guaranty, or any Guarantor shall default in the due performance or observance of any term, covenant or agreement on its part to be performed or observed pursuant to Section 11 hereof or the Guaranty; or

"8.9 Bankruptcy. If any Bankruptcy Event shall occur with respect to any member of the Consolidated Group; or

"8.10 Bank of Oklahoma Agreement Events of Default. If there shall occur any event referred to in Section 9.1(g), 9.1(h), 9.1(i), 9.1(j) or 9.1(k) of the Bank of Oklahoma Agreement. For purposes hereof, all of the provisions of Sections 9.1(g), 9.1(h), 9.1(i), 9.1(j) and 9.1(k) of the Bank of Oklahoma Agreement, together with related definitions and ancillary provisions, are hereby incorporated herein by reference, mutatis mutandis, and shall be deemed to continue in effect for the Bank's benefit as in effect on the Second Amendment Effective Date, whether or not the Bank of Oklahoma Agreement remains in effect or is amended, waived or otherwise modified by the parties thereto.

"8.11 INTENTIONALLY DELETED.

"8.12 INTENTIONALLY DELETED.

"8.13 INTENTIONALLY DELETED.

"8.14 INTENTIONALLY DELETED."

(xxi) Subsection 9.1 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new Subsection 9.1:

"9.1 Acceleration. If any Bankruptcy Event shall occur with respect to the Borrower, the obligations of the Bank hereunder (including the Revolving Commitment) shall automatically be terminated and the Revolving Note and all other obligations of the Borrower hereunder or under any of the other Loan Documents shall become immediately due and payable, all with notice or demand. If any other Event of Default shall occur, the Bank may, at its option, without notice or demand, terminate its obligations hereunder (including the Revolving Commitment) and declare the Revolving Note and all other obligations of the Borrower hereunder or under any of the other Loan Documents to be

immediately due and payable, whereupon the same shall become forthwith due and payable."

(xxii) Subsection 10.3 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new Subsection 10.9:

"10.3 Notices. All notices, requests and demands required or authorized hereunder (other than Disbursement Requests and Conversion/Continuation Notices) shall be served in person, delivered by certified mail, return receipt requested, or transmitted by telefacsimile, addressed as follows:

Borrower: Helmerich & Payne International Drilling Co.

c/o Helmerich & Payne, Inc. Utica at Twenty-First Tulsa, Oklahoma 74114

Attn: Chief Financial Officer (with a copy

to the General Counsel)

Fax: 918-743-2671

Parent: Helmerich & Payne, Inc.

Utica at Twenty-First Tulsa, Oklahoma 74114

Attn: Chief Financial Officer (with a copy

to the General Counsel)

Fax: 918-743-2671

The Bank: Bank One, Oklahoma, N.A.

c/o Bank One, NA
Mail Code IL1-0362
One Bank One Plaza

Chicago, Illinois 60707-0362

Attn: Kenneth J. Fatur Fax: 312-732-3055

or at such other address as any party hereto shall designate for such purpose in a written notice to the other parties hereto. Any notice given hereunder to the Bank or the Borrower shall also constitute effective notice to each of the participating lenders or the Borrower Entities, respectively. Notices served in person shall be effective and deemed given when delivered, notices sent by certified mail shall be effective and deemed given three (3) Business Days after being deposited in the U.S. mail, postage prepaid, and notices transmitted by telefacsimile will be deemed given when sent, as indicated by the sender's written confirmation of transmission."

(xxiii) Subsection 10.9 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new Subsection 10.9:

"10.9 INTENTIONALLY DELETED."

(xxiv) Subsection 10.11 of the Existing Agreement is hereby amended by deleting it in its entirety and substituting in lieu thereof the following new Subsection 10.11:

"10.11 Jurisdiction and Venue. All actions or proceedings with respect to this Agreement or any of the other Loan Documents may be instituted in any state or federal court sitting in Tulsa, Oklahoma, as the Bank may elect, and by execution and delivery of this Agreement, the Borrower irrevocably and unconditionally (i) submits to the non-exclusive jurisdiction (both subject matter and person) of each such court, and (ii) waives (A) any objection that the Borrower may now or hereafter have to the laying of venue in any of such courts, and (B) any claim that any action or proceeding brought in any such court has been brought in an inconvenient forum."

(xxv) The Existing Agreement is hereby amended by adding thereto after Section 10 of the Existing Agreement the following new Sections 11 and 12:

"11. PARENT GUARANTY.

"11.1 Guaranty. The Parent hereby guarantees absolutely unconditionally and irrevocably, and without limitation as to amount, the prompt performance and payment when due (whether at a stated maturity or earlier by reason of acceleration or otherwise) of the Revolving Credit Facility and all Advances made thereunder, and all other indebtedness, liabilities and obligations now or hereafter owing by the Borrower to the Bank under the Credit Agreement, the Revolving Note and the other Loan Documents, and any amount due under any Interest Rate Swap with the Bank or any of its Affiliates, including, without limitation, principal, interest, fees, reasonable attorney's fees, filing and recording costs, out-of-pocket expenses, collection costs, all interest, fees and other monetary obligations incurred or accrued during the pendency of any bankruptcy, insolvency, receivership or other similar proceeding, regardless of whether allowed or allowable in such proceeding, and all renewals, extensions and modifications of any of the foregoing (all of the foregoing indebtedness, liabilities and obligations being hereinafter referred to as the "Obligations"), as and when the same shall become due and payable, whether at the stated maturity, upon acceleration or otherwise, in accordance with the terms hereof and thereof. If the Borrower fails to pay when due any Obligation guaranteed hereby, the Parent unconditionally agrees to cause such payment to be made punctually as and when the same shall become due and payable, whether at the stated maturity, upon acceleration or otherwise.

"11.2 Guarantee Unconditional. The obligations of the Parent under this Section 11 are absolute and unconditional. Without limiting the generality of the

foregoing, the obligations of the Parent under this Section 11 shall not be impaired, released, discharged or otherwise affected by:

- "(i) any extension, renewal, settlement, compromise, waiver or release in respect of any obligation of the Borrower or any other Credit Party under this Agreement or any other Loan Document, by operation of law or otherwise;
- "(ii) any modification, amendment or waiver of or supplement to this Agreement or any Loan Document;
- "(iii) any release, impairment or invalidity of any guarantee or other liability of any other Credit Party or third party for any obligation of the Borrower under this Agreement or any other Loan Document;
- "(iv) any change in the corporate existence, structure or ownership of the Borrower or any other Credit Party or any insolvency, bankruptcy, reorganization or other similar proceeding affecting the Borrower or any other Credit Party;
- "(v) the existence of any claim, set-off or other rights which the Parent may have at any time against the Borrower or any other Credit Party, the Bank or any other Person, whether or not arising in connection with the Advances and this Agreement or any unrelated transaction;
- "(vi) any invalidity or unenforceability relating to or against the Borrower or any other Credit Party for any reason of this Agreement or any other Loan Document, or any provision of applicable law or regulation purporting to prohibit the payment by any other Credit Party of any amount payable by it under this Agreement or any other Loan Document; or
- "(vii) any other act or omission to act or delay of any kind by any other Credit Party, the Bank or any other Person or any other circumstance which might, but for the provisions of this Section 11.2, constitute a legal or equitable discharge of the Parent's obligations under this Section 11.
- "11.3 Discharge Only Upon Payment in Full; Reinstatement in Certain Circumstances. The Parent's obligations under this Section 11 constitute a continuing guaranty and shall remain in full force and effect until the Revolving Commitment shall have been terminated, and all amounts payable under this Agreement and the Loan Documents shall have been indefeasibly paid in full. If at any time any amount payable by the Borrower under this Agreement or any other Loan Document is rescinded or must be otherwise restored or returned upon the insolvency, bankruptcy or reorganization of the Borrower or any other Credit Party or otherwise, the Parent's obligations under this Section 11 with respect to such payment shall be reinstated at such time as though such payment had become due but had not been made at such time.

- "11.4 Waiver. The Parent irrevocably waives acceptance hereof, presentment, demand, protest and any notice not provided for herein, as well as any requirement that at any time any action be taken by any Person against the Borrower or other Credit Party or any other Person.
- "11.5 Subrogation. If the Parent makes any payment under this

Section 11 with respect to the obligations of the Borrower, the Parent shall be subrogated to the rights of the payee against the Borrower with respect to the portion of such obligations paid by the Parent; provided that the Parent shall not enforce any payment by way of subrogation or contribution against the Borrower so long as any amount payable under this Agreement or any other Loan Document remains unpaid.

- "11.6 Stay of Acceleration. If acceleration of the time for payment of any amount payable by the Borrower under this Agreement or any other Loan Document is stayed upon the insolvency, bankruptcy or reorganization of the Borrower, all such amounts otherwise subject to acceleration under the terms of such Loan Document shall nonetheless be payable by the Parent under this Section 11 forthwith on demand by the Bank.
- "11.7 Successors and Assigns. The guarantee by the Parent under this Section 11 is for the benefit of the Bank and its successors and assigns. If any Advances or other amounts payable under this Agreement and the other Loan Documents are assigned to any other Person, the rights of the Bank under this Section 11, to the extent applicable to the indebtedness so assigned, shall be transferred with such indebtedness.

"12. SPIN-OFF/MERGER; OTHER PROPOSED TRANSACTIONS.

- "12.1 Spin-off/Merger. Anything in this Agreement to the contrary notwithstanding, nothing in this Agreement shall prohibit, and no Material Adverse Effect or Default or Event of Default hereunder shall result or be deemed to result from, (i) the consummation by the Parent of the Spin-off/Merger or any transaction contemplated thereby, or
- (ii) in the event the Spin-off/Merger is not consummated, the sale or other disposition of the oil and gas exploration and production business of the Consolidated Group and/or the stock of Cimarex, Helmerich & Payne Energy Services, Inc. or Mountain Acquisition Co.
- "12.2 Other Proposed Transactions. The parties acknowledge that one or more members of the Consolidated Group intend to undertake and consummate the following proposed transactions (the "Proposed Transactions"): (i) the sale and issuance of the Medium Term Notes, and (ii) an internal reorganization of the Consolidated Group pursuant to which (x) substantially all of the Properties of the Parent (other than those primarily related to the oil and gas exploration, production, marketing and sales operations of the Parent) will be transferred to the Borrower or one or more Subsidiaries of the Borrower and (y) the current direct and indirect

Subsidiaries of the Parent will become direct or indirect subsidiaries of the Borrower. The parties agree that, notwithstanding any provision of this Agreement to the contrary, provided that the Proposed Transactions are carried out without a breach of the covenants in Sections 7.11 or 7.12 and, with respect to the internal reorganization, in compliance with the requirements of Section 7.13 (if applicable), none of the Proposed Transactions will be deemed to have a Material Adverse Effect or otherwise to constitute or create a Default or Event of Default under this Agreement."

(xxvi) The Existing Agreement is hereby amended by deleting in its entirety Exhibit A" to the Existing Agreement (Form of Revolving Note) and substituting in lieu thereof Exhibit "A" attached hereto (Form of Revolving Note).

SECTION 3. REPRESENTATIONS AND WARRANTIES. In order to induce the Bank to execute and deliver this Second Amendment, the Borrower hereby confirms, reaffirms and restates as of the Second Amendment Effective Date the representations and warranties set forth in Section 5 of the Existing Agreement as amended by this Second Amendment provided that such representations and warranties shall be and hereby are amended as follows: each reference therein to "this Agreement" (including, without limitation, each such a reference included in the term "Loan Documents" and all indirect references such as "hereby", "herein", "hereof" and "hereunder") shall be deemed to be a collective reference to the Existing Agreement, this Second Amendment and the Existing Agreement as amended by this Second Amendment. An Event of Default under and as defined in the Existing Agreement as amended by this Second Amendment shall be deemed to have occurred if any representation or warranty made pursuant to the foregoing sentence of this Section 3 shall be materially false as of the date on which made.

SECTION 4. RELEASE OF CERTAIN GUARANTORS. In consideration of the execution and delivery of the Guaranty (as defined in the Existing Agreement as amended by this Second Amendment) by the Guarantors (as defined in the Existing Agreement as amended by this Second Amendment) and effective only upon the satisfaction of the conditions precedent set forth in Section 6 of this Second Amendment, the Bank hereby releases each of the Borrower, The Space Center, Inc., an Oklahoma corporation, and Utica Square Shopping Center, Inc., an Oklahoma corporation, from any and all liability and obligation under the Guaranty Agreement dated of October 27, 1998 (the "Existing Guaranty") executed and delivered by the guarantors party thereto pursuant to the Existing Agreement; provided, however, it is expressly agreed and understood that the release provided for in this

Section 4 shall not and does not modify or impair, or release the Borrower from, any of its obligations under the Assignment and Assumption Agreement, the Second Amendment, the Existing Agreement as amended by the Second Amendment or any of the other Loan Documents other than the Existing Guaranty.

SECTION 5. ADVANCES TO BORROWER. The Parent, the Borrower and the Bank hereby agree that simultaneously with the effectiveness of this Second Amendment in accordance with the provisions of Section 6 hereof and notwithstanding anything to the contrary set forth in the Existing Agreement or the Existing Agreement as amended by this Second Amendment, (i) the Parent shall be deemed to have paid to the Bank the \$50,000,000 principal amount (but not any interest accrued and unpaid thereon) of the Advances outstanding under the Existing Agreement

on the Second Amendment Effective Date (as defined in Section 6 hereof) without paying or being obligated to pay any funding indemnification, penalty or premium under either Section 2.12 of the Existing Agreement or Section 2.21 of the Existing Agreement as amended pursuant to this Second Amendment, (ii) the Bank shall be deemed to have made a \$50,000,000 Advance to the Borrower on the Second Amendment Effective Date, and (iii) the Parent has assigned to the Borrower, and the Borrower has assumed, the indebtedness, obligations and liabilities (including, without limitation, the obligation to pay all interest accrued but unpaid on the \$50,000,000 of Advances outstanding under the Existing Agreement on the Second Amendment Effective Date) of the Parent under the Existing Agreement and the other Loan Documents (as defined in the Existing Agreement) pursuant to the Assignment and Assumption Agreement referred to below. The Bank shall make appropriate entries in its books and records to reflect the provisions of this Section 5 and such entries shall be the presumptive evidence of the principal amount owing under the Revolving Credit Facility.

SECTION 6. CONDITIONS PRECEDENT. This Second Amendment and the amendments to the Existing Agreement provided for herein shall become effective as of the date (the "Second Amendment Effective Date") on which all of the following conditions precedent are satisfied:

- (a) This Second Amendment shall have been duly executed and delivered by the Bank and the Bank shall have received a counterpart of this Second Amendment duly executed and delivered by the Borrower and the Parent.
- (b) The Parent and the Borrower shall have duly executed and delivered to the Bank an Assignment and Assumption Agreement dated as of the Second Amendment Effective Date in substantially the form of Exhibit "B" attached hereto (the "Assignment") pursuant to which the Borrower becomes the Borrower under the Agreement, and the Bank shall have duly executed and delivered such Assignment.
- (c) The Borrower shall have duly executed and delivered to the Bank a Revolving Note (as defined in the Existing Agreement as amend by this Second Amendment). Promptly after its receipt of such new Revolving Note, the Bank will mark the existing Revolving Note delivered by the Parent at the Closing under the Existing Agreement "Cancelled" and return it to the Parent.
- (d) The Guarantors (as defined in the Existing Agreement as amended by this Second Amendment) shall have duly executed and delivered to the Bank a Guaranty Agreement in substantially the form of Exhibit "C" attached hereto, and the Bank shall have received counterparts of such Guaranty Agreement duly executed and delivered by such Guarantors.
- (e) The Borrower shall have furnished to the Bank:
- (i) A copy, certified as of the Second Amendment Effective Date by the Secretary or an Assistant Secretary of the Borrower, of resolutions of the Borrower's Board of Directors authorizing the Borrower's execution and delivery of

the Assignment, this Second Amendment and the New Revolving Note referred to above.

- (ii) A copy, certified as of the Second Amendment Effective Date by the Secretary or an Assistant Secretary of the Borrower, of (A) the Borrower's certificate of incorporation and all amendments thereto, and (B) the Borrower's by-laws and all amendments thereto.
- (iii) An incumbency certificate, executed by the Secretary or an Assistant Secretary of the Borrower as of the Second Amendment Effective Date, which shall identify by name and title and bear the signatures of the officers of the Borrower authorized to act on behalf of the Borrower with respect to the Assignment, this Second Amendment, the new Revolving Note referred to above and the Existing Agreement as amended by this Second Amendment, upon which certificate the Bank shall be entitled to rely until informed of any change in writing by the Borrower.
- (iv) Opinions of McAfee & Taft A Professional Corporation, counsel to the Borrower, and Steven R. Mackey, General Counsel of the Parent, such matters with respect to the Assignment, this Second Amendment and the new Revolving Note referred to above as the Bank or its counsel may reasonably request (such opinions may be based upon such customary assumptions and may contains such customary qualifications and exceptions as may be approved by the Bank and its counsel).
- (v) Such other documents as the Bank or its counsel may have reasonably requested.
- (f) The Parent shall have furnished to the Bank:
- (i) A copy, certified as of the Second Amendment Effective Date by the Secretary or an Assistant Secretary of the Parent, of resolutions of the Parent's Board of Directors authorizing the Parent's execution and delivery of this Second Amendment and its guaranty pursuant to Section 11 of the Existing Agreement as amended hereby of the Borrower's obligations under the Existing Agreement as amended hereby.
- (ii) An incumbency certificate, executed by the Secretary or an Assistant Secretary of the Parent as of the Second Amendment Effective Date, which shall identify by name and title and bear the signatures of the officers of the Parent authorized to act on behalf of the Parent with respect to this Second Amendment and its guaranty pursuant to Section 11 of the Existing Agreement as amended hereby of the Parent's obligations under the Existing Agreement as amended hereby, upon which certificate the Bank shall be entitled to rely until informed of any change in writing by the Parent.

- (g) Each Guarantor (other than the Parent) shall have furnished to the Bank:
- (i) A copy, certified as of the Second Amendment Effective Date by the Secretary or an Assistant Secretary of such Guarantor, of resolutions of the Board of Directors of such Guarantor authorizing the execution and delivery of the Guaranty Agreement by such Guarantor.
- (ii) An incumbency certificate, executed by the Secretary or an Assistant Secretary of such Guarantor as of the Second Amendment Effective Date, which shall identify by name and title and bear the signatures of the officers of such Guarantor authorized to act on behalf of such Guarantor with respect to the Guaranty Agreement, upon which certificate the Bank shall be entitled to rely until informed of any change in writing by such Guarantor.
- (h) No Default or Event of Default shall have occurred and be continuing.

SECTION 7. EFFECT ON THE EXISTING AGREEMENT. Except as expressly amended hereby, all of the representations, warranties, terms, covenants and conditions of the Existing Agreement and the other Loan Documents (i) shall remain unaltered, (ii) shall continue to be, and shall remain, in full force and effect in accordance with their respective terms, and (iii) are hereby ratified and confirmed in all respects. Upon the effectiveness of this Second Amendment, all references in the Existing Agreement (including references in the Existing Agreement as amended by this Second Amendment) to "this Agreement" (including, without limitation, each such a reference included in the term "Loan Documents" and all indirect references such as "hereby", "herein", "hereof" and "hereunder") shall be deemed to be a collective reference to the Existing Agreement as amended by this Second Amendment.

SECTION 8. EXPENSES. The Borrower shall reimburse the Bank for any and all reasonable costs, internal charges and out-of-pocket expenses (including attorneys' fees and time charges of attorneys for the Bank, which attorneys may be employees of the Bank) paid or incurred by the Bank in connection with the preparation, review, execution and delivery of this Second Amendment.

SECTION 9. ENTIRE AGREEMENT. This Second Amendment, the Existing Agreement as amended by this Second Amendment and the other Loan Documents embody the entire agreement and understanding between the parties hereto and supersede any and all prior agreements and understandings between the parties hereto relating to the subject matter hereof.

SECTION 10. HEADINGS. The headings, captions, and arrangements used in this Second Amendment are for convenience only and shall not affect the interpretation of this Second Amendment.

SECTION 11. GOVERNING LAW. THIS SECOND AMENDMENT SHALL BE DEEMED TO BE A CONTRACT MADE UNDER THE LAWS OF THE STATE OF OKLAHOMA, AND ALL RIGHTS AND INDEBTEDNESS HEREUNDER, INCLUDING

MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE, SHALL BE GOVERNED BY THE LAWS OF THE STATE OF OKLAHOMA.

SECTION 12. COUNTERPARTS. This Second Amendment may be executed in any number of counterparts, all of which taken together shall constitute one agreement, and any of the parties hereto may execute this Second Amendment by signing any such counterpart.

IN WITNESS WHEREOF, the parties hereto have caused this Second Amendment to be duly executed as of the date first above written.

HELMERICH & PAYNE
INTERNATIONAL DRILLING CO.

HELMERICH & PAYNE INTERNATIONAL DRILLING CO.
<u>By:</u>
Title: Vice President
HELMERICH & PAYNE, INC.
<u>By:</u>
Title: Vice President
BANK ONE, OKLAHOMA, N.A.
<u>By:</u>
<u>Title:</u>
Pursuant to subparagraph 7(b) of that certain Participation Agreement dated October 29, 1998 between Bank One, Oklahoma, N.A. and Bank One, NA (main office Chicago), as successor by merger to Bank One, Texas, N.A., Bank One, NA (main office Chicago) hereby consents to the transactions and the amendments and other modifications to the Credit Agreement and the other Loan Documents provided for in the foregoing Second Amendment and the Assignment and Assumption Agreement referred to therein.
BANK ONE, NA (main office Chicago)
<u>By:</u>
Title:

Page 30

CREDIT AGREEMENT

dated July 16, 2002

among

HELMERICH & PAYNE INTERNATIONAL DRILLING CO.

HELMERICH & PAYNE, INC.

THE SEVERAL LENDERS FROM TIME TO TIME PARTY HERETO

and

BANK OF OKLAHOMA, NATIONAL ASSOCIATION, as Administrative Agent

TABLE OF CONTENTS

		Page
ADMICI II	I - DEFINITIONS	
	Definitions	
1.1	Construction and Interpretation	
1.2	Construction and Interpretation	
	II - TERMS OF THE CREDIT FACILITY	
2.1	Commitments	
2.2	Method of Borrowing	
2.3	Continuation and Conversion	
	Notes	
2.5	Reduction, Termination and Extension of Revolving Commitments	
2.6	Additional Provisions Relating to Letters of Credit	
2.7	Additional Provisions Relating to Swingline Loans	
2.8	Pro Rata Treatment	
2.9	Records of Loans	
	III - INTEREST, FEES AND REPAYMENT	
3.1	Interest	
3.2	Fees	
3.3	Maturity; Repayment	
3.4	Prepayments	
3.5	Sharing of Payments	
3.6	Payments	
3.7	Computations of Interest and Fees	
3.8	Maximum Lawful Interest Rate	
	IV - YIELD PROTECTION; OTHER PROVISIONS RELATING TO CREDIT FACILITY	
4.1	Capital Adequacy	
4.2	Limitation on LIBOR Rate Loans	
4.3	Illegality	
4.4	Requirements of Law	
4.5	Treatment of Affected Loans	36
4.6	Taxes	36
4.7	Compensation	
4.8	Replacement of Lenders	
ARTICLE	V - CLOSING; CONDITIONS PRECEDENT	
5.1	Closing Date	40
5.2	Closing Conditions	40
5.3	Conditions to All Extensions of Credit	42
ARTICLE	VI - REPRESENTATIONS AND WARRANTIES	43
6.1	Organization; Existence; Compliance with Law	43
6.2	Existence and Authorization	43
6.3	Enforceability	44

6.4	No Legal Bar
6.5	Financial Condition44
6.6	No Material Litigation and Disputes44
6.7	No Defaults
6.8	Ownership and Operation of Property45
6.9	Licenses
6.10	No Burdensome Restrictions45
6.11	Taxes
6.12	ERISA Matters45
6.13	Governmental Regulations46
6.14	Subsidiaries
6.15	Purpose of Extensions of Credit47
6.16	Environmental Matters48
6.17	Labor Matters48
6.18	No Material Misstatements
ARTICLE	VII - AFFIRMATIVE COVENANTS
7.1	Reporting Requirements
7.2	Preservation of Existence and Franchises51
7.3	Books and Records
7.4	Compliance with Law
7.5	Payment of Taxes and Other Indebtedness
7.6	Insurance
7.7	Maintenance of Property52
7.8	Performance of Obligations
7.9	Use of Proceeds
7.10	Audits/Inspections
7.11	Financial Covenants53
7.12	Liquidity Maintenance
7.13	Additional Guarantors53
ARTICLE	VIII - NEGATIVE COVENANTS
8.1	Indebtedness54
8.2	Liens54
8.3	Merger and Consolidation55
8.4	Dissolution
8.5	Asset Dispositions
8.6	Transactions with Affiliates
8.7	Fiscal Year; Organizational Documents
8.9	Contingent Debt
ARTICLE	IX - EVENTS OF DEFAULT
9.1	Events of Default
9.2	Acceleration; Remedies
9.3	Waiver of Event of Default
9.4	Allocation of Payments After Event of Default
ARTICLE	X - AGENCY PROVISIONS
10.1	Appointment Powers and Immunities

10.2	Reliance by Administrative Agent	.63
10.3	Defaults	.64
10.4	Rights as a Lender	.64
10.5	Indemnification	.64
10.6	Non-Reliance on Administrative Agent and Other Lenders	.65
10.7	Successor Administrative Agent	.65
ARTICLE	XI - PARENT GUARANTY	.65
ARTICLE	XII - GENERAL PROVISIONS	.67
12.1	Notices	.67
12.2	Right of Set-Off; Adjustments	.68
12.3	Benefit of Agreement	.69
12.4	No Waiver; Remedies Cumulative	.71
12.5	Expenses; Indemnification	.71
12.6	Amendments, Waivers and Consents	.72
12.7	Counterparts	.74
12.8	Headings	.74
12.9	Survival	.74
12.10	Governing Law	.74
12.11	Waiver of Jury Trial	.74
12.12	Severability	.74
12.13	Entire Agreement	.75
12.14	Binding Effect	.75
12.15	Termination	.75
12.16	Confidentiality	.75
12.17	Spinoff/Merger; Other Proposed Transactions	.76

SCHEDULES AND EXHIBITS

SCHEDULES

Schedule 1.1 Schedule 2.6(a) Schedule 6.5 Schedule 6.14 Schedule 6.17 Schedule 7.6	Material Subsidiaries Existing Letters of Credit Subsequent Events Subsidiaries Labor Matters Insurance
Exhibit "A" Exhibit "A-1" Exhibit "B" Exhibit "C" Exhibit "D" Exhibit "E" Exhibit "F" Exhibit "G" Exhibit "H"	Form of Promissory Note Form of Swingline Note Form of Subsidiary Guaranty Agreement Form of Notice of Borrowing Form of Notice of Request for Letter of Credit Form of Notice of Continuation/Conversion Form of Opinion Form of Compliance Certificate Form of Assignment and Acceptance

CREDIT AGREEMENT

THIS CREDIT AGREEMENT (this "Agreement"), dated July 16, 2002, is entered into by and among HELMERICH & PAYNE INTERNATIONAL DRILLING CO., a Delaware corporation (the "Borrower"), HELMERICH & PAYNE, INC., a Delaware corporation (the "Parent"), the Lenders (as hereinafter defined), and BANK OF OKLAHOMA, NATIONAL ASSOCIATION, as Administrative Agent for the Lenders (in such capacity, the "Administrative Agent"), with reference to the following:

- A. The Borrower is primarily engaged in the business of the contract drilling of oil and gas wells for others.
- B. The Borrower has requested that the Lenders provide a revolving credit facility in the maximum principal amount of \$125,000,000 to be used for the purposes hereinafter stated.
- C. The Lenders have agreed to establish the requested revolving credit facility in favor of the Borrower on the terms and conditions hereinafter set forth.

NOW, THEREFORE, in consideration of the foregoing and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

ARTICLE I DEFINITIONS

- 1.1 Definitions. As used in this Agreement, the following terms have the meanings specified below, unless the context otherwise requires:
- "Adjusted LIBOR Rate" means, with respect to any LIBOR Rate Loan for any Interest Period therefor, the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) determined by the Administrative Agent to be equal to the quotient obtained by dividing (a) the LIBOR Rate for such LIBOR Rate Loan for such Interest Period by (b) 1 minus the Eurodollar Reserve Requirement for such LIBOR Rate Loan for such Interest Period.
- "Administrative Agent Fee Letter" means the letter agreement dated as of the date hereof between the Borrower and the Administrative Agent, as amended, modified, restated or supplemented from time to time.
- "Affiliate" means, with respect to any Person, any other Person (i) directly or indirectly controlling or controlled by or under direct or indirect common control with such Person or (ii) directly or indirectly owning or holding five percent (5%) or more of the Capital Stock in such Person. For purposes of this definition, "control" means when used with respect to any Person, the power to direct the management and policies of such Person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise.

"Applicable Margins" means, for any day, the rates per annum set forth below opposite the applicable Consolidated Debt to EBITDA Ratio, it being understood that (i) the Applicable Margin for LIBOR Rate Loans shall be the percentage set forth under the column "LIBOR Rate Loans," and (ii) the Applicable Margin for Prime Rate Loans shall be the percentage set forth under the column "Prime Rate Loans."

Pricing Level	Consolidated Debt to EBITDA Ratio	LIBOR Rate Loans	Prime Rate Loans
Tier I Tier II	Less than 2.0:1.0 Greater than or equal to	(Plus) 0.875% (87.5 bps) (Plus) 1.125% (112.5 bps)	(Minus) 1.750% (175 bps) (Minus) 1.500% (150 bps)
	2 0 1 0		

Until the first Rate Determination Date, the Applicable Margins shall those applicable to the "Tier I Pricing Level" shown in the table above. The Applicable Margins shall be determined and adjusted, if appropriate, quarterly on each Rate Determination Date; provided that in the event the applicable Required Financial Information is not delivered to the Administrative Agent by the date required by Section 7.1(a) or 7.1(b), as applicable, the Applicable Margins shall be based on the highest pricing level until such time as the Required Financial Information is delivered. Subject to such proviso, each Applicable Margin shall be effective from one Rate Determination Date until the next Rate Determination Date. The Administrative Agent shall determine the appropriate Applicable Margins promptly upon receipt of the Required Financial Information and shall promptly notify the Borrower and the Lenders of any changes therein. Such determinations by the Administrative Agent shall be conclusive, absent manifest error. Adjustments in the Applicable Margins shall be effective as to existing Extensions of Credit as well as new Extensions of Credit made thereafter.

[&]quot;Aggregate Revolving Committed Amount" means the aggregate amount of Revolving Commitments in effect from time to time, as referenced in Section 2.1(a), as such amounts may be reduced from time to time pursuant to Section 2.5. The initial Aggregate Revolving Committed Amount is \$125,000,000.

[&]quot;Agreement" (and such terms as "herein," "hereof," "hereto," "hereby," "hereunder" and the like) means and refers to this Credit Agreement, together with all exhibits and schedules attached hereto, as it may be amended, modified, restated or supplemented from time to time,

[&]quot;Applicable Lending Office" means, for each Lender, the office of such Lender (or of an Affiliate of such Lender) as such Lender may from time to time specify to the Administrative Agent and the Borrower by written notice as the office at which its LIBOR Rate Loans are made and maintained.

[&]quot;Asset Disposition" means and includes the sale, lease (including any Sale/Leaseback Transaction), transfer or other disposition of any Property by a member of the Consolidated

Group (including the Capital Stock of a Subsidiary), whether pursuant to a single transaction or a series of related transactions.

"Bank of Oklahoma" means Bank of Oklahoma, National Association, and its successors.

"Bank One Agreement" means the Credit Agreement dated as of October 27, 1998, between the Parent and Bank One, Oklahoma, N.A., as amended pursuant to a First Amendment to Credit Agreement dated as of May 20, 1999, as assigned to and assumed by the Borrower pursuant to an Assignment and Assumption Agreement dated as of July 16, 2002, among the Parent, the Borrower and Bank One, Oklahoma, N.A., as further amended pursuant to a Second Amendment dated as of July 16, 2002, among the Parent, the Borrower and Bank One, Oklahoma, N.A., and as it may be further amended, supplemented, assigned or otherwise modified from time to time.

"Bankruptcy Code" means the United States Bankruptcy Code (Title 11 of the United States Code), as amended, modified, succeeded or replaced from time to time.

"Bankruptcy Event" means, with respect to any Person, the occurrence of any of the following with respect to such Person: (i) a court or governmental agency having jurisdiction in the premises shall enter a decree or order for relief in respect of such Person in an involuntary case under the Bankruptcy Code or any other applicable insolvency or other similar Law now or hereafter in effect, or appointing a receiver, liquidator, assignee, custodian, trustee, sequestrator (or similar official) of such Person or for any substantial part of its Property or ordering the winding up or liquidation of its affairs; or (ii) there shall be commenced against such Person an involuntary case under the Bankruptcy Code or any other applicable insolvency or other similar Law now or hereafter in effect, or any case, proceeding or other action for the appointment of a receiver, liquidator, assignee, custodian, trustee, sequestrator (or similar official) of such Person or for any substantial part of its Property or for the winding up or liquidation of its affairs, and such involuntary case or other case, proceeding or other action shall remain undismissed, undischarged or unbonded for a period of 90 consecutive days; or (iii) such Person shall commence a voluntary case under the Bankruptcy Code or any other applicable insolvency or other similar Law now or hereafter in effect, or consent to the entry of an order for relief in an involuntary case under any such law, or consent to the appointment or taking possession by a receiver, liquidator, assignee, custodian, trustee, sequestrator (or similar official) of such Person or for any substantial part of its Property or make any general assignment for the benefit of creditors; or (iv) such Person shall be unable to pay or shall fail to pay, or shall admit in writing its inability to pay, its debts generally as they become due.

"Business Day" means a day other than a Saturday, Sunday or other day on which commercial banks in Tulsa, Oklahoma, or New York, New York, are authorized or required by law to be closed, except that, when used in connection with a LIBOR Rate Loan, such day shall also be a day on which dealings between banks are carried on in Dollar deposits in London, England.

"Capital Lease" means, as applied to any Person, any lease of any Property (whether real, personal or mixed) by that Person as lessee which, in accordance with GAAP, is or should be accounted for as a capital lease on the balance sheet of that Person.

"Capital Stock" means (i) in the case of a corporation, capital stock,

(ii) in the case of a partnership, partnership interests (whether general or limited), (iii) in the case of a limited liability company, membership interests, (iv) in the case of an association or other business entity, shares, interests, participations, rights or other equivalents (however designated) of capital stock, and (v) any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distributions of assets of, the issuing Person.

"Change of Control" means (i) the acquisition by any Person, or two or more Persons acting in concert, of beneficial ownership (within the meaning of Rule 13d-3 of the SEC under the Securities Exchange Act) of 35% or more of the outstanding shares of Voting Stock of the Parent, (ii) the existence of any circumstance or the occurrence of any event whereby the individuals who comprise the Incumbent Board shall cease for any reason to constitute at least two-thirds of the voting members of the Board of Directors of the Parent, or (iii) the failure of the Borrower to be and remain a wholly-owned Subsidiary of the Parent (unless the Borrower and the Parent have been merged or consolidated with each other in a merger or consolidation permitted by Section 8.3(ii)). For purposes of this definition, the "Incumbent Board" means the individuals who are members of the Board of Directors of the Parent as of the Closing Date and any individual who is hereafter elected to the Board of Directors by the Parent's common stockholders after his or her nomination for election as a new director is approved by a vote of at least two-thirds of the Incumbent Board, provided, however, that no individual shall be considered a member of the Incumbent Board if such individual initially assumes office as a result of either an actual or threatened "election contest" (as described in Rule 14A-11 of the SEC promulgated under the Securities Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board of Directors, including an individual elected by reason of any agreement intended to avoid or settle any election contest or proxy contest.

"Charter Documents" means, with respect to any Credit Party, its Articles or Certificate of Incorporation, Articles of Organization or other similar publicly-filed organizational documents, and its bylaws, partnership agreement, operating agreement and other similar internal governance documents, as the same may be amended or modified from time to time (subject to the terms of this Agreement).

"Cimarex" means Cimarex Energy Co., a Delaware corporation, formerly known as Helmerich & Payne Exploration and Production Co.

"Closing Date" means the time and date, as specified in Section 5.1, on which the initial Extensions of Credit are made hereunder.

"Commitment Period" means the period from and including the Closing Date to but not including the earlier of (i) the Revolving Commitment Termination Date, or (ii) the date on

which the Revolving Commitments terminate in accordance with the provisions of this Agreement.

"Commitments" means any of the Revolving Commitments, the LOC Commitments or the Swingline Commitments.

"Committed Amount" means any of the Revolving Committed Amount, the LOC Committed Amount or the Swingline Committed Amount, as applicable.

"Consolidated Debt to Capitalization Ratio" means, for the Consolidated Group as of the last day of each fiscal quarter, the ratio of (i) Consolidated Total Funded Debt on such day, minus the aggregate amount of any Funded Debt incurred as the direct result of Forward Sale Contracts relating to securities held in the Investment Portfolio, as long as such Funded Debt is fully secured by Marketable Securities, to (ii) Consolidated Total Capitalization on such day.

"Consolidated Debt to EBITDA Ratio" means, for the Consolidated Group as of the last day of each fiscal quarter, the ratio of (i) Consolidated Total Funded Debt on such day, minus the sum of (A) the aggregate amount of any Funded Debt incurred as the direct result of Forward Sale Contracts relating to securities held in the Investment Portfolio, as long as such Funded Debt is fully secured by Marketable Securities, and (B) an amount equal to the after-tax fair market value of the cash, cash equivalents and Marketable Securities held in the Investment Portfolio (excluding any securities which are subject to a Forward Sale Contract or which secure any margin loan, Forward Sale Contract or other Indebtedness), to (ii) Consolidated EBITDA for the period of four consecutive fiscal quarters ending as of such day. In no event, however, shall the amount determined under clause (i) be less than zero.

"Consolidated EBITDA" means, for any period for the Consolidated Group, the sum of (i) net income (or loss) determined on a consolidated basis in accordance with GAAP, plus (ii) to the extent deducted in determining consolidated net income, (A) all interest expense (net of interest income), including the amortization of debt discount and premium, and the interest component under Capital Leases, (B) income taxes, (C) depreciation and amortization, (D) non-recurring, non-cash charges and adjustments, in either case related to impairment of long-lived assets, and (E) other non-cash extraordinary items, in each case determined on a consolidated basis in accordance with GAAP, minus (iii) to the extent included in determining consolidated net income, non-recurring gains (including gains on the sale of Marketable Securities), in each case determined on a consolidated basis in accordance with GAAP; provided, however, that Consolidated EBITDA shall exclude any EBITDA attributable to or arising from the oil and gas exploration and production business of the Consolidated Group.

"Consolidated Group" means the Parent and its Subsidiaries.

"Consolidated Tangible Net Worth" means, as of any date, consolidated shareholders' equity or net worth of the Consolidated Group, as determined in accordance with GAAP, minus intangible assets of the Consolidated Group, as determined in accordance with GAAP, including (i) deferred charges, (ii) the amount of any write-up in the book value of any assets reflected on

any balance sheet resulting from revaluation thereof or any write-up in excess of the cost of such assets acquired (excluding Marketable Securities), and (iii) the aggregate of all amounts reflected on the assets side of any such balance sheet for franchises, licenses, permits, patents, patent applications, copyrights, trademarks, trade names, goodwill, experimental or organizational expenses and other like intangibles.

"Consolidated Total Capitalization" means, as of any date, the sum of Consolidated Total Funded Debt and Consolidated Tangible Net Worth, each calculated at such time.

"Consolidated Total Funded Debt" means, as of any date, the aggregate Funded Debt of the Consolidated Group determined on a consolidated basis in accordance with GAAP.

"Contingent Debt" means, with respect to any Person, without duplication, any contingent liabilities, obligations or indebtedness of such Person (other than endorsements in the ordinary course of business of negotiable instruments for deposit or collection), including (i) any obligations or similar undertakings to guarantee any Indebtedness of any other Person in any manner, whether direct or indirect, and including any obligation to purchase any such Indebtedness or any Property constituting security therefor, to advance or provide funds or other support for the payment or purchase of any such Indebtedness or to maintain working capital, solvency or other balance sheet condition of such other Person (including keep well agreements, maintenance agreements, comfort letters or similar agreements or arrangements) for the benefit of any holder of Indebtedness of such other Person, to lease or purchase Property, securities or services primarily for the purpose of assuring the holder of such Indebtedness, or otherwise to assure or hold harmless the holder of such Indebtedness against loss in respect thereof, (ii) obligations to indemnify other Persons against liability or loss, to the extent not arising in the ordinary course of business, and (iii) warranty obligations and other contractually assumed obligations, to the extent not arising in the ordinary course of business.

"Contractual Obligation" means, as to any Person, any provision of any security issued or guaranteed by such Person or of any material agreement, instrument or undertaking to which such Person is a party or by which it or any of its Property is bound.

"Credit Documents" means a collective reference to this Agreement, the Notes, the Subsidiary Guaranty, each Subsidiary Guaranty Joinder Agreement, the LOC Documents, the Administrative Agent Fee Letter, the Fee Letter, and all other related agreements and documents issued or delivered hereunder or thereunder or pursuant hereto or thereto (in each case as the same may be amended, modified, restated, supplemented, extended, renewed or replaced from time to time), and "Credit Document" means any one of them.

"Credit Parties" means, collectively, the Borrower, the Parent and the Material Subsidiaries, and "Credit Party" means any one of them.

"Default" means any event, act or circumstance which, with the giving of notice or the lapse of time, or both, would constitute an Event of Default.

"EBITDA" means, for any period and for any member of the Consolidated Group, the sum of (i) its net income (or loss) determined in accordance with GAAP, plus (ii) to the extent deducted in determining such net income, (A) interest expense (net of interest income), including the amortization of debt discount and premium, and the interest component under Capital Leases, in each case determined in accordance with GAAP, (B) income taxes, (C) depreciation and amortization, (D) non-recurring, non-cash charges and adjustments, in either case related to impairment of long-lived assets, and (E) other non-cash extraordinary items, minus (iii) to the extent included in determining net income, non-recurring gains (including gains on the sale of Marketable Securities), in each case determined in accordance with GAAP.

"Eligible Assignee" means (i) a Lender, (ii) an Affiliate of a Lender or any fund that makes purchases, holds or invests in bank loans or similar extensions of credit and is managed or advised by a Lender (or an Affiliate of a Lender) or an investment advisor (or an Affiliate of an investment advisor) to a Lender, or (iii) any other Person approved by the Administrative Agent and the Borrower (such approval by the Administrative Agent or the Borrower not to be unreasonably withheld or delayed and such approval to be deemed given by the Borrower if no objection is received by the assigning Lender and the Administrative Agent from the Borrower within five Business Days after notice of such proposed assignment has been provided by the assigning Lender to the Borrower), unless an Event of Default has occurred and is continuing at the time any assignment is effected in accordance with Section 12.3, in which case no approval from the Borrower or the Administrative Agent shall be required; provided, however, that neither the Borrower nor an Affiliate of the Borrower shall qualify as an Eligible Assignee.

"Environmental Law" means any law, statute, rule, regulation, code or ordinance of any Governmental Authority and any order, decree, judgment, injunction, notice or binding agreement issued, promulgated or entered into by any Governmental Authority, relating in any way to (i) the environment, (ii) the preservation or reclamation of natural resources, (iii) emissions, discharges, releases or threatened releases of Hazardous Materials into the environment, including ambient air, surface water, ground water or land, (iii) the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Materials, (iv) the health effects of Hazardous Materials, or (v) health, industrial hygiene, environmental conditions or safety matters.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and any successor statute thereto, as interpreted by the rules and regulations thereunder, all as the same may be in effect from time to time.

"ERISA Affiliate" means an entity which, at the relevant time, is under common control with any member of the Consolidated Group within the meaning of

Section 4001(a)(14) of ERISA, or is a member of a group which includes any member of the Consolidated Group and which is treated as a single employer under

Section 414(b) or (c) of the IRC.

"ERISA Event" means (i) with respect to any Single Employer Plan or Multiple Employer Plan, the occurrence of a Reportable Event or the substantial cessation of operations (within the meaning of Section 4062(e) of ERISA), (ii) the withdrawal by any member of the Consolidated

Group or any ERISA Affiliate from a Multiple Employer Plan during a plan year in which it was a substantial employer (as such term is defined in Section 4001(a)(2) of ERISA), or the termination of a Multiple Employer Plan, (iii) the distribution of a notice of intent to terminate or the actual termination of a Plan pursuant to Section 4041(a)(2) or 4041A of ERISA, (iv) the institution of proceedings to terminate or the actual termination of a Plan by the PBGC under

Section 4042 of ERISA, (v) any event or condition which could reasonably constitute grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan, (vi) the complete or partial withdrawal of any member of the Consolidated Group or any ERISA Affiliate from a Multiemployer Plan, (vii) the conditions for imposition of a lien under Section 302(f) of ERISA exist with respect to any Single Employer Plan or Multiple Employer Plan, or (viii) the adoption of an amendment to any Plan requiring the provision of security to such Plan pursuant to Section 307 of ERISA.

"Eurodollar Reserve Requirement" means, at any time, the maximum rate at which reserves (including any marginal, special, supplemental, or emergency reserves) are required to be maintained under regulations issued from time to time by the Board of Governors of the Federal Reserve System (or any successor) by member banks of the Federal Reserve System against "Eurocurrency liabilities" (as such term is used in Regulation D). Without limiting the effect of the foregoing, the Eurodollar Reserve Requirement shall reflect any other reserves required to be maintained by such member banks with respect to (i) any category of liabilities which includes deposits by reference to which the Adjusted LIBOR Rate is to be determined, or (ii) any category of extensions of credit or other assets which include LIBOR Rate Loans. The Adjusted LIBOR Rate shall be adjusted automatically on and as of the effective date of any change in the Eurodollar Reserve Requirement.

"Event of Default" has the meaning assigned to such term in Section 9.1.

"Executive Officer" of any Person means the chief executive officer, chief operating officer, president, vice president, chief financial officer or treasurer of such Person.

"Extension of Credit" means, as to any Lender, the making of, or participation in, a Loan by such Lender (including continuations and conversions thereof other than a conversion of a LIBOR Rate Loan into a Prime Rate Loan) or the issuance or extension of, or participation in, a Letter of Credit.

"Facility" means the senior credit facility established by the Lenders under Article II.

"Facility Maturity Date" means June 30, 2005, or such later date to which the Facility Maturity Date may be extended from time to time pursuant to Section 2.5(c).

"Federal Funds Rate" means, for any day, the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (i) if such day is not a Business Day, the

Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (ii) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate charged to the Administrative Agent (in its individual capacity) on such day on such transactions as determined by the Administrative Agent.

"Fee Letter" means that certain letter agreement, dated as of the date hereof, between the Administrative Agent (on behalf of the Lenders) and the Borrower, as amended, modified, restated or supplemented from time to time.

"Fees" means all fees payable pursuant to Section 3.2.

"Forward Sale Contract" means a prepaid forward sale agreement in which the Borrower receives an up-front payment in exchange for a commitment to deliver securities in the future, with the number of shares to be delivered varying with the share price at maturity.

"Funded Debt" means, with respect to any Person, without duplication,

(i) all obligations of such Person for borrowed money, (ii) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, or upon which interest payments are customarily made, (iii) all purchase money Indebtedness (including for purposes hereof, indebtedness and obligations in respect of conditional sale or title retention arrangements and obligations in respect of the deferred purchase price of property or services) of such Person, including the principal portion of all obligations of such Person under Capital Leases, (iv) all Contingent Debt of such Person with respect to Funded Debt of another Person, (v) the maximum available amount of all standby letters of credit or acceptances issued or created for the account of such Person, (vi) all Funded Debt of another Person secured by a Lien on any Property of such Person, whether or not such Funded Debt has been assumed, (vii) the principal balance outstanding under Synthetic Leases, and (viii) the Funded Debt of any partnership or joint venture in which such Person is a general partner or joint venturer, but only to the extent to which there is recourse to such Person for the payment of such Funded Debt.

"GAAP" means generally accepted accounting principles in the United States applied on a consistent basis and subject to the terms of Section 1.3

"Governmental Authority" means any federal, state, local or foreign court or governmental agency, authority, instrumentality or regulatory body.

"Guarantors" means the Parent and each of the current or future Material Subsidiaries.

"Hazardous Materials" means all explosive or radioactive substances or wastes and all hazardous or toxic substances, wastes or other pollutants, including petroleum or petroleum distillates, asbestos, or asbestos-containing materials, polychorinated biphenyls, radon gas, infectious or medical waste and all other substances or wastes of any nature regulated pursuant to any Environmental Law as hazardous, toxic, a pollutant or a contaminant.

"Hedging Agreement" means any interest rate protection agreement, including an interest rate swap, cap or collar agreement or similar arrangement, foreign currency exchange rate protection agreement, commodity price protection or other interest rate, currency exchange rate or commodity price hedging agreement.

"IRC" means the Internal Revenue Code of 1986, as amended, and any successor statute thereto, as interpreted by the rules and regulations issued thereunder, in each case as in effect from time to time.

"Indebtedness" means, with respect to any Person, without duplication,

- (i) all obligations of such Person for borrowed money, (ii) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, or upon which interest payments are customarily made, (iii) all obligations of such Person under conditional sale or other title retention agreements relating to any Properties purchased by such Person (other than customary reservations or retentions of title under agreements with suppliers entered into in the ordinary course of business), (iv) all obligations of such Person issued or assumed as the deferred purchase price of Property or services purchased by such Person (other than trade debt incurred in the ordinary course of business and due within six months of the incurrence thereof) which would appear as liabilities on a balance sheet of such Person, (v) all obligations of such Person under take-or-pay or similar arrangements or under commodities agreements, (vi) all Indebtedness of others secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien on, or payable out of the proceeds of production from, Property owned or acquired by such Person, whether or not the obligations secured thereby have been assumed, (vii) all Contingent Debt of such Person with respect to Indebtedness of another Person, (viii) the principal portion of all obligations of such Person under Capital Leases, (ix) all obligations of such Person under Hedging Agreements,
- (x) the maximum amount of all standby letters of credit issued or bankers' acceptances facilities created for the account of such Person and, without duplication, all drafts drawn thereunder (to the extent unreimbursed), (xi) all preferred Capital Stock issued by such Person and which by the terms thereof could be (at the request of the holders thereof or otherwise) subject to mandatory sinking fund payments, repurchase, redemption or other acceleration any time during the period ending one year after the term of the Agreement,
- (xii) the principal portion of all obligations of such Person under Synthetic Leases, and (xiii) the Indebtedness of any partnership or unincorporated joint venture in which such Person is a general partner or a joint venturer, but only to the extent to which there is recourse to such Person for the payment of such Indebtedness.

"Interest Payment Date" means (i) as to any Prime Rate Loan, the last day of each March, June, September and December, beginning September 30, 2002, and the Facility Maturity Date, as applicable, and (ii) as to any LIBOR Rate Loan, the last day of each Interest Period for such Loan, the date of repayment of principal of such Loan and the Facility Maturity Date, as applicable.

"Interest Period" means, as to any LIBOR Rate Loan, a period of 30, 60 or 90 days' duration, as the Borrower may elect, commencing in each case on the date of the borrowing (including conversions, extensions and renewals); provided, however, (A) if any Interest Period

would end on a day which is not a Business Day, such Interest Period shall be extended to the next succeeding Business Day (except that in the case of LIBOR Rate Loans where the next succeeding Business Day falls in the next succeeding calendar month, then on the next preceding Business Day), (B) no Interest Period shall extend beyond the Facility Maturity Date, and (C) in the case of LIBOR Rate Loans, where an Interest Period begins on a day for which there is no numerically corresponding day in the calendar month in which the Interest Period is to end, such Interest Period shall end on the last day of such calendar month.

"Investment Portfolio" means the Marketable Securities and cash or cash equivalents maintained by the Credit Parties.

"Issuing Lender" means Bank of Oklahoma.

"Issuing Lender Fees" has the meaning assigned to such term in Section 3.2(b)(ii).

"Lenders" means each of the Persons identified as a "Lender" on the signature pages of this Agreement, and their respective successors and permitted assigns. Unless the context otherwise expressly requires, the term includes the Issuing Lender and the Swingline Lender.

"Letter of Credit" means any standby letter of credit issued by the Issuing Lender for the account of the Borrower in accordance with the terms of Section 2.1(b).

"Letter of Credit Fee" has the meaning assigned to such term in Section 3.5(b)(i).

"LIBOR Rate" means, with respect to any LIBOR Rate Loan for any Interest Period therefor, the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) equal to the "London Interbank Offered Rate" quoted as the British Bankers' Association Interest Settlement Rate in U.S. Dollars at approximately 11:00 a.m. (London time) on the Business Day prior to the first day of such Interest Period for a term comparable to such Interest Period, as shown on the LIBOR01 page of the Reuters Information Service; provided, however, if more than one rate is specified on such LIBOR01 Page, the applicable rate shall be the arithmetic mean of all such rates (rounded upwards, if necessary, to the nearest 1/100 of 1%). If the Reuters Information Service shall cease to publish such rate, the Administrative Agent shall determine such rate as the average of quotations for three (3) major New York money center banks of whom the Administrative Agent shall inquire as the "London Interbank Offered Rate" for deposits in U.S. Dollars at approximately 11:00 a.m. (London time) on the Business Day prior to the first day of such Interest Period for a term comparable to such Interest Period.

"LIBOR Rate Loan" means any Loan that bears interest at a rate based upon the LIBOR Rate.

"License" means any license, permit, certificate, consent, franchise, concession, authorization, approval, filing, registration, notification or other grant of authority from or with any Governmental Authority or other Person obtained or required to be obtained from any Gover-

nmental Authority in connection with the management or operation of the business, or the ownership, lease, license or use of any Property, of any member of the Consolidated Group.

"Lien" means any mortgage, pledge, hypothecation, assignment, deposit arrangement, security interest, encumbrance, lien (statutory or otherwise), preference, priority or charge of any kind (including any agreement to give any of the foregoing, any conditional sale or other title retention agreement, any financing or similar statement or notice filed under the UCC or other similar recording or notice statute, and any lease in the nature thereof).

"Loan" or "Loans" means the Revolving Loans, the Swingline Loans or the Term Loans, and the Prime Rate Loans and LIBOR Rate Loans comprising such Loans.

"LOC Commitment" means the commitment of the Issuing Lender to issue Letters of Credit in an aggregate face amount at any time outstanding (together with the amounts of any unreimbursed drawings thereon) of up to the LOC Committed Amount.

"LOC Committed Amount" means the sum of \$25,000,000.

"LOC Documents" means, with respect to any Letter of Credit, such Letter of Credit, any amendments thereto, any documents delivered in connection therewith, any application therefor, and any agreements, instruments, guarantees or other documents (whether general in application or applicable only to such Letter of Credit) governing or providing for the rights and obligations of the parties concerned or at risk.

"LOC Obligations" means, at any time, the sum of (i) the maximum amount which is, or at any time thereafter may become, available to be drawn under Letters of Credit then outstanding, assuming compliance with all requirements for drawings referred to in such Letters of Credit plus (ii) the aggregate amount of all drawings under Letters of Credit honored by the Issuing Lender but not theretofore reimbursed.

"Marketable Securities" means readily marketable publicly-traded securities, including any stock or other equity security publicly-traded on the New York Stock Exchange, the American Stock Exchange or the National Association of Securities Dealers Automated Quotation System (NASDAQ) and, if approved by the Administrative Agent, any other stock traded on a recognized over-the-counter market.

"Material Adverse Effect" means a material adverse effect on or material impairment of (i) the validity or enforceability of any Credit Document or the rights, benefits or remedies of the Administrative Agent or the Lenders under any Credit Document, (ii) the condition (financial or otherwise), operations, business, assets, liabilities or prospects of the Consolidated Group taken as a whole, or (iii) the ability of the Credit Parties to perform or fulfill their obligations under the Credit Documents.

"Material Subsidiary" means each Subsidiary of the Borrower listed on Schedule 1.1 and each other Subsidiary (whether now existing or hereafter acquired or formed) of the Parent

whose net assets comprise (or on a pro forma basis would comprise) more than 5% of the Consolidated Tangible Net Worth of the Consolidated Group or whose EBITDA for either of the two most recent fiscal years account (or on a pro forma basis would account) for more than 5% of Consolidated EBITDA; provided, however, that

(i) Helmerich & Payne de Venezuela, C.A. will not be considered a Material Subsidiary unless and until such time as its net assets comprise more than 10% of the Consolidated Tangible Net Worth of the Consolidated Group or its EBITDA for either of the two most recent fiscal years account for more than 10% of Consolidated EBITDA, and (ii) no Subsidiary now existing or hereafter formed in connection with the Spinoff/Merger (including Cimarex) or any other disposition permitted by Section 8.5(i) shall constitute a Material Subsidiary for purposes of this Agreement so long as such Subsidiary has no material Properties other than those assets associated with the oil and gas exploration and production business of the Consolidated Group.

"Medium Term Notes" means the \$200,000,000 aggregate principal amount of unsecured medium term Senior Notes to be issued and sold by the Borrower substantially on the terms and conditions set forth in the Note Purchase Agreement (draft dated June 26, 2002) relating thereto, a copy of which has been provided to the Administrative Agent (and furnished by the Administrative Agent to the Lenders).

"Multiemployer Plan" means any Plan that is a "multiemployer plan" as defined in Section 3(37) or 4001(a)(3) of ERISA.

"Multiple Employer Plan" means a Plan (other than a Multiemployer Plan) which any member of the Consolidated Group or any ERISA Affiliate and at least one employer other than the members of the Consolidated Group or any ERISA Affiliate are contributing sponsors.

"Non-Use Fee" has the meaning assigned to such term in Section 3.2(a).

"Notes" means the promissory notes of the Borrower in favor of each of the Lenders (or nominees thereof) evidencing the Loans, each in substantially the form attached as Exhibit "A" (with appropriate insertions) or, in the case of the Swingline Loans, Exhibit "A-1," as such promissory notes may be amended, modified, supplemented, extended, renewed or replaced from time to time, and "Note" means any of the Notes.

"Notice of Borrowing" means a written request for a borrowing, in substantially the form of Exhibit "C."

"Notice of Continuation/Conversion" means a written notice of continuation or conversion, in substantially the form of Exhibit "E."

"Notice of Request for Letter of Credit" means a written request for the issuance or extension of a Letter of Credit, in substantially the form of Exhibit "D."

"Obligations" means, collectively, the Revolving Loans, the Swingline Loans, the LOC Obligations, and the Term Loans, including principal, interest, fees, indemnities and other

amounts payable under the Credit Documents, and whether or not allowed as a claim in any bankruptcy proceeding.

"Operating Lease" means any lease which is not a Capital Lease.

"Other Taxes" has the meaning assigned to such term in Section 4.6.

"Participation Interest" means the purchase by a Lender of a participation interest in LOC Obligations as provided in Section 2.6(a), in Swingline Loans as provided in Section 2.7 and in Loans as provided in Section 3.5.

"PBGC" means the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA and any successor thereof.

"Percentage" means, with respect to each Lender and subject to any adjustments provided for in this Agreement, (i) until and including the Revolving Commitment Termination Date, such Lender's Revolving Commitment Percentage, and (ii) at any time after the Revolving Commitment Termination Date, the percentage obtained by dividing such Lender's Term Loan outstanding at such time by the aggregate amount of all Term Loans outstanding at such time.

"Permitted Liens" means Liens permitted under Section 8.2.

"Person" means any individual, partnership, joint venture, firm, corporation, limited liability company, association, trust or other enterprise (whether or not incorporated) or any Governmental Authority.

"Plan" means any employee benefit plan (as defined in Section 3(3) of ERISA) which is covered by ERISA and with respect to which any member of the Consolidated Group or any ERISA Affiliate is (or, if such plan were terminated at such time, would under Section 4069 of ERISA be deemed to be) an "employer" within the meaning of Section 3(5) of ERISA.

"Prime Rate" means the national "prime rate," as reported in the "Money Rates Section" of The Wall Street Journal (Southwest Edition), changing as of the effective date of each change in such rate. In the event such rate is no longer published as provided herein, the Administrative Agent shall designate a comparable index rate as the "Prime Rate" and notify the Borrower and the Lenders of such designation.

"Prime Rate Loan" means any Loan bearing interest at a rate determined by reference to the Prime Rate.

"Property" means any kind of property or asset, whether real, personal or mixed and whether tangible or intangible, including any interest therein.

"Rate Determination Date" means the date that is five Business Days after the date by which the Required Financial Information is required to be delivered in accordance with the provisions of Section 7.1(a) or Section 7.1(b), as applicable.

"Regulation D, O, T, U, or X" means Regulation D, O T, U or X, respectively, of the Board of Governors of the Federal Reserve System as from time to time in effect and any successor to all or a portion thereof.

"Reportable Event" means any of the events set forth in Section 4043(c) of ERISA, other than those events as to which the notice requirement has been waived by regulation.

"Required Financial Information" means the annual and quarterly Compliance Certificates and related financial statements and information required by the provisions of Sections 7.1(a), (b) and (c).

"Required Lenders" means, at any time, Lenders having more than 66.6% of the aggregate Commitments or, if the Commitments have been terminated, Lenders having more than 66.6% of the aggregate principal amount of the Obligations outstanding (taking into account in each case Participation Interests or obligation to participate therein); provided that, in making a determination of Required Lenders, there shall be excluded the Commitments of, and outstanding principal amount of Obligations (taking into account Participation Interests therein) owing to, any Lender that, at the time in question, (i) has failed to make a Loan or purchase a Participation Interest required pursuant to the terms of this Agreement within one Business Day of when due, (ii) other than as set forth in subdivision (i) above, has failed to pay to the Administrative Agent or any Lender an amount owed by such Lender pursuant to the terms of this Agreement within one Business Day of when due, unless such amount is subject to a good faith dispute, or (iii) has been deemed insolvent or has become subject to a bankruptcy or insolvency proceeding or with respect to which (or with respect to any of the assets of which) a receiver, trustee or similar official has been appointed.

"Requirement of Law" means, as to any Person, any requirement or provision of the Charter Documents of such Person, or of any law, statute, rule, regulation, code or ordinance, or of any order, decree, judgment, injunction or other determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or to which any of its material Properties is subject.

"Revolving Commitment" means, with respect to each Lender, the commitment of such Lender to make Revolving Loans in an aggregate principal amount at any time outstanding of up to such Lender's Revolving Committed Amount, as such amount may be reduced from time to time in accordance with the provisions hereof.

"Revolving Commitment Percentage" means, for each Lender, a fraction (expressed as a percentage), the numerator of which is the Revolving Committed Amount of such Lender at such time and the denominator of which is the Aggregate Revolving Committed Amount at such time.

The initial Revolving Commitment Percentages are set out on the signature pages to this Agreement.

"Revolving Commitment Termination Date" means the date which is 364 days following the Closing Date, or such later date to which the Revolving Commitment Termination Date may be extended from time to time pursuant to Section 2.5(c).

"Revolving Committed Amount" means the amount of each Lender's Revolving Commitment as set out on the signature pages to this Agreement.

"Revolving Loans" has the meaning assigned to such term in Section 2.1(a).

"Revolving Obligations" means, collectively, the Revolving Loans, the Swingline Loans and the LOC Obligations.

"Sale/Leaseback Transaction" means any arrangement pursuant to which any member of the Consolidated Group, directly or indirectly, becomes liable as lessee, guarantor or other surety with respect to any lease, whether an Operating Lease or a Capital Lease, of any Property (i) which such member of the Consolidated Group has sold or transferred (or is to sell or transfer) to, or arranged the purchase by, a Person which is not a member of the Consolidated Group or (ii) which such member of the Consolidated Group intends to use for substantially the same purpose as any other Property which has been sold or transferred (or is to be sold or transferred) by such member of the Consolidated Group to another Person which is not a member of the Consolidated Group in connection with such lease.

"SEC" means the Securities and Exchange Commission, or any agency which succeeds to its functions.

"Securities Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Single Employer Plan" means any Plan which is covered by Title IV of ERISA, but which is not a Multiemployer Plan or a Multiple Employer Plan.

"Spinoff/Merger" means the proposed series of transactions (i) reported by the Parent in its Current Report on Form 8-K dated February 25, 2002, and filed with the SEC pursuant to which (A) the Parent will transfer to Cimarex certain assets primarily related to the oil and gas exploration, production, marketing and sales operations of the Parent, (B) Cimarex will assume certain liabilities of the Parent, and (C) the Parent will distribute to its stockholders approximately 0.53 shares of common stock of Cimarex for each share of common stock of the Parent, and (ii) contemplated by (A) the Agreement and Plan of Merger dated as of February 3, 2002, by and among the Parent, Cimarex, Mountain Acquisition Co. and Key Production Company, Inc., (B) the Distribution Agreement by and between the Parent and Cimarex, and (C) any agreement referenced in the agreements referred to in the foregoing Agreement and Plan of Merger and Distribution Agreement.

"Subsidiary" means, as to any Person at any time, (i) any corporation more than 50% of whose Capital Stock of any class or classes having by the terms thereof ordinary voting power to elect a majority of the directors of such corporation (irrespective of whether or not at such time, any class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at such time owned by such Person directly or indirectly through Subsidiaries, and (ii) any partnership, association, joint venture or other entity of which such Person directly or indirectly through Subsidiaries owns at such time more than 50% of the Capital Stock; provided that no Plan shall be considered to be a Subsidiary of the Parent or the Borrower.

"Subsidiary Guaranty" means the Guaranty Agreement, substantially in the form attached hereto as Exhibit "B," to be given by the Material Subsidiaries in favor of the Lenders and the Administrative Agent with respect to the obligations of the Borrower under the Credit Documents, as amended or modified from time to time.

"Subsidiary Guaranty Joinder Agreement" means the Joinder Agreement, substantially in the form of Schedule 1 to the Subsidiary Guaranty, to be executed and delivered by each new Material Subsidiary in accordance with the provisions of Section 7.13.

"Swingline Commitment" means the commitment of the Swingline Lender to make Swingline Loans in an aggregate principal amount at any time outstanding up to the Swingline Committed Amount and the commitment of the Lenders to purchase participation interests in the Swingline Loans up to their respective Revolving Commitment Percentage as provided in Section 2.1(c), as such amounts may be reduced from time to time in accordance with the provisions hereof.

"Swingline Committed Amount" means the sum of \$10,000,000.

"Swingline Lender" means Bank of Oklahoma.

"Swingline Loan" means a swingline revolving loan made by the Swingline Lender pursuant to the provisions of Section 2.1(c).

"Synthetic Lease" means any synthetic lease, tax retention operating lease, off-balance sheet loan or similar off-balance sheet financing product where such transaction is considered borrowed money indebtedness for tax purposes but is classified as an Operating Lease under GAAP.

"Taxes" has the meaning assigned to such term in Section 4.6.

"Term Loans" means the Term Loans into which the Revolving Loans outstanding on the Revolving Commitment Termination Date shall be converted, as provided in Section 2.1(d).

"UCC" means the Uniform Commercial Code as in effect from time to time in the State of Oklahoma or other applicable jurisdiction.

"Voting Stock" means, with respect to any Person, Capital Stock issued by such Person the holders of which are ordinarily, in the absence of contingencies, entitled to vote for the election of

directors (or persons performing similar functions) of such Person, even though the right so to vote has been suspended by the happening of such a contingency.

- 1.2 Construction and Interpretation. For purposes of this Agreement, the following rules of construction shall apply, unless elsewhere specifically indicated to the contrary: (i) all terms defined herein in the singular shall include the plural, as the context requires, and vice-versa; (ii) pronouns stated in the neuter gender shall include the masculine, the feminine and the neuter genders; (iii) for purposes of the computation of periods of time hereunder, the word "from" means "from and including" and the words "to" and "until" each mean "to but excluding"; (iv) the term "or" is not exclusive; (iv) the term "including" (or any form thereof) shall not be limiting or exclusive;
- (vi) references to article and sections are references to the respective articles and sections of this Agreement, and references to exhibits and schedules are references to the exhibits and schedules attached to this Agreement; (vii) all references to this Agreement or any of the other Credit Documents shall include any and all modifications, amendments or supplements hereto or thereto and any and all renewals and extensions hereof or thereof; and
- (viii) references to specific provisions or sections of statutes and other laws (such as the Bankruptcy Code, ERISA and the IRC) shall be construed also to refer to any successor provisions or sections.
- 1.3 Accounting Terms. Except as otherwise expressly provided herein, all accounting terms used herein shall be interpreted, and all financial statements and certificates and reports as to financial matters required to be delivered to the Lenders hereunder shall be prepared, in accordance with GAAP applied on a consistent basis. All calculations made for the purposes of determining compliance with this Agreement shall (except as otherwise expressly provided herein) be made by application of GAAP applied on a basis consistent with the most recent annual or quarterly financial statements delivered pursuant to Section 7.1.

ARTICLE II TERMS OF THE CREDIT FACILITY

2.1 Commitments.

(a) Revolving Commitments. During the Commitment Period, and subject to the terms and conditions of this Agreement, each Lender severally agrees to make revolving loans (the "Revolving Loans") to the Borrower from time to time in amounts not to exceed at any time such Lender's Revolving Commitment, minus its Revolving Commitment Percentage of the sum of (x) all LOC Obligations then outstanding and (y) the aggregate principal amount of all Swingline Loans then outstanding. At no time shall the aggregate principal amount of Revolving Loans outstanding hereunder, together with all other Revolving Obligations then outstanding, exceed the Aggregate Revolving Credit Commitment, and with regard to each Lender individually, such Lender's Revolving Commitment Percentage of the Revolving Loans and other Revolving Obligations at any time outstanding shall not exceed such Lender's Revolving Committed Amount. The Revolving Loans may consist of Prime Rate Loans or LIBOR Rate Loans, or a combination thereof, as the Borrower may from time to time request, and may be repaid

and reborrowed in accordance with the provisions hereof. No Lender shall be responsible for any failure by any other Lender to perform its obligation to make any Revolving Loan hereunder nor shall the Revolving Commitment of any Lender be increased or decreased as a result of any such failure. The Revolving Commitments of the Lenders shall expire on the Revolving Commitment Termination Date.

- (b) Letter of Credit Sublimit Commitment. During the Commitment Period, and subject to the terms and conditions of this Agreement and the LOC Documents, if any, the Issuing Lender shall issue, and the Lenders shall severally participate in, such Letters of Credit as the Borrower may request, in form acceptable to the Issuing Lender, for the purposes hereinafter set forth; provided that the aggregate amount of LOC Obligations shall not at any time exceed the LOC Committed Amount. The renewal or extension of any outstanding Letter of Credit shall, for purposes hereof, be treated in all respects the same as the issuance of a new Letter of Credit hereunder; provided that the Issuing Lender may renew or extend outstanding Letters of Credit (but shall not issue new Letters of Credit) after the expiration of the Commitment Period. Letters of Credit issued hereunder (including renewals and extensions) shall have an expiry date which in any event does not extend beyond the date which is five Business Days prior to the Facility Maturity Date.
- (c) Swingline Sublimit Commitment. During the Commitment Period, and subject to the terms and conditions of this Agreement, the Swingline Lender agrees to make certain revolving loans (the "Swingline Loans") to the Borrower; provided that the aggregate principal amount of Swingline Loans at any time outstanding shall not exceed the Swingline Committed Amount. All Swingline Loans shall be Prime Rate Loans, and may be repaid and reborrowed in accordance with the provisions hereof.
- (d) Term Loan Commitment. On the Revolving Commitment Termination Date, and subject to the terms and conditions of this Agreement and provided that no Event of Default has occurred and is then continuing, each Lender severally agrees, automatically and without any request from the Borrower, to make a term loan (each a "Term Loan," and collectively, the "Term Loans"), in the principal amount equal to its Revolving Commitment Percentage of the aggregate Revolving Loans outstanding at the close of business on the Revolving Commitment Termination Date. The Term Loans may be comprised of Prime Rate Loans or LIBOR Rate Loans, or a combination thereof, as the Borrower may from time to time request. Amounts repaid on the Term Loans may not be reborrowed. Except as provided in Section 2.6(c), no borrowings may be requested under the Term Loans after the conversion of the Revolving Loans outstanding at the close of business on the Revolving Commitment Termination Date.
- 2.2 Method of Borrowing; Use of Proceeds.
- (a) Notice of Request for Extensions of Credit. The Borrower shall request an Extension of Credit as follows:

- (i) Revolving Loans. In the case of a request for Revolving Loans, by delivering a written Notice of Borrowing (or by giving telephonic notice promptly confirmed by a Notice of Borrowing in writing) to the Administrative Agent not later than 11:00 a.m. (Tulsa, Oklahoma time) on the Business Day prior to the date of the requested borrowing in the case of Prime Rate Loans, and on the third Business Day prior to the date of the requested borrowing in the case of LIBOR Rate Loans. Each Notice of Borrowing shall be irrevocable and shall specify (A) that a Revolving Loan is requested, (B) the date of the requested borrowing (which shall be a Business Day),
- (C) the aggregate principal amount to be borrowed, (D) whether the borrowing is to be comprised of Prime Rate Loans, LIBOR Rate Loans or a combination thereof (it being understood that each Loan extended as a LIBOR Rate Loan shall be in the minimum principal amount of \$5,000,000 or in an integral multiple of \$500,000 in excess thereof), and (E) if LIBOR Rate Loans are requested, the Interest Period(s) therefor. If the Borrower shall fail to specify the type of loan requested, the Borrower shall be deemed to have requested a Prime Rate Loan, and if the Borrower shall fail to specify the applicable Interest Period in the case of a request for a LIBOR Rate Loan, the Borrower shall be deemed to have requested an Interest Period of 30 days. The Administrative Agent shall give notice to each Lender promptly upon receipt of each Notice of Borrowing pursuant to this Section 2.2(a)(i), the contents thereof and each such Lender's share of any borrowing to be made pursuant thereto.
- (ii) Letters of Credit. In the case of a request for the issuance or extension of a Letter of Credit, by delivering a Notice of Request for Letter of Credit to the Issuing Lender, with a copy to the Administrative Agent, not later than 11:00 a.m. (Tulsa, Oklahoma time) on the third Business Day prior to the date of the requested issuance or extension (or such shorter period as may be agreed by the Issuing Lender). Each such request for issuance or extension of a Letter of Credit shall be irrevocable and shall specify, among other things, (A) that a Letter of Credit is requested, (B) the date of the requested issuance or extension, (C) the type, amount, expiry date and terms on which the Letter of Credit is to be issued or extended, and (D) the beneficiary.
- (iii) Swingline Loans. In the case of a request for a Swingline Loan, by delivering a written Notice of Borrowing (or by giving telephonic notice promptly confirmed by a Notice of Borrowing in writing) to the Swingline Lender not later than 11:00 a.m. (Tulsa, Oklahoma time) on the Business Day of the requested borrowing. Each such request for borrowing shall be irrevocable and shall specify (A) that a Swingline Loan is requested, (B) the date of the requested borrowing (which shall be a Business Day), (C) the aggregate principal amount to be borrowed, and (D) the requested maturity. Each Swingline Loan shall have a maturity date as the Borrower may request and the Swingline Lender may agree, but in no event shall the term of any Swingline Loan be longer than 10 days from the date of borrowing.

- (b) Minimum Amounts. Each request for a Revolving Loan shall be in the minimum principal amount of \$5,000,000 (or the remaining Aggregate Revolving Committed Amount, if less than \$5,000,000, but only if the request is for a Prime Rate Loan), or in an integral multiple of \$500,000 in excess thereof. Each request for a Swingline Loan shall be in the minimum principal amount of \$500,000 (or the remaining Aggregate Revolving Committed Amount, if less than \$500,000), or in an integral multiple of \$250,000 in excess thereof.
- (c) Use of Proceeds. Proceeds of the Revolving Loans and Swingline Loans shall be used by the Borrower for working capital and general corporate purposes of the Consolidated Group, including capital expenditures. The proceeds of the initial Revolving Loan shall be used to repay the existing revolving loans, if any, made by Bank of Oklahoma to the Parent.
- (d) Maximum Number of LIBOR Rate Loans. Not more than 10 LIBOR Rate Loans (whether as original borrowings or as continuations or conversions pursuant to Section 2.3) may be outstanding at any time. For purposes of this Section 2.2(d), a LIBOR Rate Loan may consist of an original borrowing and a continuation and/or conversion pursuant to
- Section 2.3 and may be considered a single LIBOR Rate Loan, provided that the borrowing and the continuation/conversion are made on the same day and have the same Interest Period.
- 2.3 Continuation and Conversion. The Borrower shall have the option, on any Business Day, to extend existing Loans into a subsequent permissible Interest Period or to convert Loans into Loans of another interest rate type; provided, however, that (i) except as provided in Section 4.3, LIBOR Rate Loans may be converted into Prime Rate Loans or extended as LIBOR Rate Loans for new Interest Periods only on the last day of the Interest Period applicable thereto,
- (ii) without the consent of the Required Lenders, LIBOR Rate Loans may be extended, and Prime Rate Loans may be converted into LIBOR Rate Loans, only if the conditions precedent set forth in Section 5.3 are satisfied on the date of continuation or conversion, and (iii) Loans converted into LIBOR Rate Loans shall be in the minimum principal amount of \$5,000,000 or in integral multiple of \$500,000 in excess thereof. Each such continuation or conversion shall be effected by the Borrower by giving a Notice of Continuation/Conversion (or telephonic notice promptly confirmed in writing) to Administrative Agent prior to 11:00 a.m. (Tulsa, Oklahoma time) on the Business Day of, in the case of the conversion of a LIBOR Rate Loan into a Prime Rate Loan, and on the third Business Day prior to, in the case of the continuation of a LIBOR Rate Loan as, or conversion of a Prime Rate Loan into, a LIBOR Rate Loan, the date of the proposed continuation or conversion, specifying the date of the proposed continuation or conversion, the Loans to be so continued or converted, the types of Loans into which such Loans are to be converted and, if appropriate, the applicable Interest Periods with respect thereto. If the Borrower shall fail to specify the applicable Interest Period in the case of a request for continuation of a LIBOR Rate Loan as, or conversion of a Prime Rate Loan into, a LIBOR Rate Loan, the Borrower shall be deemed to have requested an Interest Period of 30 days. Each request for continuation or conversion shall be irrevocable and shall constitute a representation and warranty by the Borrower of the matters specified in

subsections (a), (b), (c) and (d) of Section 5.3. In the event the Borrower fails to request continuation or conversion of any LIBOR Rate Loan in accordance with this Section 2.3, or any such conversion or continuation is not permitted or required by this Section, then such LIBOR Rate Loan shall be automatically converted into a Prime Rate Loan at the end of the Interest Period applicable thereto. The Administrative Agent shall give each Lender notice as promptly as practicable of any such proposed extension or conversion affecting any Loan.

2.4 Notes. The Loans from time to time outstanding shall be evidenced by the Notes, which shall be made, executed and delivered by the Borrower payable to the order of the respective Lenders on or before the Closing Date.

- 2.5 Reduction, Termination and Extension of Revolving Commitments.
- (a) Voluntary Reduction of Revolving Commitments. The Revolving Commitments may be terminated or permanently reduced in whole or in part upon three (3) Business Days' prior written notice to the Administrative Agent, provided that (i) after giving effect to any voluntary reduction, the aggregate amount of Revolving Obligations shall not exceed the Aggregate Revolving Committed Amount, as reduced, and (ii) partial reductions shall be in a minimum principal amount of \$5,000,000 and in integral multiples of \$1,000,000 in excess thereof.
- (b) Termination of Revolving Commitments. The Revolving Commitments hereunder shall terminate on the Revolving Commitment Termination Date.
- (c) Extension of Revolving Commitment Termination Date. By submitting a request for extension to the Administrative Agent (an "Extension Request") at least 60 days prior to the then-current Revolving Commitment Termination Date, the Borrower may request that the Lenders extend the Facility (and correspondingly, the Revolving Commitment Termination Date and the Facility Maturity Date) for an additional period of 364 days beyond the then-current Revolving Commitment Termination Date and Facility Maturity Date. The Lenders agree to consider each Extension Request, but no Extension Request shall be approved unless it is agreed to by all of the Lenders. The Administrative Agent shall give the Borrower notice of the Lenders' decision with respect to the Borrower's Extension Request not less than 15 days prior to the then-current Revolving Commitment Termination Date; provided, however, that if the Administrative Agent fails to notify the Borrower of the Lenders' decision, the Extension Request will be deemed not to have been approved. In the event that the Lenders approve the Extension Request, (i) the Revolving Commitment Termination Date and the Facility Maturity Date shall each be extended for an additional period of 364 days, and (ii) the terms and provisions of this Agreement shall continue in full force and effect, except as may otherwise be agreed in writing by the Borrower, the Lenders and the Administrative Agent. Notwithstanding the foregoing, the Lenders shall not be obligated, and nothing contained in this Agreement shall be construed as obligating the Lenders, to agree or consent to any extension request, and any determination by the Lenders not to consent to any Extension Request shall be final.
- 2.6 Additional Provisions Relating to Letters of Credit.
- (a) Participation. Each Lender, with respect to the outstanding letters of credit identified on Schedule 2.6(a) (the "Existing Letters of Credit"), hereby purchases, without recourse to the Issuing Lender, a risk participation in such Existing Letters of Credit and the obligations arising thereunder, and with respect to Letters of Credit issued on or after the Closing Date, upon issuance of a Letter of Credit, shall be deemed to have purchased, without recourse to the Issuing Lender, a risk participation from the Issuing Lender in such Letter of Credit and the obligations arising thereunder, in each case in an amount equal to its pro rata share of the obligations under such Letter of Credit (based on the res-

pective Percentages of the Lenders) and shall absolutely, unconditionally and irrevocably assume, as primary obligor and not as surety, and be obligated to pay to the Issuing Lender therefor and discharge when due, its pro rata share of the obligations arising under such Letter of Credit. Without limiting the scope and nature of each Lender's participation in any Letter of Credit, to the extent that the Issuing Lender has not been reimbursed as required hereunder or under any such Letter of Credit, each Lender shall pay to the Issuing Lender its pro rata share of such unreimbursed drawing in same day funds on the day of notification by the Issuing Lender of an unreimbursed drawing pursuant to the provisions of subsection (c) of this Section

2.6. The obligation of each Lender to so reimburse the Issuing Lender shall be absolute and unconditional and shall not be affected by the occurrence of a Default, an Event of Default or any other occurrence or event. Any such reimbursement shall not relieve or otherwise impair the obligation of the Borrower to reimburse the Issuing Lender under any Letter of Credit, together with interest as hereinafter provided.

(b) Reimbursement. In the event of any drawing under any Letter of Credit, the Issuing Lender will promptly notify the Borrower. Unless the Borrower shall immediately notify the Issuing Lender that the Borrower intends to otherwise reimburse the Issuing Lender for such drawing, the Borrower shall be deemed to have requested that the Lenders make a Revolving Loan (or to increase the amount of their outstanding Term Loans) in the amount of the drawing as provided in subsection (c) hereof on the related Letter of Credit, the proceeds of which will be used to satisfy the related reimbursement obligations. The Borrower promises to reimburse the Issuing Lender on the day of drawing under any Letter of Credit (either with the proceeds of a Loan obtained hereunder or otherwise) in same day funds. If the Borrower shall fail to reimburse the Issuing Lender as provided hereinabove, the unreimbursed amount of such drawing shall bear interest at a per annum rate equal to the Prime Rate plus the sum of (i) the Applicable Margin and (ii) two percent (2%). The Borrower's reimbursement obligations hereunder shall be absolute and unconditional under all circumstances irrespective of any rights of setoff, counterclaim or defense to payment the Borrower may claim or have against the Issuing Lender, the Administrative Agent, the Lenders, the beneficiary of the Letter of Credit drawn upon or any other Person, including any defense based on any failure of the Borrower or any other Credit Party to receive consideration or the legality, validity, regularity or unenforceability of the Letter of Credit. The Issuing Lender will promptly notify the other Lenders of the amount of any unreimbursed drawing and each Lender shall promptly pay to the Administrative Agent for the account of the Issuing Lender in Dollars and in immediately available funds, the amount of such Lender's Percentage of such unreimbursed drawing. Such payment shall be made on the day such notice is received by such Lender from the Issuing Lender if such notice is received at or before 12:00 noon (Tulsa, Oklahoma time); otherwise, such payment shall be made at or before 11:00 a.m. (Tulsa, Oklahoma time) on the Business Day next succeeding the day such notice is received. If such Lender does not pay such amount to the Issuing Lender in full upon such request, such Lender shall, on demand, pay to the Administrative Agent for the account of the Issuing Lender interest on the unpaid amount during the period from the date of such drawing until such Lender pays such amount to the Issuing Lender in full at a

rate per annum equal to, if paid within two Business Days of the date that such Lender is required to make payments of such amount pursuant to the preceding sentence, the Federal Funds Rate and thereafter at a rate equal to the Prime Rate. Each Lender's obligation to make such payment to the Issuing Lender, and the right of the Issuing Lender to receive the same, shall be absolute and unconditional, shall not be affected by any circumstance whatsoever and without regard to the termination of this Agreement or the Commitments hereunder, the existence of a Default or Event of Default or the acceleration of the obligations of the Borrower hereunder and shall be made without any offset, abatement, withholding or reduction whatsoever. Simultaneously with the making of each such payment by a Lender to the Issuing Lender, such Lender shall, automatically and without any further action on the part of the Issuing Lender or such Lender, acquire a Participation Interest in the unreimbursed drawing in an amount equal to such payment (excluding the portion of such payment, if any, constituting interest owing to the Issuing Lender).

(c) Repayment with Revolving Loan or Term Loan Increase. On any day on which the Borrower shall have requested, or been deemed to have requested, a Revolving Loan advance or a Term Loan increase to reimburse a drawing under a Letter of Credit, the Administrative Agent shall give notice to the Lenders that a Revolving Loan or a Term Loan increase has been requested or deemed requested by the Borrower to be made in connection with a drawing under a Letter of Credit, in which case a Revolving Loan advance or Term Loan increase comprised of Prime Rate Loans (or LIBOR Rate Loans to the extent the Borrower has complied with the procedures of Section 2.2(a)(i) with respect thereto) shall be immediately made to the Borrower by all Lenders (notwithstanding any termination of the Commitments pursuant to Section 9.2) pro rata based on the respective Percentages of the Lenders (determined before giving effect to any termination of the Commitments pursuant to Section 9.2) and the proceeds thereof shall be paid directly to the Issuing Lender for application to the respective LOC Obligations. Each Lender hereby irrevocably agrees to make its pro rata share of each such Revolving Loan or Term Loan increase immediately upon any such request or deemed request in the amount, in the manner and on the date specified in the preceding sentence notwithstanding (i) the amount of such borrowing may not comply with the minimum amount for advances of Revolving Loans otherwise required hereunder, (ii) whether any conditions specified in

Section 5.3 are then satisfied, (iii) whether a Default or an Event of Default then exists, (iv) failure for any such request or deemed request for Revolving Loan or Term Loan increase to be made by the time otherwise required hereunder, (v) whether the date of such borrowing is a date on which Revolving Loans are otherwise permitted to be made hereunder or (vi) any termination of the Commitments relating thereto immediately prior to or contemporaneously with such borrowing. In the event that any Revolving Loan or Term Loan increase cannot for any reason be made on the date otherwise required above (including as a result of the commencement of a proceeding under the Bankruptcy Code with respect to the Borrower or any Credit Party), then each such Lender hereby agrees that it shall forthwith purchase (as of the date such borrowing would otherwise have occurred, but adjusted for any payments received from the Borrower on or after such date and prior to such purchase)

from the Issuing Lender such participation in the outstanding LOC Obligations as shall be necessary to cause each such Lender to share in such LOC Obligations ratably (based upon the respective Percentages of the Lenders (determined before giving effect to any termination of the Commitments pursuant to Section 9.2)), provided that in the event such payment is not made on the day of drawing, such Lender shall pay in addition to the Issuing Lender interest on the amount of its unfunded Participation Interest at a rate equal to, if paid within two Business Days of the date of drawing, the Federal Funds Rate, and thereafter at the Prime Rate.

- (d) Terms of Letters of Credit. Each Letter of Credit shall be a standby letter of credit issued for the purpose stated in the applicable Notice of Request for Letter of Credit. Notwithstanding anything to the contrary set forth in this Agreement, including Section 2.2(a)(ii) hereof, a Letter of Credit issued hereunder may contain a statement to the effect that such Letter of Credit is issued for the account of a Credit Party, provided that notwithstanding such statement, the Borrower shall be the actual account party for all purposes of this Agreement for such Letter of Credit and such statement shall not affect the Borrower's reimbursement obligations hereunder with respect to such Letter of Credit. The Issuing Lender may issue the Letters of Credit subject to The Uniform Customs and Practice for Documentary Credits, as published as of the date of issue by the International Chamber of Commerce.
- (e) Indemnification; Nature of Issuing Lender's Duties. In addition to its other obligations under this Section 2.6, the Borrower hereby agrees to protect, indemnify, pay and save the Issuing Lender harmless from and against any and all claims, demands, liabilities, damages, losses, costs, charges and expenses (including reasonable attorneys' fees) that the Issuing Lender may incur or be subject to as a consequence, direct or indirect, of (A) the issuance of any Letter of Credit or (B) the failure of the Issuing Lender to honor a drawing under a Letter of Credit as a result of any act or omission, whether rightful or wrongful, of any Governmental Authority (all such acts or omissions, herein called "Government Acts"). As between the Borrower and the Issuing Lender, the Borrower shall assume all risks of the acts, omissions or misuse of any Letter of Credit by the beneficiary thereof. The Issuing Lender shall not be responsible: (A) for the form, validity, sufficiency, accuracy, genuineness or legal effect of any document submitted by any party in connection with the application for and issuance of any Letter of Credit, even if it should in fact prove to be in any or all respects invalid, insufficient, inaccurate, fraudulent or forged; (B) for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign any Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, that may prove to be invalid or ineffective for any reason; (C) for errors, omissions, interruptions or delays in transmission or delivery of any messages, by mail, cable, telegraph, telex or otherwise, whether or not they be in cipher; (D) for any loss or delay in the transmission or otherwise of any document required in order to make a drawing under a Letter of Credit or of the proceeds thereof; and (E) for any consequences arising from causes beyond the control of the Issuing Lender, including any Government Acts. None of the above shall affect, impair, or prevent t

furtherance and extension and not in limitation of the specific provisions set forth in this subsection (e), any action taken or omitted by the Issuing Lender, under or in connection with any Letter of Credit or the related certificates, if taken or omitted in good faith, shall not put such Issuing Lender under any resulting liability to the Borrower or any other Credit Party. It is the intention of the parties that this Agreement shall be construed and applied to protect and indemnify the Issuing Lender against any and all risks involved in the issuance of the Letters of Credit, all of which risks are hereby assumed by the Borrower (on behalf of itself and each of the other Credit Parties), including any and all Government Acts. The Issuing Lender shall not, in any way, be liable for any failure by the Issuing Lender or anyone else to pay any drawing under any Letter of Credit as a result of any Government Acts or any other cause beyond the control of the Issuing Lender. Nothing in this subsection (e) is intended to limit the reimbursement obligations of the Borrower contained in subsection (b) above. The obligations of the Borrower under this subsection (e) shall survive the termination of this Agreement. No act or omissions of any current or prior beneficiary of a Letter of Credit shall in any way affect or impair the rights of the Issuing Lender to enforce any right, power or benefit under this Agreement. Notwithstanding anything to the contrary contained in this subsection

- (e), the Borrower shall have no obligation to indemnify the Issuing Lender in respect of any liability incurred by the Issuing Lender (A) arising out of the gross negligence or willful misconduct of the Issuing Lender, as determined by a court of competent jurisdiction, or (B) caused by the Issuing Lender's failure to pay under any Letter of Credit after presentation to it of a request strictly complying with the terms and conditions of such Letter of Credit, as determined by a court of competent jurisdiction, unless such payment is prohibited by any law, regulation, court order or decree.
- (f) Responsibility of Issuing Lender. It is expressly understood and agreed that the obligations of the Issuing Lender hereunder to the Lenders are only those expressly set forth in this Agreement and that the Issuing Lender shall be entitled to assume that the conditions precedent set forth in Section 5.3 have been satisfied unless it shall have acquired actual knowledge that any such condition precedent has not been satisfied; provided, however, that nothing set forth in this Section 2.6 shall be deemed to prejudice the right of any Lender to recover from the Issuing Lender any amounts made available by such Lender to the Issuing Lender pursuant to this Section 2.6 in the event that it is determined by a court of competent jurisdiction that the payment with respect to a Letter of Credit constituted gross negligence or willful misconduct on the part of the Issuing Lender.
- (g) Reports. The Issuing Lender will provide to the Administrative Agent for dissemination to the Lenders at least quarterly, and more frequently upon request, a detailed summary report on its Letters of Credit and the activity thereon, including, among other things, the Credit Party for whose account the Letter of Credit is issued, the beneficiary, the face amount, and the expiry date. The Issuing Lender will provide copies of the Letters of Credit to the Administrative Agent and the Lenders promptly upon request.

(h) Conflict with LOC Documents. Solely as among the parties hereto, in the event of any conflict between this Agreement and any LOC Document (including any letter of credit application), this Agreement shall control.

2.7 Additional Provisions Relating to Swingline Loans. The Swingline Lender may, at any time, in its sole discretion, by written notice to the Borrower and the Lenders, demand repayment of its Swingline Loans by way of a Revolving Loan advance, in which case the Borrower shall be deemed to have requested a Revolving Loan advance comprised solely of Prime Rate Loans in the amount of such Swingline Loans; provided, however, that any such demand shall be deemed to have been given one Business Day prior to the Revolving Commitment Termination Date and on the date of the occurrence of any Event of Default described in Section 9.1 and upon acceleration of the indebtedness hereunder and the exercise of remedies in accordance with the provisions of Section 9.2. Each Lender hereby irrevocably agrees to make its Revolving Commitment Percentage of each such Revolving Loan in the amount, in the manner and on the date specified in the preceding sentence notwithstanding (i) the amount of such borrowing may not comply with the minimum amount for advances of Revolving Loans otherwise required hereunder, (ii) whether any conditions specified in Section 5.3 are then satisfied, (iii) whether a Default or an Event of Default then exists, (iv) failure of any such request or deemed request for Revolving Loan to be made by the time otherwise required hereunder, (v) whether the date of such borrowing is a date on which Revolving Loans are otherwise permitted to be made hereunder or (vi) any termination of the Commitments relating thereto immediately prior to or contemporaneously with such borrowing. In the event that any Revolving Loan cannot for any reason be made on the date otherwise required above (including as a result of the commencement of a proceeding under the Bankruptcy Code with respect to the Borrower or any other Credit Party), then each Lender hereby agrees that it shall forthwith purchase (as of the date such borrowing would otherwise have occurred, but adjusted for any payments received from the Borrower on or after such date and prior to such purchase) from the Swingline Lender such Participation Interests in the outstanding Swingline Loans as shall be necessary to cause each such Lender to share in such Swingline Loans ratably based upon its Revolving Commitment Percentage of the Revolving Committed Amount (determined before giving effect to any termination of the Commitments pursuant to Section 9.2), provided that (A) all interest payable on the Swingline Loans shall be for the account of the Swingline Lender until the date as of which the respective Participation Interest is funded and (B) at the time any purchase of Participation Interests pursuant to this sentence is actually made, the purchasing Lender shall be required to pay to the Swingline Lender, to the extent not paid to the Swingline Lender by the Borrower in accordance with the terms of Section 3.3(b), interest on the principal amount of Participation Interests purchased for each day from and including the day upon which such borrowing would otherwise have occurred to but excluding the date of payment for such Participation Interests, at the rate equal to the Federal Funds Rate.

2.8 Pro Rata Treatment. Except to the extent otherwise provided herein:

- (a) Loans. Each Revolving Loan advance, each payment or prepayment of principal of any Loan (other than Swingline Loans) or reimbursement obligations arising from drawings under Letters of Credit, each payment of interest on the Loans (other than Swingline Loans) or reimbursement obligations arising from drawings under Letters of Credit, each payment of Non-Use Fees, each payment of Letter of Credit Fees, each reduction of the Revolving Committed Amount and each conversion or extension of any Loan (other than Swingline Loans) shall be allocated pro rata among the Lenders in accordance with the respective Revolving Commitment Percentages.
- (b) Advances. No Lender shall be responsible for the failure or delay by any other Lender in its obligation to make its ratable share of a borrowing hereunder; provided, however, that the failure of any Lender to fulfill its obligations hereunder shall not relieve any other Lender of its obligations hereunder. Unless the Administrative Agent shall have been notified by any Lender prior to the date of any requested borrowing that such Lender does not intend to make available to the Administrative Agent its ratable share of such borrowing to be made on such date, the Administrative Agent may assume that such Lender has made such amount available to the Administrative Agent on the date of such borrowing, and the Administrative Agent in reliance upon such assumption, may (in its sole discretion but without any obligation to do so) make available to the Borrower a corresponding amount. If such corresponding amount is not in fact made available to the Administrative Agent, the Administrative Agent shall be able to recover such corresponding amount from such Lender. If such Lender does not pay such corresponding amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent will promptly notify the Borrower, and the Borrower shall immediately pay such corresponding amount to the Administrative Agent. The Administrative Agent shall also be entitled to recover from the Lender or the Borrower, as the case may be, interest on such corresponding amount in respect of each day from the date such corresponding amount was made available by the Administrative Agent to the Borrower to the date such corresponding amount is recovered by the Administrative Agent at a per annum rate equal to (i) from the Borrower at the applicable rate for the applicable borrowing pursuant to the Notice of Borrowing and (ii) from a Lender at the Federal Funds Rate.
- 2.9 Records of Loans. The Administrative Agent shall make reasonable efforts to maintain an accurate record of (i) the amount, type and Interest Period of each Loan hereunder, (ii) the amount of any principal or interest due and payable or to become due and payable to each Lender hereunder, and (iii) the amount of any sum received by the Administrative Agent hereunder from or for the account of the Borrower and each Lender's share thereof, and each Lender shall likewise make reasonable efforts to maintain an accurate record of each Loan made by such Lender, including the amounts of principal and interest payable and paid to such Lender from time to time under this Agreement. The entries made by each Lender in its records, if consistent with the entries made by the Administrative Agent in its records, shall be prima facie evidence of the existence and amounts of the obligations of the Borrower therein recorded; provided, however, that the failure of any Lender or the Administrative Agent to maintain any

such records, or any error therein, shall not in any manner affect the obligation of the Borrower to repay the Loans and other obligations owing to such Lender.

ARTICLE III INTEREST, FEES AND REPAYMENT

- 3.1 Interest. Subject to the provisions of Article IV, the Loans hereunder shall bear interest at a per annum rate, payable in arrears on each applicable Interest Payment Date (or at such other times as may be specified herein), as follows:
- (a) Prime Rate Loans. During such periods as the Loans shall be comprised of Prime Rate Loans, the sum of the Prime Rate minus the Applicable Margin.
- (b) LIBOR Rate Loans. During such periods as the Loans shall be comprised of LIBOR Rate Loans, the sum of the Adjusted LIBOR Rate plus the Applicable Margin.
- (c) Default Rate. Upon the occurrence and during the continuation of any Event of Default, the principal of and, to the extent permitted by law, interest on the Loans and any other amounts owing hereunder or under the other Credit Documents shall bear interest, payable on demand, at a per annum rate 2% greater than the rate which would otherwise be applicable (or if no rate is applicable, whether in respect of interest, fees or other amounts, then the Prime Rate plus the then effective Applicable Margin for Prime Rate Loans plus 2%.
- 3.2 Fees.
- (a) Non-Use Fee. In consideration of the Revolving Commitments, hereunder, the Borrower agrees to pay to the Administrative Agent for the benefit of each Lender a non-use fee (the "Non-Use Fee") for the period from the Closing Date to the Revolving Commitment Termination Date calculated at the rate of 0.15% (15 bps) per annum on the average daily unused amount of each Lender's Revolving Committed Amount for the applicable period. The Non-Use Fee shall be payable quarterly in arrears on the 15th day following the last day of each calendar quarter for the immediately preceding quarter (or portion thereof) beginning with the first such date to occur after the Closing Date and on the Revolving Commitment Termination Date. For purposes of computing the Non-Use Fee, outstanding Letters of Credit shall be considered usage of the Aggregate Revolving Committed Amount; however, outstanding Swingline Loans shall be considered usage of the Revolving Committed Amount of Bank of Oklahoma only.
- (b) Letter of Credit Fees.
- (i) Letter of Credit Fee. In consideration of the issuance of each Letter of Credit hereunder, the Borrower promises to pay to the Administrative Agent for the benefit of each Lender a fee (the "Letter of Credit Fee") on each Lender's Percentage of the actual daily maximum amount available to be drawn under each

outstanding Letter of Credit from the date of its issuance to the date of its stated expiration, computed at a per annum rate for each day equal to the Applicable Margin for LIBOR Rate Loans. The Letter of Credit Fee shall be payable in advance upon the issuance of each Letter of Credit (or, in the case of the Existing Letters of Credit, upon the Closing Date); provided, however, that if any Letter of Credit has a stated expiration of more than one year from the date of issuance (or provides for automatic renewals beyond a period of one year from the date of issuance), the Letter of Fee shall be payable in advance upon issuance for the first year and on each anniversary date for each subsequent year thereafter.

- (ii) Issuing Lender Fees. In addition to the Letter of Credit Fee payable pursuant to clause (i) above, the Borrower promises to pay to the Issuing Lender, without sharing by the other Lenders, the customary charges from time to time of the Issuing Lender with respect to the issuance, amendment, transfer, administration, cancellation and conversion of, and drawings under, each Letter of Credit.
- (iii) Default Fee. Upon the occurrence and during the continuation of any Event of Default, the Letter of Credit Fee shall accrue at a per annum rate 2% greater than the rate which would otherwise be applicable.
- (c) Administrative Fees. The Borrower agrees to pay to the Administrative Agent, for its own account, the fees referred to in and payable from time to time under the Administrative Agent Fee Letter.
- (d) Commitment Fees. On the Closing Date, the Borrower agrees to pay to the Administrative Agent for the benefit of each of the Lenders, the commitment fees referred to in the Fee Letter.
- (e) Modification Fee. In the event that the Lenders and the Administrative Agent are requested to enter into or to approve any amendment, modification, interpretation, termination, waiver or consent with respect to this Agreement or any of the other Credit Documents, other than an extension of the Facility in accordance with Section 2.5(c), the Borrower shall pay to the Administrative Agent a modification fee of \$5,000 (in addition to any out-of-pocket expenses reimbursable under Section 12.5(a)).
- 3.3 Maturity; Repayment.
- (a) Swingline Loans. The principal amount of each Swingline Loan shall be due and payable on the earlier of (i) the maturity date agreed to by the Swingline Lender and the Borrower at the time such Swingline Loan is made, but in any event not more than 10 days from the date such Swingline Loan was made, and (ii) the Revolving Commitment Termination Date.

- (b) Term Loans. The principal amount of all Term Loans outstanding shall be due and payable in full on the Facility Maturity Date.
- 3.4 Prepayments. The Loans may be repaid in whole or in part without premium or penalty (except, in the case of LIBOR Rate Loans, breakage costs related to prepayments not made on the last day of the relevant Interest Period, as provided in Section 4.7) and partial prepayments shall be minimum principal amounts of \$5,000,000 (except Swingline Loans, which may be prepaid in minimum principal amounts of \$500,000), and in integral multiples of \$500,000 in excess thereof. Unless otherwise directed by the Borrower, prepayments on the Loans shall be applied first to Prime Rate Loans and then to LIBOR Rate Loans in direct order of Interest Period maturities. Amounts prepaid on the Revolving Obligations may, subject to the terms and conditions hereof, be reborrowed. Amounts prepaid on the Term Loans may not be reborrowed.
- 3.5 Sharing of Payments. The Lenders agree among themselves that, in the event that any Lender shall obtain payment in respect of any Loan, LOC Obligations or any other obligation owing to such Lender under this Agreement through the exercise of a right of setoff, banker's lien or counterclaim, or pursuant to a secured claim under Section 506 of the Bankruptcy Code or other security or interest arising from, or in lieu of, such secured claim, received by such Lender under any applicable bankruptcy, insolvency or other similar law or otherwise, or by any other means, in excess of its pro rata share of such payment as provided for in this Agreement, such Lender shall promptly purchase from the other Lenders a Participation Interest in such Loans, LOC Obligations and other obligations in such amounts, and make such other adjustments from time to time, as shall be equitable to the end that all Lenders share such payment in accordance with their respective ratable shares as provided for in this Agreement. The Lenders further agree among themselves that if payment to a Lender obtained by such Lender through the exercise of a right of setoff, banker's lien, counterclaim or other event as aforesaid shall be rescinded or must otherwise be restored, each Lender which shall have shared the benefit of such payment shall, by repurchase of a Participation Interest theretofore sold, return its share of that benefit (together with its share of any accrued interest payable with respect thereto) to each Lender whose payment shall have been rescinded or otherwise restored. The Borrower agrees that any Lender so purchasing such a Participation Interest may, to the fullest extent permitted by law, exercise all rights of payment, including setoff, banker's lien or counterclaim, with respect to such Participation Interest as fully as if such Lender were a holder of such Loan, LOC Obligations or other obligation in the amount of such Participation Interest. Except as otherwise expressly provided in this Agreement, if any Lender or the Administrative Agent shall fail to remit to the Administrative Agent or any other Lender an amount payable by such Lender or the Administrative Agent to the Administrative Agent or such other Lender pursuant to this Agreement on the date when such amount is due, such payments shall be made together with interest thereon for each date from the date such amount is due until the date such amount is paid to the Administrative Agent or such other Lender at a rate per annum equal to the Federal Funds Rate. If under any applicable bankruptcy, insolvency or other similar law, any Lender receives a secured claim in lieu of a setoff to which this Section 3.5 applies, such Lender shall, to the extent practicable, exercise its rights in respect of such secured claim in a manner consistent with the

rights of the Lenders under this Section 3.5 to share in the benefits of any recovery on such secured claim.

- 3.6 Payments. Except as otherwise specifically provided herein, all payments hereunder shall be made to the Administrative Agent, at the address specified in Section 12.1, in Dollars and in immediately available funds, without setoff, deduction, counterclaim or withholding of any kind, not later than 11:00 a.m. (Tulsa, Oklahoma time) on the date when due. Payments received after such time shall be deemed to have been received on the next succeeding Business Day. The Administrative Agent may (but shall not be obligated to) debit the amount of any such payment which is not made by such time to any ordinary deposit account of the Borrower maintained with the Administrative Agent (with notice to the Borrower). The Borrower shall, at the time it makes any payment under this Agreement, specify to the Administrative Agent the Loans, LOC Obligations, Fees, interest or other amounts payable by the Borrower hereunder to which such payment is to be applied (and in the event that it fails so to specify, or if such application would be inconsistent with the terms hereof, the Administrative Agent shall distribute such payment to the Lenders in such manner as the Administrative Agent may determine to be appropriate in respect of obligations owing by the Borrower hereunder, subject to the terms of Section 2.8(a)). The Administrative Agent will distribute such payments to such Lenders, if any such payment is received prior to 11:00 a..m. (Tulsa, Oklahoma time) on a Business Day in like funds as received prior to the end of such Business Day and otherwise the Administrative Agent will distribute such payment to such Lenders on the next succeeding Business Day. Whenever any payment hereunder shall be stated to be due on a day which is not a Business Day, the due date thereof shall be extended to the next succeeding Business Day (subject to accrual of interest and Fees for the period of such extension), except that in the case of LIBOR Rate Loans, if the extension would cause the payment to be made in the next following calendar month, then such payment shall instead be made on the next preceding Business Day.
- 3.7 Computations of Interest and Fees. Except as expressly provided otherwise herein, all computations of interest and fees shall be made on the basis of actual number of days elapsed over a year of 360 days. Interest shall accrue from and include the date of borrowing, but exclude the date of payment.
- 3.8 Maximum Lawful Interest Rate. It is not the intention of the Lenders or the Borrower to violate the laws of any applicable jurisdiction relating to usury or other restrictions on the maximum lawful interest rate. The Credit Documents and all other agreements between the Borrower and the Lenders, whether now existing or hereafter arising and whether written or oral, are hereby limited so that in no event shall the interest paid or agreed to be paid to the Lenders for the use, forbearance or detention of money loaned, or for the payment or performance of any covenant or obligation contained herein or in any other Credit Document, exceed the maximum amount permissible under applicable law. If from any circumstances whatsoever fulfillment of any provision hereof or of any other Credit Document, at the time the performance of such provision shall be due, shall involve transcending the limit of validity prescribed by law, then, ipso facto, the obligation to be fulfilled shall be reduced to the limit of such validity. If from any such circumstances the Lenders should ever receive anything of value deemed interest under applicable law which would exceed interest at the highest lawful rate, such excessive interest shall be

applied to the reduction of the principal amount owing hereunder, and not to the payment of interest, or if such excessive interest exceeds any unpaid balance of principal, such excess shall be refunded to the Borrower. All sums paid or agreed to be paid to the Lenders for the use, forbearance or detention of monies advances under the Facility shall, to the extent permitted by applicable law, be amortized, prorated, allocated and spread throughout the full term of the Obligations until payment in full so that the rate of interest on account of the Obligations is uniform throughout the term thereof. This Section 3.8 shall control every other provision of the Credit Documents and all other agreements between the Lenders and the Borrower contemplated thereby.

ARTICLE IV YIELD PROTECTION; OTHER PROVISIONS RELATING TO CREDIT FACILITY

- 4.1 Capital Adequacy. If any Lender has determined that, after the Closing Date, the adoption or the becoming effective of, or any change in, or any change by any Governmental Authority, central bank or comparable agency charged with the interpretation or administration thereof in the interpretation or administration of, any law, rule or regulation regarding capital adequacy applicable to such Lender (or its parent holding company or its Applicable Lending Office), or compliance by such Lender (or its parent holding company or its Applicable Lending Office) with any request or directive regarding capital adequacy (whether or not having the force of law) of any such Governmental Authority, central bank or comparable agency, has or would have the effect of reducing the rate of return on such Lender's capital or assets as a consequence of its commitments or obligations hereunder to a level below that which such Lender could have achieved but for such adoption, effectiveness, change or compliance (taking into consideration such Lender's policies with respect to capital adequacy), then, upon notice from such Lender to the Borrower, the Borrower shall be obligated to pay to such Lender such additional amount or amounts as will compensate such Lender for such reduction. Each determination by any such Lender of amounts owing under this Section 4.1 shall, absent manifest error, be conclusive and binding on the parties hereto.
- 4.2 Limitation on LIBOR Rate Loans. If on or prior to the first day of any Interest Period for any LIBOR Rate Loan, (i) the Administrative Agent determines (which determination shall be conclusive) that by reason of circumstances affecting the relevant market, adequate and reasonable means do not exist for ascertaining the Adjusted LIBOR Rate for such Interest Period, or
- (ii) the Required Lenders determine (which determination shall be conclusive) and notify the Administrative Agent that the Adjusted LIBOR Rate will not adequately and fairly reflect the cost to the Lenders of funding LIBOR Rate Loans for such Interest Period, then in either of such events the Administrative Agent shall give the Borrower prompt notice thereof, and so long as such condition remains in effect, the Lenders shall be under no obligation to make additional LIBOR Rate Loans, continue LIBOR Rate Loans, or to convert Prime Rate Loans into LIBOR Rate Loans, and the Borrower shall, on the last day(s) of the then current Interest Period(s) for the outstanding LIBOR Rate Loans, either prepay such LIBOR Rate Loans or convert such LIBOR Rate Loans into Prime Rate Loans in accordance with the terms of this Agreement.
- 4.3 Illegality. Notwithstanding any other provision of this Agreement, in the event that it becomes unlawful for any Lender (or its Applicable Lending Office) to make, maintain, or

fund LIBOR Rate Loans hereunder, then such Lender shall promptly notify the Borrower and Administrative Agent thereof and such Lender's obligation to make or continue LIBOR Rate Loans and to convert Prime Rate Loans into LIBOR Rate Loans shall be suspended until such time as such Lender may again make, maintain, and fund LIBOR Rate Loans (in which case the provisions of Section 4.5 shall be applicable).

- 4.4 Requirements of Law. If, after the date hereof, the adoption of any applicable law, rule, or regulation, or any change in any applicable law, rule, or regulation, or any change in the interpretation or administration thereof by any Governmental Authority, central bank, or comparable agency charged with the interpretation or administration thereof, or compliance by any Lender (or its Applicable Lending Office) with any request or directive (whether or not having the force of law) of any such Governmental Authority, central bank, or comparable agency:
- (i) shall subject such Lender (or its Applicable Lending Office) to any Taxes (as defined in Section 4.6(a)) with respect to any LIBOR Rate Loans, its Notes, any Letter of Credit or any Lender's participation therein, any LOC Documents, or its obligation to make LIBOR Rate Loans, or change the basis of taxation of any amounts payable to such Lender (or its Applicable Lending Office) under this Agreement or its Notes in respect of any LIBOR Rate Loans (other than a change in the rate of tax);
- (ii) shall impose, modify, or deem applicable any reserve, special deposit, assessment, or similar requirement (other than the Eurodollar Reserve Requirement utilized in the determination of the Adjusted LIBOR Rate) relating to any extensions of credit or other assets of, or any deposits with or other liabilities or commitments of, such Lender (or its Applicable Lending Office), including the Commitment of such Lender hereunder; or
- (iii) shall impose on such Lender (or its Applicable Lending Office) or the London interbank market any other condition affecting this Agreement or its Notes or any of such extensions of credit or liabilities or commitments (including any Letter of Credit or any Lender's participation therein or any LOC Documents);

and the result of any of the foregoing is to increase the cost to such Lender (or its Applicable Lending Office) of making, converting into, continuing, or maintaining any LIBOR Rate Loans or any Letter of Credit or any Lender's participation therein or any LOC Documents or to reduce any sum received or receivable by such Lender (or its Applicable Lending Office) under this Agreement or its Notes with respect to any LIBOR Rate Loans or any Letter of Credit or LOC Document, then the Borrower shall pay to such Lender on demand such amount or amounts as will compensate such Lender for such increased cost or reduction. If any Lender requests compensation by the Borrower under this Section 4.4, the Borrower may, by notice to such Lender (with a copy to the Administrative Agent), suspend the obligation of such Lender to make or continue LIBOR Rate Loans, or to convert Prime Rate Loans into LIBOR Rate Loans, until the event or condition giving rise to such request ceases to be in effect (in which case the provisions of Section 4.5 shall be applicable); provided that such suspension shall not affect the

right of such Lender to receive the compensation so requested. Each Lender shall promptly notify the Borrower and the Administrative Agent of any event of which it has knowledge, occurring after the date hereof, which will entitle such Lender to compensation pursuant to this Section 4.4 and will designate a different Applicable Lending Office if such designation will avoid the need for, or reduce the amount of, such compensation and will not, in the sole judgment of such Lender, be otherwise disadvantageous to it; provided, that in determining whether designating a different Applicable Lending Office would be disadvantageous to such Lender, such Lender shall disregard any economic disadvantage that the Borrower agrees in form and substance satisfactory to such Lender to indemnify and hold such Lender harmless therefrom. Any Lender claiming compensation under this Section 4.4 shall furnish to the Borrower and the Administrative Agent a statement setting forth the additional amount or amounts to be paid to it hereunder which shall be conclusive in the absence of manifest error. In determining such amount, such Lender may use any reasonable averaging and attribution methods.

- 4.5 Treatment of Affected Loans. If the obligation of any Lender to make any LIBOR Rate Loan or to continue, or to convert Prime Rate Loans into, LIBOR Rate Loans shall be suspended pursuant to Section 4.3 or 4.4, such Lender's LIBOR Rate Loans shall be automatically converted into Prime Rate Loans on the last day(s) of the then current Interest Period(s) for such LIBOR Rate Loans (or, in the case of a conversion required by Section 4.4, on such earlier date as such Lender may specify to the Borrower with a copy to the Administrative Agent) and, unless and until such Lender gives notice as provided below that the circumstances specified in Section 4.3 or 4.4 that gave rise to such conversion no longer exist:
- (i) to the extent that such Lender's LIBOR Rate Loans have been so converted, all payments and prepayments of principal that would otherwise be applied to such Lender's LIBOR Rate Loans shall be applied instead to its Prime Rate Loans; and
- (ii) all Loans that would otherwise be made or continued by such Lender as LIBOR Rate Loans shall be made or continued instead as Prime Rate Loans, and all Prime Rate Loans of such Lender that would otherwise be converted into LIBOR Rate Loans shall remain as Prime Rate Loans.

If such Lender gives notice to the Borrower (with a copy to the Administrative Agent) that the circumstances specified in Section 4.3 or 4.4 that gave rise to the conversion of such Lender's LIBOR Rate Loans pursuant to this Section 4.5 no longer exist (which such Lender agrees to do promptly upon such circumstances ceasing to exist) at a time when LIBOR Rate Loans made by other Lenders are outstanding, such Lender's Prime Rate Loans shall be automatically converted, on the first day(s) of the next succeeding Interest Period(s) for such outstanding LIBOR Rate Loans, to the extent necessary so that, after giving effect thereto, all Loans held by the Lenders holding LIBOR Rate Loans and by such Lender are held pro rata (as to principal amounts, interest rate basis, and Interest Periods) in accordance with their respective Commitments.

4.6 Taxes.

- (a) No Deductions. Any and all payments by the Borrower to or for the account of any Lender or the Administrative Agent hereunder or under any other Credit Document shall be made free and clear of and without deduction for any and all present or future taxes, duties, levies, imposts, deductions, charges or withholdings, and all liabilities with respect thereto, excluding, in the case of each Lender and the Administrative Agent, taxes imposed on its net income, and franchise taxes imposed in lieu of a tax on net income, by the jurisdiction under the laws of which such Lender (or its Applicable Lending Office) or the Administrative Agent (as the case may be) is organized or maintained or any political subdivision thereof (all such non-excluded taxes, duties, levies, imposts, deductions, charges, withholdings, and liabilities being hereinafter referred to as "Taxes"). If the Borrower shall be required by law to deduct or withhold any Taxes from or in respect of any sum payable under this Agreement or any other Credit Document to any Lender or the Administrative Agent, (i) except as provided in Section 4.6(e), the sum payable shall be increased as necessary so that after making all required deductions or withholdings for or on account of Taxes applicable to additional sums payable under this Section 4.6) such Lender or the Administrative Agent receives an amount equal to the sum it would have received had no such deductions been made, (ii) the Borrower shall make such deductions or withholdings, (iii) the Borrower shall pay the full amount deducted or withheld to the relevant taxation authority or other Governmental Authority in accordance with applicable law, and (iv) within thirty (30) days after the date of any payment of Taxes, the Borrower shall furnish to the Administrative Agent, at its address referred to in Section 12.1, the original or a certified copy of a receipt evidencing payment thereof.
- (b) Other Taxes. In addition, the Borrower agrees to pay any and all present or future stamp or documentary taxes and any other excise or property taxes or charges or similar levies which arise from any payment made under this Agreement or any other Credit Document or from the execution or delivery of, or otherwise with respect to, this Agreement or any other Credit Document (hereinafter referred to as "Other Taxes").
- (c) Indemnity. Except as provided in Section 4.6(e), the Borrower agrees to indemnify each Lender and the Administrative Agent for the full amount of Taxes and Other Taxes (including any Taxes or Other Taxes imposed or asserted by any jurisdiction on amounts payable under this Section 4.6 for or on account of Taxes or Other Taxes) paid by such Lender or the Administrative Agent (as the case may be) and any liability (including penalties, interest, and expenses) arising therefrom or with respect thereto.
- (d) Exemption from Backup Withholding. If requested in writing by the Borrower or the Administrative Agent, each Lender shall provide the Borrower and the Administrative Agent with accurate and complete original signed copies of Internal Revenue Service Form W-9 or any successor form prescribed by the Internal Revenue Service certifying that such Lender is entitled to a complete exemption from United States withholding tax on payments pursuant to this Agreement or any of the other Credit Documents.

(e) Non-U.S. Lenders. Each Lender that is not a United States person under Section 7701(a)(30) of the IRC for U.S. federal income tax purposes, on or prior to the date on which it becomes a Lender under this Agreement pursuant to Section 4.8 or 12.3), and from time to time thereafter if (x) a lapse in time or change in circumstances renders the previous certification obsolete or inaccurate in any material respect or (y) requested in writing by the Borrower or the Administrative Agent (but only so long as such Lender remains lawfully able to do so), shall provide the Borrower and the Administrative Agent with (i) two accurate and complete original signed copies of Internal Revenue Service Form W-8BEN or W-8ECI, as appropriate, or any successor form prescribed by the Internal Revenue Service, certifying that such Lender is entitled to benefits under an income tax treaty to which the United States is a party which completely eliminates United States withholding tax on all payments hereunder or under any other Credit Documents or certifying that all such payments are effectively connected with the conduct of a trade or business in the United States, or (ii) if such Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the IRC and cannot deliver either Internal Revenue Service Form W-8BEN (with respect to a complete exemption under an income tax treaty) or Form W-8ECI pursuant to clause (i), (A) a certificate, in form satisfactory to the Borrower and the Administrative Agent, and (B) two accurate and complete original signed copies of Form W-8BEN or any successor form prescribed by the Internal Revenue Service, certifying that such Lender is entitled to a complete exemption from United States withholding tax on payments of interest pursuant to this Agreement or any of the other Credit Documents. For any period with respect to which a Lender has failed to provide the Borrower and the Administrative Agent with the appropriate form pursuant to this Section 4.6(e) (unless such failure is due to a change in treaty, law, or regulation (or in the official interpretation thereof) enacted or promulgated subsequent to the date on which a form with respect to such Lender originally was required to be provided), such Lender shall not be entitled to indemnification under Section 4.6(a) or 4.6(c) with respect to Taxes imposed by the United States; provided, however, that should a Lender, which is otherwise exempt from United States withholding tax, become subject to Taxes because of its failure to deliver a form required hereunder, the Borrower shall take such steps as such Lender shall reasonably request to assist such Lender, at such Lender's expense, to recover such Taxes. In addition, for any period with respect to which a Lender has provided the Borrower and the Administrative Agent with the Internal Revenue Service forms specified in clause (ii) of the first sentence of this Section 4.6 (e), such Lender shall not be entitled to indemnification under Section 4.6(a) or 4.6(c) with respect to Taxes imposed by the United States other than Taxes imposed on interest.

(f) Lending Offices. If the Borrower is required to pay additional amounts to or for the account of any Lender pursuant to this Section 4.6, then such Lender will agree to use reasonable efforts to change the jurisdiction of its Applicable Lending Office so as to eliminate or reduce any such additional payment which may thereafter accrue if such change, in the sole judgment of such Lender, is not otherwise materially disadvantageous to such Lender; provided, that in determining whether changing the jurisdiction of an Applicable Lending Office would be disadvantageous to such Lender, such Lender shall

disregard any economic disadvantage that the Borrower agrees in form and substance satisfactory to such Lender to indemnify and hold such Lender harmless from.

- (g) Refund to Borrower. If any Lender determines that it has actually received or realized any refund of tax, any reduction of, or credit against, its tax liabilities or otherwise recovered any amount in connection with any deduction or withholding, or payment of any additional amount, by the Borrower pursuant to Section 4.4 or 4.6, such Lender shall reimburse the Borrower an amount that the Lender shall, in its sole discretion, determine is equal to the net benefit, after tax, which was actually obtained by the Lender as a consequence of such refund, reduction, credit or recovery; provided, that nothing in this subsection (f) shall require any Lender to make available its tax returns (or any other information relating to its taxes which it deems to be confidential).
- (h) Survival. Without prejudice to the survival of any other agreement of the Borrower hereunder, the agreements and obligations of the Borrower contained in this Section 4.6 shall survive the repayment of the Loans, the LOC Obligations and other obligations under the Credit Documents and the termination of the Commitments hereunder.
- 4.7 Compensation. Upon the request of any Lender, the Borrower shall pay to such Lender such amount or amounts as shall be sufficient (in the reasonable opinion of such Lender) to compensate it for any loss, cost, or expense (excluding loss of anticipated profits) incurred by it as a result of:
- (i) any payment, prepayment, or conversion of a LIBOR Rate Loan for any reason (including the acceleration of the Loans pursuant to Section 9.2) on a date other than the last day of the Interest Period for such Loan; or (ii) any failure by the Borrower for any reason (including the failure of any condition precedent specified in Article V to be satisfied) to borrow, convert, continue, or prepay a LIBOR Rate Loan on the date for such borrowing, conversion, continuation, or prepayment specified in the relevant Notice of Borrowing, prepayment, continuation, or conversion under this Agreement. With respect to LIBOR Rate Loans, such indemnification may include an amount equal to the excess, if any, of (i) the amount of interest which would have accrued on the amount so prepaid, or not so borrowed, converted or continued, for the period from the date of such prepayment or of such failure to borrow, convert or continue to the last day of the applicable Interest Period (or, in the case of a failure to borrow, convert or continue, the Interest Period that would have commenced on the date of such failure) in each case at the applicable rate of interest for such LIBOR Rate Loans provided for herein (excluding, however, the Applicable Margin included therein, if any) over (ii) the amount of interest (as reasonably determined by such Lender) which would have accrued to such Lender on such amount by placing such amount on deposit for a comparable period with leading banks in the interbank Eurodollar market. Without prejudice to the survival of any other agreement of the Borrower hereunder, the covenants of the Borrower set forth in this Section 4.7 shall survive the repayment of the Loans, LOC Obligations and other obligations under the Credit Documents and the termination of the Commitments hereunder.
- 4.8 Replacement of Lenders. The Borrower shall have the right, if no Default then exists, to replace any Lender (a "Replaced Lender") with one or more other Eligible Assignees

reasonably acceptable to the Administrative Agent (collectively, the "Replacement Lender") if such Lender is charging the Borrower increased costs pursuant to Section 4.4 or 4.6 in excess of those being charged generally by the other Lenders or such Lender becomes incapable of making LIBOR Rate Loans as provided in Section 4.3 when other Lenders are generally able to do so; provided, however, that (i) at the time of any replacement pursuant to this

Section 4.8, the Replacement Lender shall enter into one or more assignment agreements in accordance with Section 12.3(b) (and with all fees payable pursuant to Section 12.3(b) to be paid by the Replacement Lender) pursuant to which the Replacement Lender shall acquire all of the Commitments and outstanding Revolving Obligations of the Replaced Lender and, in connection therewith, shall pay to (x) the Replaced Lender, an amount equal to the sum of (A) the principal of, and all accrued interest on, all outstanding Loans of the Replaced Lender, (B) all other Obligations owing to such Replaced Lender, together with all then unpaid interest with respect thereto at such time, and (C) all accrued, but theretofore unpaid, fees owing to the Replaced Lender pursuant to Section 3.2, and (y) the Issuing Lender, an amount equal to such Replaced Lender's Revolving Commitment Percentage of a participation in any LOC Obligations (which at such time remains a LOC Obligation) to the extent such amount was not theretofore purchased by such Replaced Lender pursuant to Section 2.6(b), and (ii) all obligations of the Borrower owing to the Replaced Lender (other than those specifically described in clause (i) above in respect of which the assignment purchase price has been, or is concurrently being, paid, but including any amounts which would be paid to a Lender pursuant to Section 4.7 if the Borrower were prepaying a LIBOR Rate Loan) shall be paid in full to such Replaced Lender concurrently with such replacement. Upon the execution of the respective assignment agreement, the payment of amounts referred to in clauses (i) and (ii) above and, if so requested by the Replacement Lender, delivery to the Replacement Lender of a Note executed by the Borrower, the Replacement Lender shall become a Lender hereunder and the Replaced Lender shall cease to constitute a Lender hereunder and be released of all its obligations as a Lender, except with respect to indemnification provisions applicable to the Replaced Lender under this Agreement, which shall survive as to such Replaced Lender.

ARTICLE V CLOSING; CONDITIONS PRECEDENT

- 5.1 Closing Date. The closing of the transactions contemplated by this Agreement will take place, and the initial Extensions of Credit will be made, on July 16, 2002, or at such other date (not later than July 31, 2002) as the parties shall mutually agree (the "Closing Date"), provided that all conditions to the closing set forth in Section 5.2 have been satisfied.
- 5.2 Closing Conditions. The obligation of the Lenders to enter into this Agreement and to make the initial Extensions of Credit hereunder is subject to satisfaction of the following conditions precedent at or as of the Closing Date (in each case in form and substance acceptable to the Lenders, unless otherwise specified):
- (a) Executed Credit Documents. Receipt by the Administrative Agent of multiple counterparts of the Credit Documents, including this Agreement, the Notes

(provided that there shall only be one original of each Note) and the Subsidiary Guaranty, each duly executed by an appropriate officer of each of the parties thereto.

- (b) Financial Information. Receipt by the Administrative Agent of such financial information regarding the Borrower and the Consolidated Group as may be requested by, and in each case in form and substance satisfactory to, the Administrative Agent and the Lenders, including, to the extent available, any of the information otherwise required to be delivered pursuant to Section 7.1.
- (c) Corporate Documents. Receipt by the Administrative Agent of the following:
- (i) Charter Documents. Copies of the Articles or Certificate of Incorporation, Articles of Organization or other similar publicly-filed organizational documents of each of the Credit Parties, each certified to be true and complete as of a recent date by the appropriate Governmental Authority of the state or other jurisdiction of its incorporation or, if certification by the appropriate Governmental Authority is not available, certified by its secretary or assistant secretary to be true and correct as of the Closing Date.
- (ii) Bylaws. A copy of the bylaws, operating agreement or other internal governance documents of each of the Credit Parties, each certified by its secretary or assistant secretary to be true and correct as of the Closing Date.
- (iii) Resolutions. Copies of resolutions of the Board of Directors of each of the Credit Parties approving and adopting the Credit Documents to which it is a party, the transactions contemplated therein and authorizing execution and delivery thereof, certified by its secretary or assistant secretary to be true and correct and in force and effect as of the Closing Date.
- (iv) Good Standing. Copies of certificates of good standing, existence or its equivalent with respect to each of the Credit Parties certified as of a recent date by the appropriate Governmental Authorities of the state or other jurisdiction of incorporation and each other jurisdiction in which the failure to so qualify and be in good standing could reasonably be expected to have a Material Adverse Effect.
- (v) Incumbency Certificate. An incumbency certificate with respect to each of the Credit Parties, certified by its secretary or assistant secretary to be true and correct as of the Closing Date.
- (d) Opinion of Counsel. Receipt by the Administrative Agent of opinions of McAfee & Taft A Professional Corporation, counsel to the Parent and the Borrower, and Steven R. Mackey, Vice President and General Counsel of the Parent, as to the matter set forth on Exhibit "F" and such other matters as the Administrative Agent or its counsel

may reasonably request at least three Business Days prior to the Closing Date (such opinions may be based upon such customary assumptions and may contains such customary qualifications and exceptions as may be approved by the Administrative Agent and its counsel).

- (e) Certificate of Chief Financial Officer. A certificate, signed by the chief financial officer of the Parent, stating that, as of the Closing Date and to the best of such officer's knowledge, (i) no Default or Event of Default has occurred and is continuing, (ii) the representations and warranties made by the Borrower and the Parent herein or by the Credit Parties in any other Credit Documents are true and correct, and (iii) except as set forth on Schedule 6.5 and except as contemplated by Section 12.17, since September 30, 2001, there has been no circumstance, development or event which has had or could reasonably be expected to have a Material Adverse Effect. Such certificate shall also set forth calculations demonstrating the Borrower's compliance with Sections 7.11(c) and 7.12.
- (f) Fees and Expenses. Payment by the Credit Parties of all fees and expenses owed by them to the Lenders and the Administrative Agent, including the reasonable fees and expenses of counsel to the Administrative Agent.
- (g) Other. Receipt by the Lenders of such other documents, instruments, agreements or information as reasonably requested by any Lender.
- (h) Bank One Agreement. Receipt by the Administrative Agent and the Lenders of satisfactory evidence that (i) any existing defaults or events of default under the Bank One Agreement have been cured or permanently waived, and (ii) the Bank One Agreement has been modified such that the affirmative and negative covenants contained therein and the events of default specified therein are no more restrictive than the covenants contained herein and the Events of Defaults specified herein.

In addition, there shall not have occurred or become known any material adverse change or any condition or event that could reasonably be expected to result in a material adverse change in the business, operations, financial condition, liabilities (contingent or otherwise) or prospects of the Consolidated Group taken as a whole since the dates of the latest financial information furnished to the Administrative Agent and the Lenders.

- 5.3 Conditions to All Extensions of Credit. The obligation of each Lender to make any Extension of Credit hereunder (including the initial Extension of Credit to be made hereunder) is subject to the satisfaction of the following additional conditions precedent on the date of making such Extension of Credit (in addition to the conditions set forth in Article II):
- (a) Representations and Warranties. The representations and warranties made by the Borrower and the Parent herein or by the Credit Parties in any other Credit Documents or which are contained in any certificate furnished at any time under or in connection herewith shall be true and correct in all material respects on and as of the date

of such Extension of Credit as if made on and as of the date of such extension or such request, as applicable (except for those which expressly relate to an earlier specified date).

- (b) No Default or Event of Default. No Default or Event of Default shall have occurred and be continuing on such date or after giving effect to the Extension of Credit to be made on such date and the application of the proceeds thereof unless such Default or Event of Default shall have been waived in accordance with this Agreement.
- (c) Bankruptcy or Insolvency. No Bankruptcy Event shall have occurred by or with respect to the Borrower, the Parent or any of the Parent's Subsidiaries.
- (d) No Material Adverse Effect. No circumstance, event or condition shall have occurred or be existing which could reasonably be expected to have a Material Adverse Effect.

Each request for an Extension of Credit (including extensions and conversions) and each acceptance by the Borrower of an Extension of Credit (including extensions and conversions) shall be deemed to constitute a representation and warranty by the Borrower as of the date of such Extension of Credit that the applicable conditions in subsections (a), (b), (c) and (d) of this Section 5.3 have been satisfied.

ARTICLE VI REPRESENTATIONS AND WARRANTIES

To induce the Lenders to make the Extensions of Credit hereunder to the Borrower, the Borrower and the Parent hereby jointly and severally represent and warrant to the Administrative Agent and to each Lender that:

- 6.1 Organization; Existence; Compliance with Law. Each of the members of the Consolidated Group (i) is duly organized, validly existing in good standing under the laws of the jurisdiction of its incorporation or organization, (ii) has the corporate or other necessary organizational power and authority, and the legal right to own and operate its Properties, to lease the Properties it operates as lessee and to conduct the business in which it is currently engaged, (iii) is duly qualified as a foreign entity and in good standing under the laws of each jurisdiction where its ownership, lease or operation of Property or the conduct of its business requires such qualification, other than in such jurisdictions where the failure to be so qualified and in good standing has not had and could not have a Material Adverse Effect, and (iv) is in compliance with all Requirements of Law, except to the extent that the failure to comply therewith has not had and could not be reasonably expected to have a Material Adverse Effect.
- 6.2 Existence and Authorization. Each of the Credit Parties has the corporate or other necessary organizational power and authority, and the legal right, to make, deliver and perform the Credit Documents to which it is a party and has taken all necessary corporate or other action to authorize the execution, delivery and performance by it of the Credit Documents to which it is

a party. No consent or authorization of, filing with, notice to or other act by or in respect of, any Governmental Authority or any other Person is required in connection with acceptance of Extensions of Credit or the making of the guaranties hereunder or with the execution, delivery or performance of any Credit Documents by the Credit Parties (other than those which have been obtained, such filings as are required by the SEC and to fulfill other reporting requirements with Governmental Authorities) or with the validity or enforceability of any Credit Document against the Credit Parties.

- 6.3 Enforceability. Each Credit Document to which it is a party constitutes a legal, valid and binding obligation of such Credit Party enforceable against such Credit Party in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether enforcement is sought by proceedings in equity or at law).
- 6.4 No Legal Bar. The execution, delivery and performance of the Credit Documents, the borrowings hereunder and the use of the Extensions of Credit will not violate any Requirement of Law or any Contractual Obligation of any member of the Consolidated Group (except those as to which waivers or consents have been obtained and those which could not reasonably be expected to have a Material Adverse Effect), and will not result in, or require, the creation or imposition of any Lien on any of its respective properties or revenues pursuant to any Requirement of Law or Contractual Obligation. No member of the Consolidated Group is in default under or with respect to any of its Contractual Obligations in any respect which has had or could reasonably be expected to have a Material Adverse Effect.
- 6.5 Financial Condition. The audited consolidated balance sheets of the Parent and its consolidated Subsidiaries dated as of September 30, 1999, September 30, 2000, and September 30, 2001, respectively, together, in each case, with the related audited statements of income and cash flows, and the unaudited company-prepared consolidated balance sheet for the Parent and its consolidated Subsidiaries, dated as of March 31, 2002, together with related consolidated statements of income and cash flows, copies of which have previously been provided to the Administrative Agent for distribution to the Lenders, were prepared in accordance with GAAP consistently applied throughout the periods covered thereby, are complete and correct in all material respects and present fairly the financial condition (including disclosure of all material liabilities, contingent or otherwise) and results of operations of the Persons and for the periods specified, subject in the case of interim company-prepared statements to normal year-end adjustments and the absence of footnotes. Except as set forth on Schedule 6.5, since September 30, 2001, and except as contemplated by Section 12.17, there has been no circumstance, development or event which has had or could reasonably be expected to have a Material Adverse Effect.

6.6 No Material Litigation and Disputes. No unsealed litigation or, to the best knowledge of the Credit Parties, claims, investigation, sealed litigation or proceeding of or before any arbitrator or Governmental Authority is pending or, to the best knowledge of the Credit Parties, threatened by or against, any members of the Consolidated Group or against any of their

respective properties or revenues which (a) relate to the Credit Documents or any of the transactions contemplated hereby or thereby or (b) if adversely determined, could, after giving effect to any applicable insurance, reasonably be expected to have a Material Adverse Effect.

- 6.7 No Defaults. No Default or Event of Default has occurred and is continuing. No default exists, nor, to the best knowledge of the Credit Parties, is any such default asserted under any Contractual Obligations to which any member of the Consolidated Group is a party which individually or in the aggregate has had or could reasonably be expected to have a Material Adverse Effect.
- 6.8 Ownership and Operation of Property. Each of the members of the Consolidated Group has good title to, or a valid leasehold interest in, all of its material real property, and good title to, or a valid leasehold interest in, all of its other material Property, and none of such Properties is subject to any Lien, except for Permitted Liens.
- 6.9 Licenses. Each of the members of the Consolidated Group has obtained all material Licenses, governmental or private, necessary to the ownership of its Properties and to the conduct of its business, which are material to the Consolidated Group taken as a whole.
- 6.10 No Burdensome Restrictions. None of the members of the Consolidated Group is subject to any Requirement of Law or any Contractual Obligation which has had or could be reasonably expected to have a Material Adverse Effect.
- 6.11 Taxes. Each of the members of the Consolidated Group has filed or caused to be filed all income tax returns (federal, state, local and foreign) and all other material tax returns which are required to be filed and has paid
- (i) all amounts shown therein to be due (including interest and penalties) and
- (ii) all other taxes, fees, assessments and other governmental charges (including mortgage recording taxes, documentary stamp taxes and intangibles taxes) owing, except for such taxes which are not yet delinquent or as are being contested in good faith by proper proceedings and against which adequate reserves are being maintained in accordance with GAAP. No tax claim or assessment has been asserted in writing against members of the Consolidated Group which has had or, if adversely determined, could reasonably be expected to have a Material Adverse Effect.

6.12 ERISA Matters.

(i) During the five-year period prior to the date on which this representation is made or deemed made: (A) no ERISA Event has occurred, and, to the best knowledge of the Credit Parties, no event or condition has occurred or exists as a result of which any ERISA Event could reasonably be expected to occur, with respect to any Plan; (B) no "accumulated funding deficiency," as such term is defined in Section 302 of ERISA and Section 412 of the IRC, whether or not waived, has occurred with respect to any Plan; (C) each Plan has been maintained, operated, and funded in compliance with its own terms and in material compliance with the provisions of ERISA, the IRC, and any other applicable federal or state laws, except to the extent any noncompliance could not

reasonably be expected to have a Material Adverse Effect; and (D) no Lien in favor of the PBGC or a Plan has arisen or is reasonably likely to arise on account of any Plan.

- (ii) The actuarial present value of all "benefit liabilities" (as defined in Section 4001(a)(16) of ERISA), whether or not vested, under each Single Employer Plan, as of the last annual valuation date prior to the date on which this representation is made or deemed made (determined, in each case, in accordance with Financial Accounting Standards Board Statement 87, utilizing the actuarial assumptions used in such Plan's most recent actuarial valuation report), did not exceed as of such valuation date the fair market value of the assets of such Plan.
- (iii) No member of the Consolidated Group nor any ERISA Affiliate has incurred, or, to the best knowledge of the Credit Parties, could be reasonably expected to incur, any withdrawal liability under ERISA to any Multiemployer Plan or Multiple Employer Plan, except to the except to the extent any such withdrawal liability could not reasonably be expected to have a Material Adverse Effect. No member of the Consolidated Group nor any ERISA Affiliate would become subject to any withdrawal liability under ERISA if any member of the Consolidated Group or any ERISA Affiliate were to withdraw completely from all Multiemployer Plans and Multiple Employer Plans as of the valuation date most closely preceding the date on which this representation is made or deemed made. No member of the Consolidated Group nor any ERISA Affiliate has received any notification that any Multiemployer Plan is in reorganization (within the meaning of Section 4241 of ERISA), is insolvent (within the meaning of Section 4245 of ERISA), or has been terminated (within the meaning of Title IV of ERISA), and no Multiemployer Plan is, to the best knowledge of the Credit Parties, reasonably expected to be in reorganization, insolvent, or terminated.
- (iv) To the best knowledge of the Credit Parties, no prohibited transaction (within the meaning of Section 406 of ERISA or Section 4975 of the IRC) or breach of fiduciary responsibility has occurred with respect to a Plan which has subjected or may subject any member of the Consolidated Group or any ERISA Affiliate to any liability under Sections 406, 409, 502(i), or 502(l) of ERISA or Section 4975 of the IRC, or under any agreement or other instrument pursuant to which any member of the Consolidated Group or any ERISA Affiliate has agreed or is required to indemnify any person against any such liability.
- (v) No member of the Consolidated Group nor any ERISA Affiliates has any liability with respect to "expected post-retirement benefit obligations" within the meaning of the Financial Accounting Standards Board Statement 106. Each Plan which is a welfare plan (as defined in Section 3(1) of ERISA) to which Sections 601-609 of ERISA and Section 4980B of the IRC apply has been administered in compliance in all material respects of such sections, except to the extent any noncompliance could not reasonably be expected to have a Material Adverse Effect.
- 6.13 Governmental Regulations.

- (a) Margin Stock. No part of the proceeds of the Extensions of Credit hereunder will be used, directly or indirectly, for the purpose of purchasing or carrying any "margin stock" within the meaning of Regulation U, or for the purpose of purchasing or carrying or trading in any other securities. If requested by any Lender or the Administrative Agent, the Borrower will furnish to the Administrative Agent and each Lender a statement to the foregoing effect in conformity with the requirements of FR Form U-1 referred to in said Regulation U. No indebtedness being reduced or retired out of the proceeds of the Extensions of Credit hereunder was or will be incurred for the purpose of purchasing or carrying any margin stock within the meaning of Regulation U or any "margin security" within the meaning of Regulation T. "Margin stock" within the meanings of Regulation U does not constitute more than 25% of the value of the consolidated assets of the Borrower and its Subsidiaries.
- (b) Securities Laws. None of the transactions contemplated by this Agreement (including the direct or indirect use of the proceeds of the Loans) will violate or result in a violation of the Securities Act of 1933, as amended, or the Securities Exchange Act, or regulations issued pursuant thereto, or Regulation T, U or X.
- (c) Investment Company Act; Public Utility Holding Company Act. None of the members of the Consolidated Group is subject to regulation under the Public Utility Holding Company Act of 1935, the Federal Power Act or the Investment Company Act of 1940, each as amended. In addition, none of the members of the Consolidated Group is
- (i) an "investment company" registered or required to be registered under the Investment Company Act of 1940, as amended, and is not controlled by such a company, or (ii) a "holding company," or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company," within the meaning of the Public Utility Holding Company Act of 1935, as amended.
- (d) Regulation O. No director, executive officer or principal shareholder of any member of the Consolidated Group is a director, executive officer or principal shareholder of any Lender. For the purposes hereof the terms "director," "executive officer" and "principal shareholder" (when used with reference to any Lender) have the respective meanings assigned thereto in Regulation O.
- 6.14 Subsidiaries. Set forth on Schedule 6.14 are all the Subsidiaries of the Parent, including the jurisdiction of organization, classes of Capital Stock or other equity interests (including options, warrants, rights of subscription, conversion and exchangeability and other similar rights) and ownership and ownership percentages thereof. The outstanding shares of Capital Stock shown have been validly issued, fully paid and are non-assessable and owned free of Liens other than Permitted Liens.
- 6.15 Purpose of Extensions of Credit. The proceeds of the Revolving Loans will be used solely for the purposes stated in Section 2.2(c).

6.16 Environmental Matters.

- (i) All of the facilities and Properties owned, leased or operated by each member of the Consolidated Group, and all of the Consolidated Group's respective operations at such facilities and Properties, are in compliance with all applicable Environmental Laws, and there is no violation of any Environmental Law with respect to such Properties or businesses, except to the extent any noncompliance or violation has not resulted and could not reasonably be expected to result in a Material Adverse Effect.
- (ii) No Hazardous Materials are located or have been released on, under or about any of the facilities and Properties owned, leased or operated by any member of the Consolidated Group in amounts or concentrations that constitute a violation of, or could give rise to liability under, any Environmental Laws, and no Hazardous Materials have been generated, treated, stored, released from or disposed of at, on or under any of such facilities or Properties or any other location, or transported from such Properties to or disposed of at any other location by or on behalf of any members of the Consolidated Group in violation of any Environmental Law, except to the extent any such violation or liability has not resulted and could not reasonably be expected to result in a Material Adverse Effect.
- (iii) No member of the Consolidated Group has (A) received any written notice of any judicial proceeding or administrative action that is pending or threatened under any Environmental Law with respect to any member of the Consolidated Group or its facilities, Properties or business, (B) entered into or been subject to any consent decree, consent order, administrative order or compliance order, or other administrative or judicial requirement that remains outstanding under any Environmental Law, or (C) been subject to any governmental enforcement action under any Environmental Laws with respect to any of its facilities or Properties or any off-site waste disposal site.
- 6.17 Labor Matters. Except as set forth in Schedule 6.17:
- (i) There are no strikes or lockouts against any members of the Consolidated Group pending or, to the best knowledge of the Credit Parties, threatened which could reasonably be expected to have a Material Adverse Effect.
- (ii) The hours worked by and payments made to employees of the Consolidated Group have not been in violation of the Fair Labor Standards Act or any other applicable federal, state, local or foreign law dealing with such matters in any case where a Material Adverse Effect has occurred or could reasonably be expected to occur as a result of the violation thereof.
- (iii) As of the date hereof, all payments due from members of the Consolidated Group, or for which any claim may be made against a member of the Consolidated Group, on account of wages and employee health and welfare insurance and other benefits, have been paid or accrued as a liability on the books of the respective members

of the Consolidated Group, except to the extent any failure to so pay or accrue could not reasonably be expected to have a Material Adverse Effect.

6.18 No Material Misstatements. None of the information contained in reports, financial statements, exhibits or schedules, taken as a whole, furnished by or on behalf of any member of the Consolidated Group to the Administrative Agent or any Lender in connection with the negotiation of the Credit Documents or included therein or delivered pursuant thereto contained, contains or will contain any material misstatement of fact or omitted, omits or will omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were, are or will be made, not materially misleading.

ARTICLE VII AFFIRMATIVE COVENANTS

The Borrower and the Parent hereby jointly and severally covenant and agree that, so long as this Agreement is in effect or any amounts payable hereunder or under any other Credit Document shall remain outstanding or any Letter of Credit is outstanding, and until all of the Commitments hereunder shall have terminated:

- 7.1 Reporting Requirements. The members of the Consolidated Group will furnish, or cause to be furnished, to the Administrative Agent (with sufficient copies for each of the Lenders):
- (a) Annual Financial Statements. As soon as available, and in any event within 95 days after the close of each fiscal year of the Consolidated Group, a consolidated balance sheet and income statement of the members of the Consolidated Group as of the end of such fiscal year, together with related consolidated statements of operations and retained earnings and of cash flows for such fiscal year, in each case setting forth in comparative form consolidated figures for the preceding fiscal year, all such financial information described above to be in reasonable form and detail and audited by independent certified public accountants of recognized national or regional standing reasonably acceptable to the Administrative Agent and whose opinion shall be to the effect that such financial statements have been prepared in accordance with GAAP (except for changes with which such accountants concur) and shall not be limited as to the scope of the audit or qualified as to the status of the members of the Consolidated Group as a going concern or any other material qualifications or exceptions. Such financial statements shall be accompanied by a certificate of the accountants conducting the annual audit stating that they have reviewed this Agreement and stating further whether, in the course of their audit, they have become aware of any Default or Event of Default and, if any such Default or Event of Default exists, specifying the nature and extent thereof.
- (b) Quarterly Financial Statements. As soon as available, and in any event within 50 days after the close of each fiscal quarter of the Consolidated Group (excluding the fourth fiscal quarter) an internally-prepared consolidated balance sheet and income statement of the members of the Consolidated Group as of the end of such fiscal quarter,

together with related consolidated statements of operations and retained earnings and of cash flows for such fiscal quarter, in each case setting forth in comparative form consolidated figures for the corresponding period of the preceding fiscal year, all such financial information described above to be in reasonable form and detail and reasonably acceptable to the Administrative Agent, and accompanied by a certificate of the chief financial officer of the Parent to the effect that such quarterly financial statements fairly present in all material respects the financial condition of the members of the Consolidated Group and have been prepared in accordance with GAAP, subject to changes resulting from audit and normal year-end audit adjustments.

- (c) Compliance Certificate. At the time of delivery of the financial statements provided for in Sections 7.1(a) and 7.1(b) above, a certificate of the chief financial officer or other Executive Officer of the Parent, in substantially the form of Exhibit "G," (i) demonstrating compliance with the financial covenants contained in
- Section 7.11 by calculation thereof as of the end of each such fiscal period, (ii) containing information regarding the amount of all Asset Dispositions during the prior fiscal year (annually only), and (iii) stating that no Default or Event of Default exists, or if any Default or Event of Default does exist, specifying the nature and extent thereof and what action the Credit Parties propose to take with respect thereto.
- (d) Auditor's Reports. Promptly upon receipt thereof, a copy of any other report or "management letter" submitted by independent accountants to any member of the Consolidated Group in connection with any annual, interim or special audit of the books of such Person.
- (e) Reports. Promptly upon transmission or receipt thereof, copies of any filings and registrations with, and reports to or from, the SEC, or any successor agency, and copies of all financial statements, proxy statements, notices and reports as the Parent shall send to its shareholders.
- (f) Notice of Default. Promptly upon any Executive Officer of a Credit Party obtaining knowledge thereof, written notice of the occurrence of a Default or Event of Default, specifying the nature and existence thereof and what action the Credit Parties propose to take with respect thereto.
- (g) Notice of Litigation and Potential Liabilities. Promptly upon any Executive Officer of a Credit Party obtaining knowledge thereof, written notice of: (i) the commencement of any litigation, arbitral or governmental proceeding against any Credit Party which, if adversely determined, would impose liability on one or more of the members of the Consolidated Group, after allowance for insurance coverage, of \$10,000,000 or more in the aggregate or which would otherwise be likely to have a Material Adverse Effect; or (ii) the institution of any proceedings against such Person with respect to, or the receipt of notice by such Person of potential liability or responsibility for violation, or alleged violation of any Requirement of Law, including any Environmental Laws, the violation of which could have a Material Adverse Effect.

- (h) ERISA. Promptly upon any Executive Officer of a Credit Party obtaining knowledge thereof, written notice of: (i) any event or condition, including, but not limited to, any Reportable Event, that constitutes, or might reasonably lead to, an ERISA Event; (ii) with respect to any Multiemployer Plan, the receipt of notice as prescribed in ERISA or otherwise of any withdrawal liability assessed against the Credit Parties or any ERISA Affiliates, or of a determination that any Multiemployer Plan is in reorganization or insolvent (both within the meaning of Title IV of ERISA); (iii) the failure to make full payment on or before the due date (including extensions) thereof of all amounts which any member of the Consolidated Group or any ERISA Affiliate is required to contribute to each Plan pursuant to its terms and as required to meet the minimum funding standard set forth in ERISA and the IRC with respect thereto; or (iv) any change in the funding status of any Plan that could have a Material Adverse Effect, together with a description of any such event or condition or a copy of any such notice and a statement by an Executive Officer of the Parent briefly setting forth the details regarding such event, condition, or notice, and the action, if any, which has been or is being taken or is proposed to be taken by the Credit Parties with respect thereto. Promptly upon request, the Credit Parties shall furnish the Administrative Agent and the Lenders with such additional information concerning any Plan as may be reasonably requested, including, but not limited to, copies of each annual report/return (Form 5500 series), as well as all schedules and attachments thereto required to be filed with the Department of Labor and/or the Internal Revenue Service pursuant to ERISA and the IRC, respectively, for each "plan year" (within the meaning of Section 3(39) of ERISA).
- (i) Lien Matters. Promptly upon any Executive Officer of a Credit Party obtaining knowledge thereof, written notice of the incurrence or existence of any Lien on or against the Properties of any member of the Consolidated Group, other than a Permitted Lien.
- (j) Other Information. With reasonable promptness upon any such request, such other information regarding the business, properties or financial condition of any member of the Consolidated Group as the Administrative Agent (on behalf of any Lender) or the Required Lenders may reasonably request.
- 7.2 Preservation of Existence and Franchises. Except as a result of or in connection with a merger or consolidation of a Subsidiary permitted under
- Section 8.3 or an Asset Disposition involving the Voting Stock of a Subsidiary permitted under Section 8.5, each Credit Party will do all things necessary to preserve and keep in full force and effect its existence, rights, franchises and authority.
- 7.3 Books and Records. Each member of the Consolidated Group will keep complete and accurate books and records of its transactions in accordance with good accounting practices on the basis of GAAP (including the establishment and maintenance of appropriate reserves).

- 7.4 Compliance with Law. Each member of the Consolidated Group will comply with all Requirements of Law applicable to it or its Properties, if noncompliance with any such Requirement of Law could have a Material Adverse Effect.
- 7.5 Payment of Taxes and Other Indebtedness. Each member of the Consolidated Group will pay and discharge (i) all taxes, assessments and governmental charges or levies imposed upon it, or upon its income or profits, or upon any of its Properties, before they shall become delinquent, (ii) all lawful claims (including claims for labor, materials and supplies) which, if unpaid, might give rise to a Lien upon any of its properties, and (iii) except as prohibited hereunder, all of its other Indebtedness as it shall become due; provided, however, that no member of the Consolidated Group shall be required to pay any such tax, assessment, charge, levy, claim or Indebtedness (A) which is being contested in good faith by appropriate proceedings and as to which adequate reserves therefor have been established in accordance with GAAP or (B) if the failure to make any such payment (x) would not give rise to an immediate right to foreclose on a Lien securing such amounts and (y) could not have a Material Adverse Effect.
- 7.6 Insurance. Each member of the Consolidated Group will at all times maintain in full force and effect insurance (including worker's compensation insurance, liability insurance and casualty insurance) or appropriate risk management programs in such amounts, covering such risks and liabilities and with such deductibles or self-insurance retentions as are reasonable or customary given the nature of its business, the circumstances and geographic area in which such business is being conducted and the availability of insurance coverage at commercially reasonable rates. The present insurance coverage and risk management programs of the members of the Consolidated Group are outlined as to carrier, policy number, expiration date, type and amount on Schedule 7.6.
- 7.7 Maintenance of Property. Each member of the Consolidated Group will maintain and preserve its Properties and equipment material to the conduct of its business in good repair, working order and condition, normal wear and tear, obsolescence and casualty and condemnation excepted, and will make, or cause to be made, in such Properties and equipment from time to time all repairs, renewals, replacements, extensions, additions, betterments and improvements thereto as may be needed or proper, to the extent and in the manner customary for companies in similar businesses.
- 7.8 Performance of Obligations. Each member of the Consolidated Group will perform all of its obligations under the terms of all material agreements, indentures, mortgages, security agreements and other debt instruments to which it is a party or by which it is bound, except where any failure to perform could not reasonably be expected to have a Material Adverse Effect.
- 7.9 Use of Proceeds. The Borrower will use the proceeds of Extensions of Credit solely for the purposes set forth in Section 2.2(c).
- 7.10 Audits/Inspections. Upon reasonable notice and during normal business hours, each member of the Consolidated Group will permit representatives appointed by the Administrative Agent, including independent accountants, agents, attorneys, and appraisers, to visit and inspect its

books and records, facilities and other business assets (other than privileged information and sensitive proprietary information), and to make photocopies or photographs thereof and to write down and record any information such representative obtains and shall permit the Administrative Agent or its representatives to investigate and verify the accuracy of information provided to the Administrative Agent or the Lenders and to discuss all such matters with the officers, employees and, subject to the right of the representatives of the Consolidated Group to be present, representatives of such Person.

7.11 Financial Covenants.

- (a) Consolidated Debt to EBITDA Ratio. The Parent will maintain a Consolidated Debt to EBITDA Ratio, calculated as of the last day of each fiscal quarter beginning September 30, 2002, of not greater than 2.75 to one (2.75:1.00).
- (b) Consolidated Debt to Capitalization Ratio. The Parent will maintain a Consolidated Debt to Capitalization Ratio, calculated as of the last day of each fiscal quarter beginning September 30, 2002, of not greater than 40%.
- (c) Consolidated Tangible Net Worth. The Parent will maintain at all times Consolidated Tangible Net Worth of not less than \$750,000,000.
- 7.12 Liquidity Maintenance. The Credit Parties will at all times maintain in the Investment Portfolio, free and clear of any and all Liens, assets consisting of cash, cash equivalents and Marketable Securities having a fair market value, calculated on a pre-tax basis, of not less than \$100,000,000.
- 7.13 Additional Guarantors. The Borrower will cause each current or future Material Subsidiary to guarantee the prompt payment and performance when due of the Obligations in accordance with the terms and provisions of the Subsidiary Guaranty. As soon as practicable and in any event within 30 days after any Person becomes a direct or indirect Material Subsidiary, the Borrower shall provide the Administrative Agent with written notice thereof and shall cause such Person to execute a Subsidiary Guaranty Joinder Agreement in substantially the same form as Schedule 1 to the Subsidiary Guaranty. Notwithstanding the foregoing, the Lenders agree to release any Material Subsidiary from its obligations under the Subsidiary Guaranty if the Capital Stock of such Subsidiary is sold or transferred pursuant to an Asset Disposition permitted under Section 8.5 or if such Subsidiary becomes a party to a merger or consolidation in connection with a permitted Asset Disposition (such release to be delivered upon the consummation of the relevant transaction).

ARTICLE VIII NEGATIVE COVENANTS

The Borrower and the Parent hereby jointly and severally covenant and agree that, so long as this Agreement is in effect or any amounts payable hereunder or under any other Credit Document shall remain outstanding or any Letter of Credit is outstanding, and until all of the Commitments hereunder shall have terminated:

- 8.1 Indebtedness. No member of the Consolidated Group will create, incur or assume any Indebtedness if any Event of Default has occurred and is continuing at the time such Indebtedness is to be created, incurred or assumed or if any Event of Default would result from the creation, incurrence or assumption of such Indebtedness.
- 8.2 Liens. No member of the Consolidated Group will contract, create, incur, assume or permit to exist, directly or indirectly, any Lien with respect to any of its Property, whether now owned or after acquired, except for the following types and categories of Liens ("Permitted Liens"):
- (i) Liens (other than Liens created or imposed under ERISA) for taxes, assessments or governmental charges or levies not yet due or Liens for taxes being contested in good faith by appropriate proceedings for which adequate reserves determined in accordance with GAAP have been established (and as to which the Property subject to any such Lien is not yet subject to foreclosure, sale or loss on account thereof);
- (ii) statutory Liens of landlords and Liens of carriers, warehousemen, mechanics, materialmen and suppliers and other Liens imposed by law or pursuant to customary reservations or retentions of title arising in the ordinary course of business, provided that such Liens secure only amounts not yet due and payable or, if due and payable, are unfiled and no other action has been taken to enforce the same or are being contested in good faith by appropriate proceedings for which adequate reserves determined in accordance with GAAP have been established (and as to which the Property subject to any such Lien is not yet subject to foreclosure, sale or loss on account thereof);
- (iii) Liens (other than Liens created or imposed under ERISA) incurred or deposits made by any member of the Consolidated Group in the ordinary course of business in connection with workers' compensation, unemployment insurance and other types of insurance, or to secure the performance of tenders, statutory obligations, bids, leases, government contracts, performance and return-of-money bonds and other similar obligations (exclusive of obligations for the payment of Funded Debt);
- (iv) Liens in connection with attachments or judgments (including judgment or appeal bonds), provided that the judgments secured shall, within 60 days after the entry

thereof, have been discharged or execution thereof stayed pending appeal, or shall have been discharged within 60 days after the expiration of any such stay;

- (v) easements, rights-of-way, covenants, restrictions (including zoning restrictions), defects or irregularities in title and other similar charges or encumbrances not, in any material respect, impairing the use of the Property for its intended purposes;
- (vi) Liens arising from UCC financing statements (or equivalent filings, registrations or agreements in foreign jurisdictions) relating to, leases permitted by this Agreement;
- (vii) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods;
- (viii) Liens created in connection with trading activities in the Investment Portfolio and secured only by Marketable Securities in the Investment Portfolio, so long as the Borrower remains at all times in compliance with the requirements of Section 7.12;
- (ix) normal and customary rights of setoff upon deposits of cash in favor of banks or other depository institutions;
- (x) Liens of a collection bank arising under Section 4-210 of the UCC on items in the course of collection; and
- (xi) other Liens (in addition to those permitted by the foregoing clauses (i) through (x)) securing Indebtedness at any time outstanding which in the aggregate does not exceed 5% of Consolidated Total Capitalization.
- 8.3 Merger and Consolidation. No member of the Consolidated Group will enter into any merger or consolidation (other than as contemplated by the Spinoff/Merger or in connection with any other disposition of the oil and gas exploration and production business of the Consolidated Group), except that:
- (i) the Parent or Borrower may enter into any merger or consolidation with any Person which is not a Credit Party so long as (A) it is the surviving entity of such merger or consolidation and remains fully obligated under the Credit Documents for payment of the Obligations, and (B) no Default or Event of Default has occurred or is continuing or would be created by or result from such merger or consolidation, including a breach of the covenants set forth in Sections 7.11 and 7.12;
- (ii) the Borrower and the Parent may merge or consolidate with or into each other, so long as no Default or Event of Default has occurred or is continuing or would be created by or result from such merger or consolidation, including a breach of the covenants set forth in Sections 7.11 and 7.12;

- (iii) any Material Subsidiary may be party to a merger or consolidation with another Material Subsidiary or with the Borrower or the Parent so long as no Default or Event of Default has occurred or is continuing or would be created by or result from such merger or consolidation, including a breach of the covenants set forth in Sections 7.11 and 7.12, provided that, if the Borrower or the Parent is a party to such transaction, it shall be the surviving entity; and
- (iv) a Subsidiary of the Borrower or of the Parent may enter into a transaction of merger or consolidation in connection with an Asset Disposition permitted under clauses (i), (viii) or (ix) of Section 8.5.
- 8.4 Dissolution. No Credit Party will dissolve, liquidate or wind up its affairs.
- 8.5 Asset Dispositions. No member of the Consolidated Group will make any Asset Disposition, except for the following:
- (i) the transactions contemplated by the Spinoff/Merger (or, in the event the Spinoff/Merger is not consummated, any other sale, transfer or other disposition, whether by spinoff or otherwise, of the oil and gas exploration and production business of the Consolidated Group or the stock of Cimarex, Helmerich & Payne Energy Services, Inc., or Mountain Acquisition Co.);
- (ii) the sale of inventory in the ordinary course of business;
- (iii) the sale, lease or other disposition of drillings rigs, machinery and other equipment which is no longer used or useful in the conduct of business or which is obsolete or worn-out;
- (iv) the sale, lease, transfer or disposition of any Property by any member of the Consolidated Group to another member of the Consolidated Group, provided that, if the Property is transferred to a member of the Consolidated Group which is not a Credit Party and such transfer would cause it to become a Material Subsidiary, the Borrower shall cause such member to provide a Subsidiary Guaranty Joinder Agreement in accordance with the requirements of Section 7.13;
- (v) use or operating agreements or short-term Operating Leases entered into by any member of the Consolidated Group as owner or lessor with respect to personal property in the ordinary course of business and on market terms;
- (vi) Operating Leases entered into by any member of the Consolidated Group as owner or lessor with respect to office space in the ordinary course of business and on market terms;

- (vii) sales and trades of securities (including the execution and performance of Forward Sale Contacts), as long as the Credit Parties remain at all times in compliance with the requirements of Section 7.12;
- (viii) the sale or other disposition of any real property (or the sale or other disposition of the Capital Stock of a Subsidiary of the Borrower or the Parent formed for the purpose of effectuating the sale or other disposition of real property), so long as no Default or Event of Default has occurred or is continuing or would be created by or result from such sale or other disposition, including a breach of the covenants set forth in Sections 7.11 and 7.12; and
- (ix) the sale, transfer, lease or other disposition (or any series of related sales, transfers, leases or other dispositions) of Properties of the Consolidated Group, including the Capital Stock of any Subsidiary of the Borrower or the Parent which is not a Credit Party, during any fiscal year which have an aggregate fair market value of less than \$25,000,000, so long as, at the time of any such sale or other disposition, no Default or Event of Default has occurred or is continuing or would be created by or result from such sale or other disposition, including a breach of the covenants set forth in Sections 7.11 and 7.12.
- 8.6 Transactions with Affiliates. No member of the Consolidated Group will enter into or permit to exist any transaction or series of transactions with any officer, director, shareholder, Subsidiary or Affiliate of such Person other than (i) extensions of credit and transfers of cash and assets by and between Credit Parties, (ii) transactions permitted by this Agreement, (iii) to the extent comparable with other members of the Borrower's industry peer group, payment of compensation (including benefits) to and reimbursement of expenses of officers and directors, including stock options and grant and stock option and grant plans, directors' fees, directors' stock option or grant plans, and directors' reimbursable expenses, and (iv) except as otherwise specifically limited in this Agreement, other transactions which are entered into in the ordinary course of such Person's business.
- 8.7 Fiscal Year; Organizational Documents. No member of the Consolidated Group will change its fiscal year or amend, modify or change any of its Charter Documents in a manner adverse to the interests of the Lenders.
- 8.8 Contingent Debt. No member of the Consolidated Group will assume, incur or permit to exist any Contingent Debt other than (i) Contingent Debt relating to Indebtedness of the Consolidated Group permitted under Section 8.1,
- (ii) the endorsement of checks or other negotiable instruments in the ordinary course of business, and (iii) Contingent Debt which does not exceed \$10,000,000 in the aggregate at any time outstanding. No member of the Consolidated Group will enter into any Hedging Agreement except in the ordinary course of business to manage existing or reasonably anticipated risks.

ARTICLE IX EVENTS OF DEFAULT

- 9.1 Events of Default. An Event of Default shall exist upon the occurrence of any of the following specified events (each an "Event of Default"):
- (a) Non-Payment. There shall occur:
- (i) a default in the payment when due of principal on any of the Loans; or
- (ii) a default in the payment when due of any interest on any of the Loans, or of any reimbursement obligations arising from drawings under Letters of Credit, or of any Fees or other amounts owing hereunder, under any of the other Credit Documents or in connection herewith or therewith, and such default shall continue unremedied for a period of five days; or
- (b) Representations. Any representation, warranty or statement made or deemed to be made by the Borrower herein or any Credit Party in any of the other Credit Documents or in any statement or certificate delivered or required to be delivered pursuant hereto or thereto, shall prove untrue in any material respect on the date as of which it was made or deemed to have been made; or
- (c) Covenants. There shall occur a
- (i) default in the due performance or observance of any term, covenant or agreement contained in Sections 7.2, 7.4, 7.9, 7.10, 7.11, 7.12 or 7.13 or in Sections 8.3, 8.4, 8.5, 8.6, 8.7 or 8.8, inclusive;
- (ii) default in the due performance or observance of any term, covenant or agreement contained in Sections 7.1(a),
- (b), (c), (d) or (j) or in Sections 8.1 or 8.2 and such default shall continue unremedied for a period of at least five days after the earlier of an Executive Officer of a Credit Party becoming aware of such default or notice thereof by the Administrative Agent; or
- (iii) default in the due performance or observance by it of any term, covenant or agreement (other than those referred to in subsection (a), (b), (c)(i) or (c)(ii) of this

Section 9.1) contained in this Agreement or any other Credit Document and such default shall continue unremedied for a period of at least 30 days after the earlier of an Executive Officer of a Credit Party becoming aware of such default or notice thereof by the Administrative Agent; or

(d) Other Credit Documents. Any Credit Document shall fail to be in full force and effect or to give the Administrative Agent and/or the Lenders the rights, powers

and privileges purported to be created thereby, or any Credit Party shall so state in writing; or

- (e) Guaranties. Except as the result of or in connection with a merger or disposition of a Subsidiary permitted under Section 8.3 or Section 8.5, the guaranty given by any Guarantor (including any Person which becomes a Guarantor after the Closing Date in accordance with Section 7.13) or any provision of Article XI or of the Subsidiary Guaranty shall cease to be in full force and effect, or any Guarantor (including any Person which becomes a Guarantor after the Closing Date in accordance with Section 7.13) or any Person acting by or on behalf of such Guarantor shall deny or disaffirm such Guarantor's obligations under Article XI or the Subsidiary Guaranty or any Guarantor shall default in the due performance or observance of any term, covenant or agreement on its part to be performed or observed pursuant to Article XI or the Subsidiary Guaranty; or
- (f) Bankruptcy. Any Bankruptcy Event shall occur with respect to any member of the Consolidated Group; or
- (g) Defaults under Other Agreements.
- (i) The occurrence of an "Event of Default" or other default and the expiration of any applicable grace period applicable thereto under (A) the Bank One Agreement or (B) the Medium Term Notes or any indenture or other agreement governing the Medium Term Notes;
- (ii) Any member of the Consolidated Group shall default in the performance or observance (beyond the applicable grace period with respect thereto, if any) of any material obligation or condition of any contract or lease material to the members of the Consolidated Group taken as a whole, and such default has had or could reasonably be expected to result in a Material Adverse Effect; or
- (iii) With respect to any Indebtedness (other than Indebtedness outstanding under this Agreement) in excess of \$5,000,000 in the aggregate for the members of the Consolidated Group taken as a whole, (A) any member of the Consolidated Group shall default in any payment (beyond the applicable grace period with respect thereto, if any) with respect to any such Indebtedness; (B) there shall occur and continue a default in the observance or performance of any provision relating to such Indebtedness or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event or condition shall occur or condition exist, the effect of which default or other event or condition is to cause, or to permit the holder or holders of such Indebtedness (or trustee or agent on behalf of such holders) to cause, any such Indebtedness to become due prior to its stated maturity; or (C) any such Indebtedness shall be declared due and payable, or required to be prepaid other than by a regularly scheduled required prepayment, prior to the stated maturity thereof; or

- (h) Judgments. One or more judgments or decrees shall be entered against one or more of the members of the Consolidated Group involving a liability of \$20,000,000 or more in the aggregate (to the extent not paid or fully covered by insurance provided by a carrier who has acknowledged coverage and has the ability to perform) and any such judgments or decrees shall not have been vacated, discharged or stayed or bonded pending appeal within 30 days from the entry thereof; or
- (i) ERISA. Any of the following events or conditions, if such event or condition could reasonably be expected to involve possible taxes, penalties, and other liabilities against one or more members of the Consolidated Group in an aggregate amount in excess of \$5.0 million: (i) any "accumulated funding deficiency," as such term is defined in Section 302 of ERISA and Section 412 of the IRC, whether or not waived, shall exist with respect to any Plan, or any lien shall arise on the assets of any member of the Consolidated Group or any ERISA Affiliate in favor of the PBGC or a Plan; (ii) an ERISA Event shall occur with respect to a Single Employer Plan, which is, in the reasonable opinion of the Administrative Agent, likely to result in the termination of such Plan for purposes of Title IV of ERISA; (iii) an ERISA Event shall occur with respect to a Multiemployer Plan or Multiple Employer Plan, which is, in the reasonable opinion of the Administrative Agent, likely to result in (A) the termination of such Plan for purposes of Title IV of ERISA, or (B) any member of the Consolidated Group or any ERISA Affiliate incurring any liability in connection with a withdrawal from, reorganization of (within the meaning of Section 4241 of ERISA), or insolvency (within the meaning of

Section 4245 of ERISA) of such Plan; or (iv) any prohibited transaction

(within the meaning of Section 406 of ERISA or Section 4975 of the IRC) or breach of fiduciary responsibility shall occur which may subject any member of the Consolidated Group or any ERISA Affiliate to any liability under Sections 406, 409, 502(i), or 502(l) of ERISA or Section 4975 of the IRC, or under any agreement or other instrument pursuant to which any member of the Consolidated Group or any ERISA Affiliate has agreed or is required to indemnify any person against any such liability; or

- (j) Licenses. Any License necessary for any Credit Party to carry on its business shall be cancelled or revoked by the applicable Governmental Authority or shall be limited or altered by the applicable Governmental Authority in a manner which has had or could reasonably be expected to result in a Material Adverse Effect (unless such cancellation, revocation, limitation or alteration is being contested by such Credit Party in good faith by appropriate legal proceedings and the effectiveness of such action has been stayed pending the outcome of such proceedings), or any suit, action, proceeding or investigation shall be commenced or initiated seeking to cancel, revoke, limit or alter any License which is necessary for any Credit Party to carry on its business (unless such suit, action, proceeding or investigation is being contested or defended in good faith by appropriate legal proceedings, including appellate review, and the effectiveness of any proposed action has been stayed pending the outcome of such proceedings); or
- (k) Ownership. There shall occur a Change of Control.

- 9.2 Acceleration; Remedies. Upon the occurrence of an Event of Default, and at any time thereafter unless and until such Event of Default has been waived by the requisite Lenders (pursuant to the voting requirements of Section 12.6) or cured to the satisfaction of the requisite Lenders (pursuant to the voting procedures in Section 12.6), the Administrative Agent may, and upon the request and direction of the Required Lenders shall, by written notice to the Borrower take any of the following actions:
- (a) Termination of Commitments. Declare the Commitments terminated whereupon the Commitments shall be immediately terminated.
- (b) Acceleration. Declare the unpaid principal of and any accrued interest in respect of all Loans, any reimbursement obligations arising from drawings under Letters of Credit and any and all other indebtedness or obligations of any and every kind owing by the Borrower to the Administrative Agent and/or any of the Lenders hereunder to be due whereupon the same shall be immediately due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby waived by the Borrower. Amounts received after termination of the Commitments and acceleration of the maturity of the Loans and Obligations hereunder, shall be shared ratably among the Lenders as set forth in Section 9.4.
- (c) Enforcement of Rights. Enforce any and all rights and interests created and existing under the Credit Documents and all rights of set-off.

Notwithstanding the foregoing, if an Event of Default specified in Section 9.1(f) shall occur with respect to any member of the Consolidated Group, then the Commitments shall automatically terminate and all Loans, all reimbursement obligations arising from drawings under Letters of Credit, all accrued interest in respect thereof, all accrued and unpaid Fees and other indebtedness or obligations owing to the Administrative Agent and/or any of the Lenders hereunder by the Borrower automatically shall immediately become due and payable without the giving of any notice or other action by the Administrative Agent or the Lenders.

9.3 Waiver of Event of Default. Notwithstanding anything contained in

Section 9.2, if at any time within 60 days after an acceleration of the Loans pursuant to Section 9.2, (i) the Borrower shall pay all arrears of interest and all payments on account of principal which shall have become due otherwise than by acceleration (with interest on principal and, to the extent permitted by law, on overdue interest, at the rates specified in this Agreement) and (ii) all Events of Default and Defaults (other than non-payment of the principal of and accrued interest on the Loans, in each case which is due and payable solely by virtue of acceleration) shall be waived pursuant to Section 12.6, the Required Lenders, by written notice to the Borrower, may at their option rescind and annul the acceleration and its consequences; but such action shall not affect any subsequent Event of Default or Default or impair any right consequent thereon. The provisions of this Section 9.3 are intended merely to bind the Lenders to a decision which may be made at the election of the Required Lenders and are not intended to benefit the Borrower and do

not grant the Borrower the right to require the Lenders to rescind or annual any acceleration hereunder, even if the conditions set forth herein are met.

9.4 Allocation of Payments After Event of Default. Notwithstanding any other provisions of this Agreement to the contrary, after the occurrence and during the continuance of an Event of Default, all amounts collected or received on or in respect of the Obligations (or other amounts owing under the Credit Documents in connection therewith) shall be paid over or delivered as follows:

FIRST, to the payment of all reasonable out-of-pocket costs and expenses (including reasonable attorneys' fees) of the Administrative Agent in connection with enforcing the rights and remedies of the Lenders under the Credit Documents and any protective advances made with respect thereto;

SECOND, to payment of any fees owed to the Administrative Agent hereunder;

THIRD, to the payment of all reasonable out-of-pocket costs and expenses (including reasonable attorneys' fees) of each of the Lenders hereunder in connection with enforcing its rights under the Credit Documents or otherwise with respect to the Obligations owing to such Lender;

FOURTH, to the payment of all accrued interest and fees on or in respect of the Obligations;

FIFTH, to the payment of the outstanding principal amount of the Obligations hereunder (including the payment or cash collateralization of the outstanding LOC Obligations);

SIXTH, to all other Obligations hereunder and other obligations which shall have become due and payable under the Credit Documents otherwise and not repaid pursuant to clauses "FIRST" through "FIFTH" above; and

SEVENTH, to the payment of the surplus, if any, to the Borrower or such other Persons as may be lawfully entitled to receive such surplus.

In carrying out the foregoing, (i) amounts received shall be applied in the order provided until exhausted prior to application to the next succeeding category; and (ii) except as otherwise provided, the Lenders shall receive amounts ratably in accordance with their respective pro rata share (based on the proportion that the then outstanding Obligations held by such Lenders bears to the aggregate amount of Obligations then outstanding) of amounts available to be applied pursuant to clauses "THIRD," "FOURTH," "FIFTH" and "SIXTH"; and (iii) to the extent that any amounts available for distribution pursuant to clause "FIFTH" above are attributable to the issued but undrawn amount of outstanding Letters of Credit, such amounts shall be held by the Administrative Agent in a cash collateral account and applied (A) first, to reimburse the Issuing Lender for any drawings under such Letters of Credit and (B) then, following the expiration of all

Letters of Credit, to all other obligations of the types described in clauses "FOURTH" and "FIFTH" above in the manner provided in this Section 9.4

ARTICLE X AGENCY PROVISIONS

10.1 Appointment, Powers and Immunities. Each Lender hereby irrevocably appoints and authorizes the Administrative Agent to act as its agent under this Agreement and the other Credit Documents with such powers and discretion as are specifically delegated to the Administrative Agent by the terms of this Agreement and the other Credit Documents, together with such other powers as are reasonably incidental thereto. Each Lender further authorizes and directs the Administrative Agent to execute and deliver releases (or similar agreements) to give effect to the provisions of this Agreement and the other Credit Documents, including specifically the provisions of Section 8.5 hereof. The Administrative Agent (which term as used in this sentence and in Section 10.5 and the first sentence of Section 10.6 hereof shall include its Affiliates and its own and its Affiliates' officers, directors, employees, and agents): (i) shall not have any duties or responsibilities except those expressly set forth in this Agreement and shall not be a trustee or fiduciary for any Lender; (ii) shall not be responsible to the Lenders for any recital, statement, representation, or warranty (whether written or oral) made by a member of the Consolidated Group or other Lender in or in connection with any Credit Document or any certificate or other document referred to or provided for in, or received by any of them under, any Credit Document, or for the value, validity, effectiveness, genuineness, enforceability, or sufficiency of any Credit Document, or any other document referred to or provided for therein or for any failure by any Credit Party or any other Person to perform any of its obligations thereunder; (iii) shall not be responsible for or have any duty to ascertain, inquire into, or verify the performance or observance of any covenants or agreements by any Credit Party or the satisfaction of any condition or to inspect the property (including the books and records) of any Credit Party or any of its Subsidiaries or Affiliates;

(iv) shall not be required to initiate or conduct any litigation or collection proceedings under any Credit Document, except as expressly provided under the Credit Documents; and (v) shall not be responsible for any action taken or omitted to be taken by it under or in connection with any Credit Document, except for its own gross negligence or willful misconduct. The Administrative Agent may employ agents and attorneys-in-fact and shall not be responsible for the negligence or misconduct of any such agents or attorneys-in-fact selected by it with reasonable care.

10.2 Reliance by Administrative Agent. The Administrative Agent shall be entitled to rely upon any certification, notice, instrument, writing, or other communication (including any thereof by telephone or telecopy) believed by it to be genuine and correct and to have been signed, sent or made by or on behalf of the proper Person or Persons, and upon advice and statements of legal counsel (including counsel for any Credit Party), independent accountants, and other experts selected by the Administrative Agent. The Administrative Agent may deem and treat the payee of any Note as the holder thereof for all purposes hereof unless and until the Administrative Agent receives and accepts an Assignment and Acceptance executed in accordance with Section 12.3(b) hereof. As to any matters not expressly provided for by this Agreement, the Administrative Agent shall not be required to exercise any discretion or take any

action, but shall be required to act or to refrain from acting (and shall be fully protected in so acting or refraining from acting) upon the instructions of the Required Lenders, and such instructions shall be binding on all of the Lenders; provided, however, that the Administrative Agent shall not be required to take any action that exposes the Administrative Agent to personal liability or that is contrary to any Credit Document or applicable law or unless it shall first be indemnified to its satisfaction by the Lenders against any and all liability and expense which may be incurred by it by reason of taking any such action.

10.3 Defaults. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of a Default or Event of Default unless the Administrative Agent has received written notice from a Lender or a Credit Party specifying such Default or Event of Default and stating that such notice is a "Notice of Default." In the event that the Administrative Agent receives such a notice of the occurrence of a Default or Event of Default, the Administrative Agent shall give prompt notice thereof to the Lenders. The Administrative Agent shall (subject to Section 12.2) take such action with respect to such Default or Event of Default as shall reasonably be directed by the Required Lenders (or such other Lenders as required by Section 12.6), provided that, unless and until the Administrative Agent shall have received such directions, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable in the best interest of the Lenders.

10.4 Rights as a Lender. With respect to its Commitment and the Loans made by it, Bank of Oklahoma (and any successor acting as Administrative Agent) in its capacity as a Lender hereunder shall have the same rights and powers hereunder as any other Lender and may exercise the same as though it were not acting as the Administrative Agent, and the term "Lender" or "Lenders" shall, unless the context otherwise indicates, include the Administrative Agent in its individual capacity. Bank of Oklahoma (and any successor acting as Administrative Agent) and its Affiliates may (without having to account therefor to any Lender) accept deposits from, lend money to, make investments in, provide services to, and generally engage in any kind of lending, trust, or other business with any Credit Party or any of its Subsidiaries or Affiliates as if it were not acting as Administrative Agent, and Bank of Oklahoma (and any successor acting as Administrative Agent) and its Affiliates may accept fees and other consideration from any Credit Party or any of its Subsidiaries or Affiliates for services in connection with this Agreement or otherwise without having to account for the same to the Lenders.

10.5 Indemnification. The Lenders agree to indemnify the Administrative Agent (to the extent not reimbursed under Section 12.5 hereof, but without limiting the obligations of the Borrower under Section 12.5) ratably (in accordance with their respective Revolving Commitments (or, if the Revolving Commitments have been terminated, their respective interests in the outstanding Loans and Participation Interests in Letters of Credit) for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses (including reasonable attorneys' fees), or disbursements of any kind and nature whatsoever that may be imposed on, incurred by or asserted against the Administrative Agent in such capacity (including by any Lender) in any way relating to or arising out of any Credit Document or the transactions contemplated thereby or any action taken or omitted by the Administrative Agent under any

Credit Document; provided that no Lender shall be liable for any of the foregoing to the extent they arise from the gross negligence or willful misconduct of the Person to be indemnified. Without limitation of the foregoing, each Lender agrees to reimburse the Administrative Agent promptly upon demand for its ratable share of any costs or expenses payable by the Borrower under

Section 12.5, to the extent that the Administrative Agent is not promptly reimbursed for such costs and expenses by the Borrower. The agreements in this

Section 10.5 shall survive the repayment of the Loans, LOC Obligations and other obligations under the Credit Documents and the termination of the Commitments hereunder.

10.6 Non-Reliance on Administrative Agent and Other Lenders. Each Lender agrees that it has, independently and without reliance on the Administrative Agent or any other Lender, and based on such documents and information as it has deemed appropriate, made its own credit analysis of the Credit Parties and their Subsidiaries and decision to enter into this Agreement and that it will, independently and without reliance upon the Administrative Agent or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own analysis and decisions in taking or not taking action under the Credit Documents. Except for notices, reports, and other documents and information expressly required to be furnished to the Lenders by the Administrative Agent hereunder, the Administrative Agent shall not have any duty or responsibility to provide any Lender with any credit or other information concerning the affairs, financial condition, or business of any Credit Party or any of its Subsidiaries or Affiliates that may come into the possession of the Administrative Agent or any of its Affiliates.

10.7 Successor Administrative Agent. The Administrative Agent may resign at any time by giving notice thereof to the Lenders and the Credit Parties. Upon any such resignation, the Required Lenders shall have the right to appoint a successor Administrative Agent. If no successor Administrative Agent shall have been so appointed by the Required Lenders and shall have accepted such appointment within 30 days after the retiring Administrative Agent's giving of notice of resignation, then the retiring Administrative Agent may, on behalf of the Lenders, appoint a successor Administrative Agent which shall be a commercial bank organized under the laws of the United States having combined capital and surplus of at least \$100,000,000. Upon the acceptance of any appointment as Administrative Agent hereunder by a successor, such successor shall thereupon succeed to and become vested with all the rights, powers, discretion, privileges, and duties of the retiring Administrative Agent, and the retiring Administrative Agent shall be discharged from its duties and obligations hereunder. After any retiring Administrative Agent's resignation hereunder as Administrative Agent, the provisions of this Article X shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as Administrative Agent.

ARTICLE XI PARENT GUARANTY

11.1 Guaranty. The Parent unconditionally and irrevocably guarantees the full and punctual payment of all existing and future Obligations of the Borrower to the Lenders or the Administrative Agent arising under or in connection with or evidenced by this Agreement or any

other Credit Document, as and when the same shall become due and payable, whether at the stated maturity, upon acceleration or otherwise, in accordance with the terms hereof and thereof. If the Borrower fails to pay when due any Obligation guaranteed hereby, the Parent unconditionally agrees to cause such payment to be made punctually as and when the same shall become due and payable, whether at the stated maturity, upon acceleration or otherwise.

- 11.2 Guarantee Unconditional. The obligations of the Parent under this Article XI are absolute and unconditional. Without limiting the generality of the foregoing, the obligations of the Parent under this Article XI shall not be impaired, released, discharged or otherwise affected by:
- (i) any extension, renewal, settlement, compromise, waiver or release in respect of any obligation of the Borrower or any other Credit Party under this Agreement or any other Credit Document, by operation of law or otherwise;
- (ii) any modification, amendment or waiver of or supplement to this Agreement or any Credit Document;
- (iii) any release, impairment or invalidity of any guarantee or other liability of any other Credit Party or third party for any obligation of the Borrower under this Agreement or any other Credit Document;
- (iv) any change in the corporate existence, structure or ownership of the Borrower or any other Credit Party or any insolvency, bankruptcy, reorganization or other similar proceeding affecting the Borrower or any other Credit Party;
- (v) the existence of any claim, set-off or other rights which the Parent may have at any time against the Borrower or any other Credit Party, the Administrative Agent, any Lender or any other Person, whether or not arising in connection with the Revolving Loans and this Agreement or any unrelated transaction;
- (vi) any invalidity or unenforceability relating to or against the Borrower or any other Credit Party for any reason of this Agreement or any other Credit Document, or any provision of applicable law or regulation purporting to prohibit the payment by any other Credit Party of any amount payable by it under this Agreement or any other Credit Document; or
- (vii) any other act or omission to act or delay of any kind by any other Credit Party, any Lender or any other Person or any other circumstance which might, but for the provisions of this Section 11.2, constitute a legal or equitable discharge of the Parent's obligations under this Article XI.
- 11.3 Discharge Only Upon Payment in Full; Reinstatement in Certain Circumstances. The Parent's obligations under this Article XI constitute a continuing guaranty and shall remain in full force and effect until the Revolving Commitments shall have been terminated, all Letters

of Credit shall have expired or been terminated, and all amounts payable under this Agreement and the Credit Documents shall have been indefeasibly paid in full. If at any time any amount payable by the Borrower under this Agreement or any other Credit Document is rescinded or must be otherwise restored or returned upon the insolvency, bankruptcy or reorganization of the Borrower or any other Credit Party or otherwise, the Parent's obligations under this Article XI with respect to such payment shall be reinstated at such time as though such payment had become due but had not been made at such time.

- 11.4 Waiver. The Parent irrevocably waives acceptance hereof, presentment, demand, protest and any notice not provided for herein, as well as any requirement that at any time any action be taken by any Person against the Borrower or other Credit Party or any other Person.
- 11.5 Subrogation. If the Parent makes any payment under this Article XI with respect to the obligations of the Borrower, the Parent shall be subrogated to the rights of the payee against the Borrower with respect to the portion of such obligations paid by the Parent; provided that the Parent shall not enforce any payment by way of subrogation or contribution against the Borrower so long as any amount payable under this Agreement or any other Credit Document remains unpaid.
- 11.6 Stay of Acceleration. If acceleration of the time for payment of any amount payable by the Borrower under this Agreement or any other Credit Document is stayed upon the insolvency, bankruptcy or reorganization of the Borrower, all such amounts otherwise subject to acceleration under the terms of such Credit Document shall nonetheless be payable by the Parent under this Article XI forthwith on demand by the Administrative Agent made at the request of the Required Lenders.
- 11.7 Successors and Assigns. The guarantee by the Parent under this Article XI is for the benefit of the Administrative Agent and the Lenders, and their respective successors and assigns. If any Revolving Loans, participations in Letters of Credit or Swingline Loans or other amounts payable under this Agreement and the other Credit Documents are assigned pursuant to Section 12.3(b), the rights under this Article XI, to the extent applicable to the indebtedness so assigned, shall be transferred with such indebtedness.

ARTICLE XII GENERAL PROVISIONS

12.1 Notices. Except as otherwise expressly provided herein, all notices and other communications shall have been duly given and shall be effective (a) when delivered, (b) when transmitted via telecopy (or other facsimile device) to the number set out below, (c) the Business Day following the day on which the same has been delivered prepaid to a reputable national overnight air courier service, or (d) the third Business Day following the day on which the same is sent by certified or registered mail, postage prepaid, in each case to the respective parties at the address, in the case of the Borrower and the Administrative Agent, set forth below, and, in the case of the Lenders, set forth on the signature pages hereto, or at such other address as such party may specify by written notice to the other parties hereto:

If to the Borrower:

Helmerich & Payne International Drilling Co. c/o Helmerich & Payne, Inc. Utica at Twenty-First Tulsa, Oklahoma 74114 Fax: (918) 743-2671 Attn: Chief Financial Officer

with a copy to:

Helmerich & Payne International Drilling Co. c/o Helmerich & Payne, Inc. Utica at Twenty-First Tulsa, Oklahoma 74114 Fax: (918) 743-2671 Attention: General Counsel

If to the Administrative Agent:

Bank of Oklahoma, N.A.
Bank of Oklahoma Tower
One Williams Center
Tulsa, Oklahoma 74192
Attn: John M. Tyson, Asst. Vice President
Fax: (918) 280-3366

12.2 Right of Set-Off; Adjustments. Upon the occurrence and during the continuance of any Event of Default, each Lender (and each of its Affiliates) is hereby authorized at any time and from time to time, to the fullest extent permitted by law, to set off and apply any and all deposits (general or special, time or demand, provisional or final) at any time held and other indebtedness at any time owing by such Lender (or any of its Affiliates) to or for the credit or the account of the Borrower against any and all of the obligations of the Borrower now or hereafter existing under this Agreement, under the Notes, under any other Credit Document or otherwise, irrespective of whether such Lender shall have made any demand under hereunder or thereunder and although such obligations may be unmatured and regardless of the adequacy of any other collateral securing the Obligations. Each Credit Party hereby irrevocably waives any obligation that any Lender may have to any set-off based on the failure of any Agent or any Lender to exercise any rights that it may have with respect to any other collateral prior to the exercise of any set-off. Each Lender agrees promptly to notify the Borrower after any such set-off and application made by such Lender; provided, however, that the failure to give such notice shall not affect the validity of such set-off and application. The rights of each Lender under this Section 12.2 are in addition to other rights and remedies (including other rights of set-off) that such Lender may have.

12.3 Benefit of Agreement.

- (a) Parties to Agreement. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective successors and assigns; provided that the Borrower may not assign or transfer any of its interests and obligations without prior written consent of each of the Lenders; provided further that the rights of each Lender to transfer, assign or grant participations in its rights and/or obligations hereunder shall be limited as set forth in this Section 12.3.
- (b) Assignment by a Lender. Each Lender may assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Loans, its Notes, Commitments and Participations); provided, however, that
- (i) each such assignment shall be to one or more Eligible Assignees;
- (ii) except in the case of an assignment to an existing Lender, an Affiliate of an existing Lender or any fund that invests in bank loans or similar extensions of credit and is advised or managed by a Lender (or an Affiliate of a Lender) or an investment advisor (or any Affiliate to an investment advisor) to an existing Lender or an assignment of all of a Lender's rights and obligations under this Agreement, any such partial assignment shall be in an amount at least equal to \$1,000,000 (or, if less, the remaining amount of the Commitment being assigned by such Lender) or such lesser amount agreed to by the Borrower and the Administrative Agent;
- (iii) each such assignment shall consist of the same percentage of all of the Obligations and Commitments of the existing Lender hereunder; and
- (iv) the parties to such assignment shall execute and deliver to the Administrative Agent for its acceptance an Assignment and Acceptance in the form of Exhibit "H" hereto, together with any Note subject to such assignment and a processing fee of \$3,500; provided, no such fee shall be payable in respect of assignments to any Affiliate of a Lender or a fund that invests in bank loans or similar extensions of credit and is advised or managed by a Lender (or an Affiliate of a Lender) or an investment advisor (or any Affiliate to an investment advisor) to an existing Lender.

Upon execution, delivery, and acceptance of such Assignment and Acceptance, the assignee thereunder shall be a party hereto and, to the extent of such assignment, have the obligations, rights, and benefits of a Lender hereunder and the assigning Lender shall, to the extent of such assignment, relinquish its rights and be released from its obligations under this Agreement. Upon the consummation of any assignment pursuant to this Section 12.3(b), the Administrative Agent and the Borrower shall make appropriate

arrangements so that, if required, new Notes are issued to the assignor and the assignee. Upon or prior to the consummation of any assignment to an assignee that is not already a Lender hereunder pursuant to this

Section 12.3(b), such assignee shall deliver to the Borrower and the Administrative Agent certification as to exemption from deduction or withholding of taxes in accordance with Section 4.6. To the extent that an assignment of all or any portion of a Lender's rights and obligations under this Agreement (including all or a portion of its Loans, its Note and its Commitment) pursuant to this Section 12.3(b) would, due to circumstances existing at the time of such assignment, result in increased costs under Section 4.6 from those being charged by the assigning Lender prior to such assignment, then the Borrower shall not be obligated to pay such increased costs (although the Borrower shall be obligated to pay other increased costs resulting from changes after the date of such assignment).

- (c) Register. The Administrative Agent shall maintain at its address referred to in Section 12.1 a copy of each Assignment and Acceptance delivered to and accepted by it and a register for the recordation of the names and addresses of the Lenders and the Commitment of, and principal amount of the Loans owing to, each Lender from time to time (the "Register"). The entries in the Register shall be conclusive and binding for all purposes, absent manifest error, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register as a Lender hereunder for all purposes of this Agreement. The Register shall be available for inspection by the Borrower or any Lender at any reasonable time and from time to time upon reasonable prior notice. Any assignment of any Loan or other obligations shall be effective only upon an entry with respect thereto being made in the Register.
- (d) Sale of Participations. Each Lender may, without notice to or consent of the Borrower, the Administrative Agent, sell participations to one or more Persons in all or a portion of its rights, obligations or rights and obligations under this Agreement (including all or a portion of its Commitment or its Loans); provided, however, that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, (iii) the participant shall be entitled to the benefit of the yield protection provisions contained in Sections 4.2 through 4.7, inclusive, and the right of set-off contained in Section 12.2, provided, however, the participant shall not be entitled to receive any greater amount pursuant to Sections 4.2 through 4.6, inclusive, than the participating Lender would have been entitled to receive in respect of the amount of the participation sold by such participating Lender to such participant had no such participation occurred, and (v) the Credit Parties shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement, and such Lender shall retain the sole right to enforce the obligations of the Credit Parties relating to the Loans and other obligations owing to such Lender and to approve any amendment, modification, or waiver of any provision of this Agreement (other than amendments, modifications, or waivers (i) decreasing the amount of principal of or the rate at which interest is payable on such Loans or Notes, (ii) extending any scheduled principal payment date or date fixed for the payment of interest on such Loans

or Notes, (iii) extending its Commitment, or (iv) except as the result of or in connection with a merger of a member of the Consolidated Group permitted under Section 8.3, release the Borrower or substantially all of the other Credit Parties from its or their obligations under the Credit Documents).

- (e) Certain Permitted Assignments. Notwithstanding any other provision set forth in this Agreement, any Lender may, without notice to or consent of the Borrower or the Administrative Agent, at any time assign and pledge all or any portion of its Loans and its Note (i) to any Federal Reserve Bank as collateral security pursuant to Regulation A and any Operating Circular issued by such Federal Reserve Bank or (ii) in the case of any Lender which has made Loans hereunder and is a fund that invests in bank loans or similar extensions of credit, any such Lender may assign or pledge all or any portion of its Loans and its Note to any holders of obligations owed, or securities issued, by such fund, as security for such obligations or securities, or to any trustee for, or any other representative of, such holders. No such assignment shall release the assigning Lender from its obligations hereunder.
- (f) Furnishing of Information. Any Lender may furnish any information concerning the members of the Consolidated Group in the possession of such Lender from time to time to assignees and participants (including prospective assignees and participants), subject, however, to the provisions of Section 12.16 hereof.
- 12.4 No Waiver; Remedies Cumulative. No failure or delay on the part of the Administrative Agent or any Lender in exercising any right, power or privilege hereunder or under any other Credit Document and no course of dealing between the Administrative Agent or any Lender and any of the Credit Parties shall operate as a waiver thereof; nor shall any single or partial exercise of any right, power or privilege hereunder or under any other Credit Document preclude any other or further exercise thereof or the exercise of any other right, power or privilege hereunder or thereunder. The rights and remedies provided herein are cumulative and not exclusive of any rights or remedies which the Administrative Agent or any Lender would otherwise have. No notice to or demand on any Credit Party in any case shall entitle the Credit Parties to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of the Administrative Agent or the Lenders to any other or further action in any circumstances without notice or demand.

12.5 Expenses; Indemnification.

(a) Reimbursement of Expenses. The Borrower agrees to pay on demand all costs and expenses reasonably incurred by the Administrative Agent in connection with the syndication, preparation, execution, delivery, administration, modification, and amendment of this Agreement, the other Credit Documents, and the other documents to be delivered hereunder, including the reasonable fees and expenses of counsel for the Administrative Agent with respect thereto and with respect to advising the Administrative Agent as to its rights and responsibilities under the Credit Documents. The Borrower further agrees to pay on demand all costs and expenses reasonably incurred by the

Administrative Agent and the Lenders, if any (including reasonable attorneys' fees and expenses), in connection with the enforcement (whether through negotiations, legal proceedings, or otherwise) of the Credit Documents and the other documents to be delivered thereunder.

- (b) Indemnification. The Borrower agrees to indemnify and hold harmless the Administrative Agent and each Lender and each of their Affiliates and their respective officers, directors, trustees, employees, agents, and advisors (each, an "Indemnified Party") from and against any and all claims, damages, losses, liabilities, costs, and expenses (including reasonable attorneys' fees, disbursements and other charges) that may be incurred by or asserted or awarded against any Indemnified Party, in each case arising out of or in connection with or by reason of (including in connection with any investigation, litigation, or proceeding and regardless of whether such Indemnified Party is a party thereto or preparation of defense in connection therewith) the Credit Documents or any of the transactions contemplated herein or the actual or proposed use of the proceeds of the Loans, except to the extent such claim, damage, loss, liability, cost, or expense is found in a final, non-appealable judgment by a court of competent jurisdiction to have resulted from such Indemnified Party's bad faith, gross negligence or willful misconduct. In the case of an investigation, litigation or other proceeding to which the indemnity in this Section 12.5 applies, such indemnity shall be effective whether or not such investigation, litigation or proceeding is brought by any of the Credit Parties, their respective directors, shareholders or creditors or an Indemnified Party or any other Person or any Indemnified Party is otherwise a party thereto and whether or not the transactions contemplated hereby are consummated. The Borrower agrees not to assert any claim against the Administrative Agent, any Lender, any of their Affiliates, or any of their respective directors, officers, employees, attorneys, agents, and advisors, on any theory of liability, for special, indirect, consequential, or punitive damages arising out of or otherwise relating to the Credit Documents, any of the transactions contemplated herein or the actual or proposed use of
- (c) Survival. Without prejudice to the survival of any other agreement of the Borrower hereunder, the agreements and obligations of the Borrower contained in this Section 12.5 shall survive the repayment of the Loans, LOC Obligations and other obligations under the Credit Documents and the termination of the Commitments hereunder.
- 12.6 Amendments, Waivers and Consents. Neither this Agreement nor any other Credit Document nor any of the terms hereof or thereof (including any condition precedent to any Extension of Credit set forth in Section 5.2 or 5.3) may be amended, changed, waived, discharged or terminated unless such amendment, change, waiver, discharge or termination is in writing entered into by, or approved in writing by, the Required Lenders and the Borrower, provided, however, that:
- (a) without the consent of each Lender affected thereby, neither this Agreement nor any other Credit Document may be amended to:

- (i) extend the Revolving Commitment Termination Date or the Facility Maturity Date (whether in accordance with Section 2.5(c) or otherwise), or the final maturity of any Loan or any reimbursement obligation, or any portion thereof, arising from drawings under Letters of Credit;
- (ii) reduce the rate or extend the time of payment of interest (other than as a result of waiving the applicability of any post-default increase in interest rates) thereon or Fees hereunder;
- (iii) reduce or waive the principal amount of any Loan or of any reimbursement obligation, or any portion thereof, arising from drawings under Letters of Credit;
- (iv) increase the Commitment of a Lender over the amount thereof in effect (it being understood and agreed that a waiver of any Default or Event of Default or mandatory reduction in the Commitments shall not constitute a change in the terms of any Commitment of any Lender);
- (v) release either the Borrower or the Parent from its obligations under the Credit Documents or, except as provided in the last sentence of Section 7.13, release any Material Subsidiary from its obligations under the Credit Documents;
- (vi) amend, modify or waive any provision of this Section 12.6 or Sections 2.8, 3.5, 3.6, 4.1 through 4.8, 9.1(a), 12.2, 12.3, 12.5 or 12.9 or the last sentence of Section 2.5(c);
- (vii) reduce any percentage specified in, or otherwise modify, the definition of Required Lenders, or change any provision requiring the consent of all Lenders; or
- (viii) consent to the assignment or transfer by the Borrower or all or substantially all of the other Credit Parties of any of its or their rights and obligations under (or in respect of) the Credit Documents except as permitted thereby.
- (b) without the consent of the Administrative Agent, no provision of Article X may be amended;
- (c) without the consent of the Issuing Lender, no provision of Section 2.6 may be amended; and
- (d) without the consent of the Swingline Lender, no provision of Section 2.7 may be amended.

Notwithstanding the fact that the consent of all the Lenders is required in certain circumstances as set forth above, (x) each Lender is entitled to vote as such Lender sees fit on any bankruptcy reorganization plan that affects the Loans, and each Lender acknowledges that the provisions of Section 1126(c) of the Bankruptcy Code supersedes the unanimous consent provisions set forth herein. In the event that any fee is imposed by the Lenders as a condition to or in connection with any amendment, modification, interpretation, termination, waiver or consent with respect to this Agreement or any of the other Credit Documents, such fee shall be paid to the Lenders in accordance with their respective Percentages.

- 12.7 Counterparts. This Agreement may be executed in any number of counterparts, each of which when so executed and delivered shall be an original, but all of which shall constitute one and the same instrument. It shall not be necessary in making proof of this Agreement to produce or account for more than one such counterpart for each of the parties hereto. Delivery by facsimile by any of the parties hereto of an executed counterpart of this Agreement shall be as effective as an original executed counterpart hereof and shall be deemed a representation that an original executed counterpart hereof will be delivered.
- 12.8 Headings. The headings of the sections and subsections hereof are provided for convenience only and shall not in any way affect the meaning or construction of any provision of this Agreement.
- 12.9 Survival. All indemnities set forth herein, including those set forth in Sections 2.6(e), 4.6, 4.7, 10.5 and 12.5, shall survive the execution and delivery of this Agreement, the making of the Loans, the issuance of the Letters of Credit, the repayment of the Loans, LOC Obligations and other obligations under the Credit Documents and the termination of the Commitments hereunder, and all representations and warranties made by the Borrower herein shall survive delivery of the Notes and the making of the Loans hereunder.
- 12.10 Governing Law. THIS AGREEMENT AND, UNLESS OTHERWISE EXPRESSLY PROVIDED THEREIN, THE OTHER CREDIT DOCUMENTS AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER AND THEREUNDER SHALL BE GOVERNED BY AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH THE LAWS OF THE STATE OF OKLAHOMA.
- 12.11 Waiver of Jury Trial. TO THE EXTENT PERMITTED BY LAW, EACH OF THE ADMINISTRATIVE AGENT, THE LENDERS, THE BORROWER AND THE PARENT HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY OF THE OTHER CREDIT DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY.
- 12.12 Severability. If any provision of any of the Credit Documents is determined to be illegal, invalid or unenforceable, such provision shall be fully severable and the remaining provisions shall remain in full force and effect and shall be construed without giving effect to the illegal, invalid or unenforceable provisions.

- 12.13 Entire Agreement. This Agreement together with the other Credit Documents represent the entire agreement of the parties hereto and thereto, and supersede all prior agreements and understandings, oral or written, if any, including any commitment letters or correspondence relating to the Credit Documents or the transactions contemplated herein and therein. To the extent that there is any irreconcilable conflict or inconsistency between any provision of this Agreement and any provision of any other Credit Document, this Agreement shall govern and control. Upon the Closing Date and the making of the initial Revolving Loan hereunder, the existing revolving credit facility established by Bank of Oklahoma in favor of the Parent (and all commitments thereunder) will terminate and all loan agreements, promissory notes and other agreements relating to such revolving credit facility will be cancelled or terminated, except that any provisions contained in such agreements relating to indemnification by the Borrower shall survive such cancellation and termination to the extent that such agreements expressly provide for such survival.
- 12.14 Binding Effect. This Agreement shall become effective at such time on or after the Closing Date when it shall have been executed by the Borrower and the Administrative Agent, and the Administrative Agent shall have received copies hereof (telefaxed or otherwise) which, when taken together, bear the signatures of each Lender, and thereafter this Agreement shall be binding upon and inure to the benefit of the Borrower, the Parent, the Administrative Agent and each Lender and their respective successors and assigns.
- 12.15 Termination. The term of this Agreement shall be effective until no Loans, LOC Obligations or any other amounts payable hereunder or under any of the other Credit Documents shall remain outstanding, no Letters of Credit shall be outstanding, and all of the Commitments hereunder shall have expired or been terminated.
- 12.16 Confidentiality. The Administrative Agent and each Lender (each, a "Lending Party") agrees to keep (and to cause its affiliates and its and their respective officers, directors, employees, agents and advisors to keep) confidential any information furnished or made available to it by the Credit Parties pursuant to this Agreement that is marked confidential; provided that nothing herein shall prevent any Lending Party from disclosing such information
- (i) to any other Lending Party or any Affiliate of any Lending Party, or any officer, director, employee, agent, or advisor of any Lending Party or Affiliate of any Lending Party, (ii) to any other Person if reasonably incidental to the administration of the credit facility provided herein, (iii) as required by any law, rule, or regulation, (iv) upon the order of any court or administrative agency or pursuant to subpoena or other legal process, (v) upon the request or demand of any regulatory agency or authority, (vi) that is or becomes available to the public or that is or becomes available to any Lending Party other than as a result of a disclosure prohibited by this Agreement, (vii) in connection with any litigation to which such Lending Party or any of its Affiliates may be a party, (viii) to the extent necessary in connection with the exercise of any remedy under this Agreement or any other Credit Document, (ix) to any direct or indirect contractual counterparty in swap agreements or such contractual counterparty's professional advisor (so long as such contractual counterparty or professional advisor to such contractual counterparty has agreed in a writing enforceable by the Borrower to be bound by the provisions of this Section 12.16) and (x) subject to

provisions the same as those contained in this Section 12.16, to any actual or proposed participant or assignee.

12.17 Spinoff/Merger; Other Proposed Transactions.

- (a) Spinoff/Merger. Anything in this Agreement to the contrary notwithstanding, nothing in this Agreement shall prohibit, and no Material Adverse Effect or Default or Event of Default hereunder shall result or be deemed to result from, (i) the consummation by the Parent of the Spinoff/Merger or any transaction contemplated thereby, or (ii) in the event the Spinoff/Merger is not consummated, the sale or other disposition of the oil and gas exploration and production business of the Consolidated Group and/or the stock of Cimarex, Helmerich & Payne Energy Services, Inc., or Mountain Acquisition Co.
- (b) Other Proposed Transactions. The parties acknowledge that one or more members of the Consolidated Group intend to undertake and consummate the following proposed transactions (the "Proposed Transactions"): (i) the sale and issuance of the Medium Term Notes, and (ii) an internal reorganization of the Consolidated Group pursuant to which (x) substantially all of the Properties of the Parent (other than those primarily related to the oil and gas exploration, production, marketing and sales operations of the Parent) will be transferred to the Borrower or one or more Subsidiaries of the Borrower and (y) the current direct and indirect Subsidiaries of the Parent will become direct or indirect subsidiaries of the Borrower. The parties agree that, notwithstanding any provision of this Agreement to the contrary, provided that the Proposed Transactions are carried out without a breach of the covenants in Sections 7.11 or 7.12 and, with respect to the internal reorganization, in compliance with the requirements of

Section 7.13 (if applicable), none of the Proposed Transactions will be deemed to have a Material Adverse Effect or otherwise to constitute or create a Default or Event of Default under this Agreement.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK] [SIGNATURES APPEAR ON THE FOLLOWING PAGES] IN WITNESS WHEREOF, each of the parties hereto has caused a counterpart of this Agreement to be duly executed and delivered as of the date first above written.

BORROWER: HELMERICH & PAYNE INTERNATIONAL

DRILLING CO.,

a Delaware corporation

	Name:
PARENT:	HELMERICH & PAYNE, INC., a Delaware corporation
	By: Name: Title:
ADMINISTRATIVE AGENT:	BANK OF OKLAHOMA, NATIONAL ASSOCIATION
	By:
Notice Address: Bank of Oklahoma Tower One Williams Center Tulsa, Oklahoma 74172	

Attn: John M. Tyson, Assistant Vice President Fax: (918) 280-3368

LENDERS: BANK OF OKLAHOMA, NATIONAL

ASSOCIATION

Ву:	Name:
Title:	
Revolving Commitment: \$50,000,000	
Notice Address: Bank of Oklahoma Tower One Williams Center	

Tulsa, Oklahoma 74192

WELLS FARGO BANK TEXAS, N.A.

By:	
Name:	
Title:	

Revolving Commitment: \$30,000,000

Notice Address: 1000 Louisiana Street, 3rd Floor Houston, Texas 77002

Attention: Philip C. Lauinger III, Vice President FAX: (713) 739-1087

MIDFIRST BANK

Ву:	
Name:_	
Title:_	

Revolving Commitment: \$20,000,000

Notice Address: 321 S. Boston, Suite 104 Tulsa, Oklahoma 74103

Attention: Ed Fariss, Senior Vice President FAX: (918) 587-0369

UMB BANK, N.A.

	By:
	Name:
	Title:
Revolving Commitment: \$15,000,000	
Notice Address:	
1437 S. Boulder	
Tulsa, Oklahoma 74119	
Attention: Richard J. Lehrter,	
Community Bank President	
FAX: ()	

COMMERCE BANK, N.A.

By:	
Name:	
Title:	

Revolving Commitment: \$10,000,000

Notice Address: 1000 Walnut Street, BB17-1 Kansas City, Missouri 64106

Attention: David Emley, Assistant Vice President FAX: (816) 234-7290

82

End of Filing



© 2005 | EDGAR Online, Inc.